Expecting More:

Relations Between Teacher Expectations, Teacher Emotions, and Student Self-Concept Outcomes

Mengnan Li

A thesis submitted in fulfilment of the requirements for the degree of Doctor of Philosophy in Education, The University of Auckland, 2022.

Abstract

Although studies of teacher expectations, teacher emotions, and student self-concept have flourished in the past several decades, little is known about how these three factors are associated with each other. However, teacher beliefs and emotions are both significant in influencing teaching practices and students' outcomes, and student self-concept also plays a critical role in student overall development. Therefore, this doctoral project aimed to explore the interplay of teacher expectations and teacher emotions, and the role of teacher expectations in the development of student academic and non-academic self-concept.

This doctoral thesis consisted of three studies in the context of the Chinese high school. In Study 1, questionnaires were collected from 135 Chinese homeroom teachers from 14 high schools. The structural equation modelling showed that positive emotions increased as teacher expectations increased, and negative emotions decreased when teacher expectations were higher. The results suggested that teachers might improve their emotional experiences by developing more positive expectations for their students. In Study 2, 12 interviews were conducted with teachers from different expectation groups to further explore how teacher expectations could explain the variance in their emotions. Interview results revealed that teachers with different levels of expectations in their emotional strategies, teaching approaches, and their use of contextual resources. Finally, by collecting two time points of students' self-concept data from 348 students, Study 3 investigated the predictive role of teacher expectations in some student self-concept outcomes, including both academic and nonacademic constructs.

Consequently, this doctoral project explored teacher expectation effects in relation to both teacher experiences and student outcomes. This thesis provided significant contributions and insights into how teacher expectations might contribute to teachers' emotional experiences and student self-concept, which could be then used for subsequent sociopsychological research and teacher professional development. The findings suggested that teachers should be facilitated to be more aware of their expectations and highexpectation principles, with the aim of preventing the detrimental effects of low expectations.

Dedication

I dedicate this work to my parents and my amazing supervisors, who supported me to make every stumble as a part of the dance.

Acknowledgments

Throughout my three-year journey, I have received numerous forms of support and assistance. Doing PhD research during the COVID-19 pandemic, I felt especially privileged to have been surrounded by a group of friends, family, and colleagues who experienced all the ups and downs with me (though we had to chat online most of the time and our relationships largely relied on internet quality). Therefore, this is an excellent chance to express my gratitude to those who contributed to this doctoral project.

To my parents: *Mum and Dad, I made it!* Thank you for your continuous love and support in whatever I pursue. It has been a tough time for all of us as we have been entirely separated throughout 3 years! Thank you for being caring, understanding, and encouraging whenever I felt frustrated or home sick. It was heart-breaking when I saw you crying on the screen, but it was even more touching when you tried to hold back your tears and encouraged me to stay strong and work harder. Much love.

To my marvellous supervisor, Professor Christine Rubie-Davies: *You are amazing!* Thank you for supporting me and being my substitute mum for 3 years. Your leadership, care, and good heart have shown through during this difficult and trying time. A massive thank to you for being my beacon in a time where everything was so uncertain. You are always there for others but rarely share your difficulties and vulnerabilities, which is respectful but also makes me want to care for you more.

To my co-supervisor, Associate Professor Mei Kuin, Lai: Your enthusiasm, support, and thoughtful feedback were truly appreciated. A massive thank you for your insightful comments on my writing, which pushed me to promote my thinking to a higher level, and for all of the opportunities I have been given to further my research skills.

To my partner, Yifei Pan: Thank you for being with me through the best and worst. The past 3 years have been an incredibly challenging time for us as we were living together in a foreign country and have stuck in long, long, long lockdowns. We have experienced exciting triumphs and literal disasters together. These experiences have indeed pushed the boundaries of what we thought we could endure. Thank you for pretending that you are interested in my research, but more importantly, for providing support to allow me to concentrate on my work. I am so lucky to have you as my driver, my courier, my comedian, but you are definitely not a good dancer, so please....

To my friends (too many to list here but you know who you are!): I could not have finished this thesis without you! Thank you for having stimulating discussions with me and providing interesting distractions to rest my mind outside of my studies. I would like to single out Georgi to thank you for being supportive throughout my time here and for everything you have done for me!

To the teachers and students who participated in this research: Thank you for your trust. I really appreciated your participation and your valuable views which helped build up this thesis.

Abstract	ii
Dedication	iii
Acknowledgments	iv
List of Tables	X
List of Figures	xii
List of Abbreviations	xiii
Chapter 1: Introduction	1
Chapter 2: Literature Review	4
2.1 A Profile of the Chinese Homeroom Teacher System	5
2.1.1 Main Commitments and Responsibilities of an HT	5
2.1.2 Appointment and Shortage of Chinese HTs	7
2.1.3 Chinese HTs' Reported Negative Emotions and Potential Influences	8
2.2 Theoretical Background	10
2.2.1 Theorising Teachers' Beliefs and Teacher Expectations	10
2.2.2 Conceptual Framework for Teacher Emotions	15
2.2.3 Theorising Students' Self-Concept	19
2.3 Relations Between Teachers' Beliefs, Their Emotions, and Students' Outcomes	21
2.3.1 Stage A: The Interactions Between Teachers' Beliefs with Teachers' Emoti	ons22
2.3.2 Stage B: Teachers' Emotional Process	
2.3.3 Stage C: Teachers' Differential Behaviour Related to Their Expectations an	ld
Emotions	29
2.3.4 Stage E: Students' Outcomes Associated with Teacher Beliefs	34
2.4 Literature Gaps and the Current Research	
2.4.1 Literature Gaps	
2.4.2 Current Research	
Chapter 3: Study 1 How Does it Feel to be a Homeroom Teacher? Relations Betw	veen
Teacher Emotions and Expectations	41
3.1 Method	41
3.1.1 Participants	41
3.1.2 Measures	42
3.1.3 Procedures	45
3.1.4 The Conceptual Model	46

Table of Contents

3.2 Data Analysis Plan	47
3.2.1 Analysis Plan of The Teacher Expectation Survey	47
3.2.2 Analysis Plan of The Teacher Emotion Questionnaire	48
3.3 Results	49
3.3.1 Identifying Teachers with Different Levels of Teacher Expectations	49
3.3.2 The Measurement Model on Teacher Emotions (Confirmatory Factor Analy	sis)54
3.3.3 Descriptive Statistics	62
3.3.4 Exploring Relations: A Structural Equation Model	63
3.4 Discussion	66
3.4.1 Understanding Emotions in Chinese HTs' Context	66
3.4.2 Understanding the Structural Relations Between HT's Emotions and TE	67
3.5 Limitations and Future Directions	69
3.6 Theoretical and Practical Implications	70
3.6.1 Theoretical Implications	70
3.6.2 Implications for Teacher Education	71
3.7 Conclusions	72
Chapter 4: Study 2 Expecting More: From Mind Full to Mindful	73
4.1 Method	73
4.1.1 Participants	73
4.1.2 Materials	75
4.1.3 Procedures	78
4.1.4 Data Analysis Plan	79
4.2 Results	81
4.2.1 Theme 1: Awareness of TE and Emotions	83
4.2.2 Theme 2: Perceptions of Teacher and Student Roles in Learning	92
4.2.3 Theme 3: Capacity for Resilience in Practices	95
4.3 Discussion	105
4.3.1 Answers to Research Question (1): In What Ways, If Any, Do HTs With Di	fferent
Levels of Teacher Expectations Show Different Patterns of Emotions?	106
4.3.2 Answers to Research Question (2): In What Ways, If Any, Do HTs With Di	fferent
Levels of Teacher Expectations and Corresponding Emotions Respond to	
Scenarios Differently?	111
4.4 Limitations	114
4.5 Implications for Future Research and Practice	114

4.6 Conclusions	116
Chapter 5: Study 3 Expecting the Best of Students: Teacher Expectations and S	tudent
Self-Concept	117
5.1 Method	117
5.1.1 Participants	117
5.1.2 Measures	119
5.1.3 Procedures	
5.1.4 The Conceptual Model	
5.1.5 Data Analysis	
5.2 Results	
5.2.1 Missing Data Analysis	
5.2.2 The Measurement Model on Student Self-Concept (CFA)	
5.2.3 Descriptive Statistics	
5.2.4 Exploring Relations: A Structural Equation Model	
5.3 Discussion	
5.3.1 The Statistically Significant and Positive Relations of TE with Student SC	
5.3.2 Variance in the Relations Between TE and Student SC	135
5.4 Limitations	137
5.5 Implications for Future Research and Practice	
5.6 Conclusion	140
Chapter 6: General Discussion	141
6.1 Overall Summary of the Key Findings	141
6.2 Overall Discussions of the Key Findings	142
6.2.1 Promising-Yet-Intricate Nexus	143
6.2.2 Contextual Interpretations	147
6.3 Theoretical Contributions	151
6.4 Practical Implications	152
6.4.1 Implications for Teacher Education and Professional Development	152
6.4.2 Implications for Stakeholders in Education	153
6.5 Limitations and Future Research Directions	155
6.6 Concluding Comments	156
References	157
Appendices	
Appendix A. Teacher Questionnaires	

Appendix B. Participant Information Sheets, Consent/Assert Forms	. 191
Appendix C. Student Self-Description Questionnaires	.214
Appendix D. Interview Protocol	.217

List of Tables

Table 3.1 All Scales and Sample Items for the Teacher Emotion Questionnaire	43
Table 3.2 Acceptable Levels of Fit Indices Regarding CFA	49
Table 3.3 Agglomeration Coefficients of Clustering the Average Residuals Using Ward	
Method (n = 135)	51
Table 3.4 Cluster Analysis Results of Teacher Expectation Groups Based on Average	
Residuals $(n = 135)$	52
Table 3.5 Mean Teacher-Level Teacher Standardised Residual Values (Relative to	
Achievement) (n = 135)	53
Table 3.6 Goodness-of-Fit Indices for Each Subscale in Teacher Emotion Questionnaire	55
Table 3.7 Teacher Emotion Subscales and Items Following the CFA Results	56
Table 3.8 Positive Emotions Subscales and Items Following the CFA Results	59
Table 3.9 Negative Emotions Subscales and Items Following the CFA Results	61
Table 3.10 Descriptive Statistics and McDonald's ω by Factor	62
Table 3.11 Pearson's Bivariate Correlations Among the Seven Emotion Factors and Teach	ner
Expectations	63
Table 4.1 Interviewees' Attributes	75
Table 4.2 Interview Questions	77
Table 4.3 Translation Procedure	79
Table 4.4 Cohen's Kappa Coefficient for the Two Transcripts Analysed for Interrater	
Reliability	81
Table 4.5 Codes and Quotations Relating to Their Emotions	85
Table 4.6 Number of Teacher Utterances Containing Codes Relating to Their Emotions	87
Table 4.7 Number of Teacher Utterances Containing Descriptors Relating to the Sources of	of
Negative Emotions	89
Table 4.8 Number of Teacher Utterances Containing Descriptors Related to Teachers'	
Priority or Foci in Teaching	101
Table 4.9 Number of Teacher Utterances Containing Codes Where Teachers Applied	
Adaptive Teaching Methods	102
Table 4.10 Number of Teacher Utterances Containing Descriptors Where Teachers Utilise	2
Their Social Networks	104
Table 4.11 Number of Teacher Utterances Containing Descriptors Where Teachers Utilise	2
Additional Materials to Support Their Work	105

Table 5.1 Demographics of Student Participants	119
Table 5.2 All Scales and Sample Items for the SC Questionnaire Measures	121
Table 5.3 Acceptable Levels of Fit Indices Regarding CFA	124
Table 5.4 Means and Standard Deviations for Variables for Different Sample Groups	125
Table 5.5 Goodness-of-Fit Indices for Each Subscale in Student SC Questionnaire	126
Table 5.6 Descriptive Statistics and McDonald's ω by Factor	127
Table 5.7 Pearson's Bivariate Correlations among the SC Factors and TE	128

List of Figures

Figure 2.1 HTs' Main Responsibilities and Commitments	
Figure 2.2 Overall Patterns of Teachers' Beliefs, Emotions, and Students' Outcomes	22
Figure 2.3 Appraisal Process	27
Figure 3.1 Time Nodes for The Teacher Expectations and Student Achievement Data	
Collection	44
Figure 3.2 Conceptual Models Showing the Hypothesised Relations Between the Varia	ables
Investigated in This Study	47
Figure 3.3 CFA Model of Joy Showing the Standardised Regression Weights for Each	Item
	55
Figure 3.4 Second-Order CFA Model for Positive Emotions	
Figure 3.5 Second-Order CFA Model for Negative Emotions	60
Figure 3.6 Structural Equation Model Predicting Paths Between Positive Emotions Va	riables
and Teacher Expectations	64
Figure 3.7 Structural Equation Model Predicting Paths Between Negative Emotions	
Variables and Teacher Expectations	65
Figure 3.8 Structural Equation Model Predicting Paths Between Both Emotions and Te	eacher
Expectations	66
Figure 4.1 Themes and Subthemes Developed Through Thematic Analysis of Interview	w Data
	83
Figure 4.2 Model of The Relations Between TE And Teachers' Emotions	107
Figure 5.1 Timeline Flow Diagram	118
Figure 5.2 Time Nodes for The TE and Student SC Data Collection	122
Figure 5.3 Conceptual Model Showing the Hypothesised Relations Between the Varia	bles
Investigated in This Study	123
Figure 5.4 Structural Equation Model Predicting Paths Between Nonacademic SC Var	iables
and TE	130
Figure 5.5 Structural Equation Model Predicting Paths Between Academic SC Variable	les and
ΤΕ	131
Figure 5.6 Structural Equation Model Predicting Paths Between Both Academic and	
Nonacademic SC Variables and TE	132
Figure 6.1 The Role of Teacher Expectations in Teaching and Student Outcomes	144

List of Abbreviations

CABI	Child Adaptive Behavior Inventory
CF	Consent Form
CFA	Confirmatory factor analysis
CFI	Comparative fit index
HiExTs	High-expectation teachers
HTs	Homeroom teachers
LoExTs	Low-expectation teachers
MeExTs	Medium-expectation teachers
PIS	Participation information sheets
RMSEA	Root mean square error of approximation
SC	Self-concept
SDQ	Self-Description Questionnaire
SEM	Structural equation modelling
TE	Teacher expectations
TEQ	Teacher Emotion Questionnaire
TES	Teacher Emotions Scales
TLI	Tucker-Lewis index
TOGA	Test of General Ability
UAHPEC	University of Auckland Human Participants Ethics Committee

Chapter 1: Introduction

This doctoral project was driven by personal as well as research interests in teacher beliefs, teacher emotions, and student self-concept development. At a personal level, as a former homeroom teacher in a Chinese high school, I perceived the pivotal role of teacher emotions and teacher expectations in teaching and learning, especially for homeroom teachers who share an intimate bond with their students and other stakeholders in schooling. However, as with most of my colleagues, I was overwhelmed with an excessive workload and the emotionally charged atmosphere related to this position. Compared to the ongoing professional development we received to improve pedagogical strategies and student academic achievement, teachers' negative emotions were more likely to be "an elephant in the room"—which is weighty but not remarked upon and mostly ignored.

On a research level, most existing explorations of these psychological factors (e.g., teaching beliefs, teacher emotions, and student self-concept) have mainly been conducted in Western contexts and at lower school levels, for example, early years in primary schools. To investigate how these factors interact in the context of Chinese high school homeroom teachers, I started this research journey in 2019 after I finished my master's research which explored the sources of homeroom teachers' professional vulnerability. Drawing upon the working journal and interview data from seven Chinese homeroom teachers, my master's study revealed the intricate relations between teacher emotions and their beliefs. These findings motivated me to further investigate how teacher expectations would predict teacher emotions, and subsequently result in different student outcomes.

As this thesis will discuss, previous literature has alluded to the relations between teacher beliefs and emotions (Frijda & Mesquita, 2000), both of which are important in influencing teaching practices and students' outcomes (Arens & Morin, 2016). As one construct of teaching beliefs, teacher expectations have been found to be influential in students' academic achievement and behaviours, nevertheless, little attention has been paid to its relations with emotions (S. Wang et al., 2018). Given the role of teacher expectations and emotions in education, it is of value to investigate how these two influential factors are related. In addition, teacher expectation effects have been largely explored in relation to student academic outcomes, with a limited number of studies attending to student self-concepts, despite their key role in student development (S. Wang et al., 2018). Therefore, this thesis particularly focused on teacher expectation effects on both teacher emotions (which has not been explored to my knowledge) and student self-concept to complement existing teacher

expectation research. Furthermore, compared to the vast number of studies on Western teacher expectations, teacher emotions, and student self-concept, insufficient efforts have been directed towards Chinese teachers and students. Specifically, little has been done to explore the experience of Chinese homeroom teachers, whose commitments are multidimensional and whose overall portrait, emerging from the literature, is of a stressful occupation. Considering that teacher expectations, teacher emotions, and student self-concept development might vary between cultural contexts, it is of interest to explore these factors further in the Chinese context.

It is for these reasons, along with my personal experiences, that I decided to explore teacher expectation effects not only in the high school student context but also in terms of teachers' own emotional experiences. In particular, this research explored whether and how teacher expectations might predict their emotions and student self-concept respectively. To achieve this, this doctoral project employed a mixed-methods explanatory research design which involved three studies.

Following this chapter (Introduction), Chapter 2 reviews the current literature on key factors explored in this project (teacher expectations, teacher emotions and student selfconcept), and how these three factors interact within the overall framework of teacher expectation effects. As will be elaborated in Chapter 2, previous studies on the significant role of teacher emotions in students' learning and achievement outcomes has been flourishing (H. Wang et al., 2019). By integrating the teacher expectation framework with emotion theories (e.g., Frenzel, 2014; Scherer, 2004), the literature review showed the fluid nature of emotions and the possible predictive role of personal emotional experiences. However, despite empirical evidence on the associations between teacher emotions with self-efficacy (Taxer & Frenzel, 2015) and teacher identities (Nichols et al., 2017), whether teacher emotions are related to teacher expectations is not yet understood. This constituted a gap that this project aimed to fill. Instead, existing teacher expectation literature has mainly focused on the predictive role of teacher expectations in instructional strategies and learning outcomes, and debates have been sparked regarding how teacher expectations are related to student self-concept (S. Wang et al., 2018). Given that student self-concept is a strong predictor of their outcomes and serves as a possible lever for further changes (Ogle et al., 2016), it is worthwhile to further explore teacher expectation effects on student self-concept in different contexts. These findings rationalised this doctoral research to investigate the relations of teacher expectations with their emotions and student self-concept, with the aim to contribute to social-psychological theory and help teachers perceive their role in handling

these factors in their teaching practices instead of being passive bearers. Chapter 2 concludes with the main research questions that are addressed in the three studies underpinning this research, which comprise Chapters 3, 4 and 5.

Chapter 3 presents the first study that drew upon questionnaires to explore the relations between teacher expectations and teachers' emotions. In particular, confirmatory factor analyses were used to test data-to-theory fit. A structural equation model was subsequently employed to test how teacher expectations predicted teachers' positive and negative emotions with a sample of 135 homeroom teachers from 14 Chinese high schools.

Following on from the first study, Chapter 4 presents the second study, which utilised and elaborated on the findings of Study 1 to enhance understanding of the way in which teachers perceived that their expectations were related to their emotional experiences with regard to their daily instructional practices. Using thematic analysis techniques, teachers' reactions to different scenarios and underlying explanations were explored. The results of Study 2 were also used to complement and explain the findings of Study 1 and the subsequently presented Study 3.

As presented in Chapter 5, Study 3 measured student self-concept at two time points from 348 students in the homeroom classes of earlier teacher participants. Confirmatory factor analyses and a structural equation model were used to investigate the relations between teacher expectations and different student self-concept constructs.

The final chapter, Chapter 6, discusses the findings from all three studies and examines why and how this doctoral research provides significant contributions and insights into areas of social-psychological research. A brief revisit of the three studies is presented in that chapter, followed by a general discussion of the findings. Then, the theoretical contributions, practical implications, the limitations of this thesis, and future research directions are presented.

Chapter 2: Literature Review

The current chapter rationalises the research aims and questions underpinning this doctoral project, by making reference to the existing and seminal body of literature about teacher expectations (TE), teacher emotions, and student self-concept (SC), and discusses the theoretical and empirical frameworks that were used in this project. In doing so, this thesis can then be contextualised as part of the existing research areas, as well as create links between what has been completed in the past, and what is being undertaken in the current project.

In particular, this doctoral project investigated three research areas in educational and social psychology that have been underexplored within the international literature and within the Chinese high school context. To be specific, the researcher aimed to explore the relations between three naturally occurring psychosocial latent constructs within the context of Chinese homeroom teachers (HTs) and high school students, namely, TE, teacher emotions, and student SC. Therefore, the following literature review will examine the current theoretical and empirical research related to these three constructs.

This literature review consists of four interrelated sections. Firstly, in order to provide international audiences with a better understanding of the contextual background, there is a brief profile of the Chinese HT system and the context of Chinese schooling where this doctoral project was located. This section also involves some recent research about Chinese HTs' emotional experiences. The second part of this chapter will then introduce different theories related to the concept of TE, their emotions and student SC, which aims to locate the study within the existing theories and specific context. The third, and major, part of this chapter will then focus on previous findings on the relations between teacher beliefs, emotions, and student outcomes. In presenting these studies, it will be shown that existing studies have alluded to the relations between these variables although little empirical research has been conducted to show how, and to what degree they are related, let alone taking account of Chinese HTs' perspectives. This constitutes one important rationale for this project because it is important to take the perspective of HTs into consideration, not only to contribute to the existing theoretical frameworks of teacher beliefs and emotions but also in relation to the educational system in both China and other contexts with parallel HT systems. To do so, the argument that TE, their emotions, and student SC might be related will be revisited and further discussed. This section is followed by a discussion of the limitations of the existing research, along with the scope of this doctoral project and the research questions.

2.1 A Profile of the Chinese Homeroom Teacher System

China has long adopted an HT system, where students study in the same homeroom class with their classmates (around 40–60 students) and typically have the same HT for several years (P. Liu et al., 2018). As shown in McNaughton and Li (2021), Chinese students are grouped into mixed-ability classes at the beginning of primary school (Grades 1–5), middle school (Grades 6–9), and senior high school (Grades 10–12). As a key role in the homeroom class structure, Chinese HTs' position has functional parallels in other countries (e.g., France and the United States) in which some staff (e.g., counsellors) are responsible for noninstructional duties in schooling (Z. Zhao, 2014). Nevertheless, as elaborated below, the commitments of Chinese HTs go well beyond what is expected from their counterparts in Western schooling (e.g., the US; P. Liu et al., 2018). To show what is specific and unique to the HTs' role in the Chinese context, the following sections begin with HTs' main commitments and appointment before moving on to their current dilemma related to their emotional experiences and professional development.

2.1.1 Main Commitments and Responsibilities of an HT

As an essential position in Chinese schooling, the job description of HTs is highly ambiguous but HTs are expected to play multiple roles from "instructor" to "counsellor" to "friend" (J.-Q. Liu, 1997; Z. Zhao, 2014). As suggested by G. Wang (1997), HTs generally serve as:

a bridge between the school, family, and society; a leader, organiser, and manager of one class who is the main teacher to carry out national educational policy, is an effective aid to school leaders who carries out educational and teaching plans in schools, and promotes all-round student development. (p. 87)

Compared to regular teachers who teach only one subject, HTs shoulder greater responsibility for students' behaviour, academic performance in all subjects, mental health and all-around development by having a close, day-to-day interaction with students in their homeroom class (S. Liu & Hallinger, 2018; Shi et al., 2014). There are no specific working hours for HTs, because they are expected to spend time with their students together on and beyond campus where they can carry out all their activities either in the class or based on the class (P. Liu et al., 2018). That is, what exactly constitutes HTs' commitments, how long they should work, and the degree to which they should be responsible for students is open to interpretation. Based on a literature search, this thesis largely classifies HTs' main commitments into three key areas (see Figure 2.1). It is worth noting that the tasks listed in

Figure 2.1 exemplify HTs' commitments, but their actual workload goes far more beyond what is indicated. Firstly, HTs have a function as subject teachers (e.g., subject teaching and marking) and take on the same administrative duties as other regular teachers, which is different to what might be expected of traditional counsellors in Western educational contexts. Secondly, HTs are responsible for students' overall academic achievement, which involves a duty to coordinate with all subject teachers and parents of their homeroom class. For example, HTs are required to arrive earlier than subject teachers every morning to ensure that all homework has been collected and sent to the respective offices on time. Once the examination results are released, HTs interpret each student's scores in all subjects and communicate with subject teachers and parents to find out how students could improve their learning. As such, compared to regular teachers who teach only one subject, HTs shoulder more responsibility for students' learning in all subjects by having more intensive teacherstudent and teacher-parents interactions (Shi et al., 2014). Finally, pastoral care as well as moral and character education constitute the third part of HTs' commitments (Lo, 2001). This aspect includes considerable responsibilities, such as addressing students' personal concerns and taking care of their mental health (S. Liu & Hallinger, 2018). After spending a long time together, the HT and the homeroom class students share an irreplaceably close relationship and make a difference together in the development of the class community (P. Liu et al., 2018). That is, Chinese HTs not only spend more time with their students in school than any other teachers, but also play a more comprehensive and important role in students' overall development than do subject teachers.

The commitments described above reveal that Chinese HTs are likely to have a deep connection with their students and play a critical role in supporting them alongside other daily operational responsibilities within the school. Also, HTs' commitments imply a substantial workload, particularly in Chinese public high schools where a homeroom class comprises around 40, 50, or even 60 students (P. Liu et al., 2018). In sum, given the crucial role of HTs in students' overall development, exploring the research questions through the lens of this group made it possible to investigate not only TE effects on students' academic SC but also on their nonacademic SC. Additionally, as indicated above and detailed below, the heavy workload and pressured emotions embedded in this position contribute to the importance of this research in exploring the relations of HTs' emotions with their TE.

HTs' Main Responsibilities and Commitments



2.1.2 Appointment and Shortage of Chinese HTs

Chinese HTs are selected from the regular teaching staff by principals. However, despite policies implemented to encourage teachers to take up this position (such as additional remuneration and opportunities for promotion), schools face the challenge of recruiting and retaining qualified HTs (Z. Zhao, 2014). The potential causes for these challenges have been underexplored in previous literature, despite frequent complaints about the imbalance between HTs' workloads and rewards (S. Liu & Onwuegbuzie, 2012). Nevertheless, whether the HT shortage can be totally attributed to structural factors (such as low salary and heavy workloads) remains a question, which calls for further exploration related to other influences such as their emotions and work experience. Whereas the topic of the source of HTs' negative experience is beyond the scope of this doctoral project, by evaluating the level of HTs expectations and emotions and exploring the patterns between these two factors, the current research could work as a foundation for future research providing practical strategies for HTs to manage their commitments and respond to their emotions positively.

Additionally, similar to the ambiguity in the job description for this position, the current criteria for cultivating and selecting HTs are also vague. As mentioned above, the

appointment of an HT is a subjective process in which Chinese HTs are selected from the regular teaching staff by principals. Even a first-year novice teacher can be invited to take on the role of HT and, from then on, HTs' training and their professional learning commonly relies on an in-service model (McNaughton & Li, 2021), which has been criticised as unsystematic and inefficient (Ye et al., 2021). Furthermore, whereas Chinese subject teachers are supposed to hold a national teacher certification in a particular domain (such as English for primary schools or mathematics for secondary schools; X. Han, 2012), no skill-specific education or qualification are asked of, or designed for, HTs (Sui, 2013). Many aspects of HTs' daily commitments described earlier (e.g., addressing students' mental health issues) require particular skills for which they have not been adequately trained (Shi et al., 2014; Yao et al., 2021). Concerns related to this situation have been recently raised in research. For example, based on semistructured interviews with 27 Chinese HTs, Yao et al. (2021) found that 15 HTs were struggling with dealing with students' mental health issues and 18 HTs reported insufficient or even nonexistent training. The interview data also indicated that whereas HT participants appeared to lack knowledge about psychological support for their students, all participants had tried to intervene when they perceived student mental health problems given their pastoral commitments. However, the tension between that responsibility and lack of knowledge seemed to cause increasing confusion in their professional life. The loophole in HTs' appointment and training risks not only compromising HTs' experiences but also increasing the public's mistrust towards this role. Therefore, taking these structural influences into account, it could be argued that it is more likely for Chinese HTs to experience negative emotions (e.g., pressure, confusion, and frustration) than for subject teachers. Additionally, the flaws in the structure mentioned above reveal that further explorations and refinements are necessary for the HT system in China, at least regarding appointment, preservice training, and in-service support. Although this research will not directly address these problems related to inadequate training and subjective selection of HT, it is hoped that the findings will act as a starting point for future explorations on improving the HT system. Furthermore, the current research could work as a foundation for future studies to provide practical strategies for HTs to manage their commitments and respond to their emotions in more effective ways.

2.1.3 Chinese HTs' Reported Negative Emotions and Potential Influences

In addition to the above concerns related to structural factors, indeed, Chinese HTs have already been reported as frequently experiencing emotional insecurity and pressures (X.

Gao, 2011; Huang & Gove, 2015). In particular, Chinese HTs' negative emotions have been found to be related to tangible factors, such as students' unfriendly attitudes (S. Liu & Onwuegbuzie, 2012), high job demands (Tsang & Kwong, 2017; Y. Wang et al., 2015), and complex interpersonal relationships (e.g., subject teachers, parents, and school leaders; X. Gao, 2011). These results will be echoed by some findings in Section 2.3, in which the literature on teacher emotions will be reviewed systematically.

Furthermore, one aspect unique to the Chinese context is that HTs are negatively influenced by contextual factors, especially cultural influences such as Confucianism and the hierarchical culture, as shown below. It is interesting to note that although the teacherreverence culture in China enhances teachers' professional authority (J. Chen, 2016), it also risks increasing Chinese teachers' negative emotions (X. Gao, 2011). Specifically, accompanied by respect for teachers, the Chinese public is empowered by the cultural tradition and there are continual demands on teachers to be morally and ethically above reproach (X. Gao, 2011). Shouldering these heavy cultural burdens, Chinese teachers might compromise their own experiences to fulfil these external requirements, which results in a relatively weak awareness of their autonomy and authority. For example, Chinese teachers have traditionally considered complaining about their workload or unfair treatment as a moral failure (Yin & Lee, 2012). Hence, Chinese teachers tend to hide their negative emotions, which could hamper their health and job satisfaction, as will be shown in Section 2.3.2. Moreover, Chinese teachers' negative experiences are further intensified by the Chinese hierarchical culture whereby power and authority were traditionally centralised at the top of the school hierarchy (Tsang & Kwong, 2017). Therefore, it is arguable that Chinese HTs' emotions are likely impacted by both the structural and cultural factors related to this position.

In sum, the above description sheds some light on how Chinese teachers—especially Chinese HTs—are embedded in their context and how they respond emotionally to external factors. These facts and empirical findings also serve as a robust complement for the discussions in Sections 2.2 and 2.3, for example, on how and why teachers deal with their emotions differently. Although the sources of Chinese HTs' negative emotions are beyond the scope of this project, the findings above led to a consideration related to whether and how these culturally induced emotions interact with TE. For example, it would be of value to explore whether being a high-expectation teacher (HiExT) helps Chinese teachers mediate extrinsic pressures or, conversely, intensifies their negative emotions (e.g., guilt, anxiety, or anger). Literature related to teachers' emotions will be reviewed in more detail in the following sections.

2.2 Theoretical Background

2.2.1 Theorising Teachers' Beliefs and Teacher Expectations

The association between TE and teachers' beliefs is controversial in the existing literature. Some researchers, such as Jussim et al. (1996), conceptualise TE as a significant component of teachers' beliefs, in parallel with other factors (e.g., teachers' efficacy beliefs), whereas others, such as Pugh (1974), perceive TE as an attitude, which is different from beliefs (V. Richardson, 1996). This research adopts the former view, assuming that TE is one component of teachers' beliefs. Following this stance, knowledge regarding teachers' beliefs is essential to provide a comprehensive background for understanding TE and their associated role in influencing instructional practices.

Teachers' Beliefs. Despite the flourishing research on teachers' beliefs over 50 years, there is no universal definition that researchers have agreed upon (e.g., Furinghetti & Pehkonen, 2002; Pajares, 1992). Much has been written in the literature related to the lack of cohesion and agreement in terms of the nature of teachers' beliefs, and concerns have been raised as this vagueness "has limited the explanatory and predictive potential of teachers' beliefs" (Fives & Buehl, 2012, p. 471). However, Fives and Buehl (2012) suggested that the difficulty was not in defining the term (because many authors have done so), rather, the difficulty was in having scholars define and use the same terms consistently when exploring these constructs. By exhaustively coding approximately 300 articles, Fives and Buehl managed to identify five characteristics of teachers' beliefs, including that 1) they are implicit and explicit, 2) they exist along a continuum of stability, 3) they are activated by context demands, 4) they are interwoven with knowledge, and 5) they are best understood as integrated systems. Based on Fives and Buehl's study, Levin (2014) further suggested that teachers hold various beliefs simultaneously, which range from their beliefs about knowledge, their subject matter, and pedagogy, to beliefs about their students, themselves, moral and ethical dilemmas, and societal issues related to their teaching. The inconsistency in how beliefs have been defined and used is beyond the scope of this research. Nevertheless, following the views of Levin (2014) and the characteristics proposed by Fives and Buehl (2012), it is reasonable to perceive TE as one of many teachers' beliefs based on the current definition of the term. In this regard, although the main focus of this project is TE, previous research related to teachers' beliefs will also be discussed and involved when it is appropriate.

Despite the above controversies, studies have yielded quite consistent conclusions that

beliefs are influential predictors of human behaviour (Bandura, 1986; Rimm-Kaufman & Sawyer, 2004; A. G. Thompson, 1992). Examined in the context of the teaching profession, teachers' perceptions of students' potential achievement affect their teaching directly, and so, if there are to be any changes in instructional practices, beliefs must first be addressed. For this reason, research that explores potential links between TE, their behaviours, and students' outcomes (SC in this research) is of particular relevance. Additional literature about the impacts of TE on teachers' instructional practices will be reviewed in Section 2.3.

Teacher Expectations. Compared to the mixed picture emerging from the definition of beliefs, TE, as one teaching belief, are widely conceived of as inferences made by teachers about their students' academic competence and subsequent performance (Good & Lavigne, 2017). TE have been found to be predicted by multiple variables, including students' previous achievement, their ethnicity, socioeconomic status, gender, and student diagnostic labelling and could exert influences in both teaching and learning (for a review, see S. Wang et al., 2018). As shown below, theories related to the types of TE and TE effects are reviewed respectively in this section. More details of the empirical evidence in these fields will be further elaborated on in Section 2.3.

Types of TE. Alongside the widely accepted definition of TE mentioned above, variations have been found in the TE measures used in different studies. In the review of H. M. Cooper and Tom (1984), they examined research on how TE influenced the schooling process and identified three general types of TE found in the literature: 1) estimate of current ability or achievement, 2) expected improvement, and 3) discrepancies between teachers' perceptions of student's present ability and the observed performance. In particular, the first type of TE concerns teachers' judgement of their student's present ability or performance, with no prediction of future outcomes. Given that this measure captures teachers' perceptions of students' ability at present only, H. M. Cooper and Tom doubted this type as a real expectation measure; however, they also suggested that this measure was somehow helpful in uncovering TE effects. An instance for measuring TE in this way is the study of Rubie-Davies et al. (2014). Operationalising TE as teachers' perceptions of children's academic capability, Rubie-Davies et al. used the "intelligent" scale of Child Adaptive Behavior Inventory (CABI; Cowan et al., 1995) as a measure of TE of their fourth-grade students' current performance.

Unlike this view, the second type of TE concerns expected changes in student ability, which is related to teachers' prediction of the possible academic progression their students would gain over a period of time, often over 1 academic year. For instance, at the beginning of the semester, teachers could estimate their students' performance for different subjects or

knowledge domains in 1 year, which is the measure used in the current thesis. This approach has been widely adopted in recent studies, for example, Z. Li and Rubie-Davies (2017) asked 50 teachers in two Chinese universities to predict the level they believed each of their students would achieve by the end of their first school year in English.

Different from these first two types of TE, the third type concerns teachers' inaccurate estimates (over- or underestimates) of their students' present achievement. This type of TE is measured by comparing teachers' estimates of student ability (the first type of TE explained above) with objective assessment outcomes (e.g., standardised test scores). H. M. Cooper and Tom (1984) questioned the accuracy of this approach given that standard measures of student ability might not be perfectly valid; nevertheless, the over- or underestimates are helpful in predicting the direction of change in children's future achievement. An example of this approach is Sorhagen (2013), who focused on teachers' inaccurate expectations and therefore measured TE by computing a discrepancy score between first-grade students' current performance and their teachers' perceptions of each individual child's academic performance (math abilities, basic reading abilities, and language skills).

As will be further elaborated on in Section 2.3.3, in addition to the variations mentioned above, researchers tend to classify TE into individual-level and class-level expectations. In terms of individual-level, there is a stance that teachers have differing TE for each individual student, whereas for class-level TE, high-/low-expectation teachers tend to hold high/low TE for all their students. However, it is worth noting here that the class-level perspective does not suggest that TE are equally high (or low) for all students in the class; instead, controlling for student prior achievement, high-expectation teachers (HiExTs) predict all their students will achieve substantial academic improvement whereas low-expectation teachers (LoExTs) anticipate limited progress (again controlling for student achievement at the beginning of the year). Brophy (1985) first indicated that class-level TE might have more significance for student outcomes than expectations for individuals and, indeed, this idea has been verified in some studies (e.g., Rubie-Davies, 2010). Therefore, it would appear to be of some consequence for this research to investigate class-level TE, rather than individual TE. This also seems appropriate given that an HT's class, as a whole, remains with them for a number of years within the Chinese context.

TE Effects. The way that TE exert an influence on students' learning process and outcomes cannot be understood without knowledge of the self-fulfilling prophecy and the Pygmalion experiment. The self-fulfilling prophecy was initially proposed and explored in the sociology realm, for example, Merton (1948) explained this sociological phenomenon

with an instance of an imaginary bank. Briefly, the story described how the Last National Bank, a flourishing institution, finally collapsed due to a rumour of insolvency which led clients to queue up to withdraw their money. Defining the self-fulfilling prophecy as the confirmation of an initially false expectation, Merton (1948) emphasised three crucial components underpinning this phenomenon: 1) beliefs about a situation, 2) behaviours led by the beliefs, and 3) confirming outcomes. As elaborated by Merton (1957), a self-fulfilling prophecy might happen when "a false definition of the situation evokes a new behaviour which makes the original false conception come true" (p. 423), which could exist in various social phenomena and areas, such as the economy and social inequity. According to H. M. Cooper and Tom (1984), teachers' beliefs about a student's potential improvement (the second type of TE mentioned above) and natural discrepancies between teachers' beliefs and observed scores (the third type of TE mentioned above) are the kinds of TE that might cause self-fulfilling prophecies. Nevertheless, the effect might not happen when a teacher accurately estimates the student's potential given that an accurate estimate does not contain any "initially false" component illustrated above.

After the establishment of the concept of the self-fulfilling prophecy, there were sparse empirical and systematic investigations in this field until Rosenthal's (1963, 1976) work on unconscious experimenter effects. Through a series of animal experiments, Rosenthal found that some laboratory rats would perform better if the researcher believed that the rats they were working with were smarter than others. Hence, the self-fulfilling prophecy was further illustrated in relation to experimenter bias in which the subjects' behaviours may have varied due to researchers' unconsciously and subtly different actions, which would in turn enhance the possibility for the researchers to verify their original hypotheses (Rosenthal, 1963, 1976).

Later, the idea of the self-fulfilling prophecy was introduced to the education field by the Pygmalion experiment of Rosenthal and Jacobson (1968), which raised attention and kindled enthusiasm in this field. Rosenthal and Jacobson induced some elementary school teachers to believe that the performance of certain students (experimental group students) would improve remarkably by the end of the school year compared with that of others (control group students). Nevertheless, there was no actual difference among these two student groups because they were randomly assigned. Therefore, teachers were manipulated to form false expectations for these students in their class. The result of the Pygmalion experiment showed that the experimental group students consequently achieved higher IQ test scores in the TOGA (Test of General Ability), a nonverbal intelligence test, over 2 years. It appeared that the student achievement gap could be attributed to the artificially differing TE because TE effects appeared to contribute to the confirmation of initially false TE. Another interesting finding from this experiment was that, following these artificially induced TE, teachers were found to be hostile to the unanticipated improvement made by the control group students when things ran against teachers' initial predictions. Consistent with the above three components (beliefs, behaviours, and outcomes) proposed by Merton (1948), Dusek et al. (1985) further explained the Pygmalion effect by showing the causal inferences: (a) the treatment resulted in higher TE (unmeasured) for the experimental group students than they would have achieved, which (b) mediated teacher behaviours (unmeasured), and in turn (c) led to better student outcomes.

The Pygmalion experiment contributed to the flourishing of the self-fulfilling prophecy in the field of educational psychology. Following the work of Rosenthal and Jacobson (1968), numerous replication studies were conducted under experimental settings (e.g., Carter, 1971; Evans & Rosenthal, 1969; Pellegrini & Hicks, 1972) and everyday situations (e.g., Babad et al., 1991; Babad et al., 1982) to explore TE effects. Nevertheless, these studies showed inconsistent findings and led to various interpretations about selffulfilment of TE on student outcomes, as well as the underlying mediators, and influential factors of TE effects (Z. Li, 2014; S. Wang et al., 2018). Agreement on the existence of selffulfilling prophecy effects of TE was partly reached when Rosenthal's series of meta-analysis and reviews were conducted (e.g., Ambady & Rosenthal, 1992; Rosenthal & Rubin, 1978). By examining the findings of 345 experiments on interpersonal expectancy effects, Rosenthal and Rubin (1978) found that interpersonal expectancy effects indeed existed by functioning as self-fulfilling prophecies. Rosenthal's finding was echoed by other reviews (e.g., Brophy, 1983; Raudenbush, 1984) which demonstrated the existence of TE effects that initial TE contributed to learners' confirmation of the teachers' perceptions and predictions. Furthermore, some scholars, such as Jussim and Eccles (1992), argued that studies in a real classroom setting over a long period were more likely to show stronger TE effects, which is the approach taken in this thesis. The substantial evidence in naturalistic settings since then further demonstrates TE effects, by showing that higher TE have significant associations with, if not impacts on, improved student psychosocial (e.g., academic motivation, selfefficacy, SC etc.; M. Zhu & Urhahne, 2015) and academic outcomes (e.g., Rubie-Davies et al., 2020). Additional empirical evidence will be presented in Section 2.3 on the relations between teacher beliefs, teacher emotions, and student outcomes.

2.2.2 Conceptual Framework for Teacher Emotions

Definition of Teacher Emotions. Similar to the ongoing debates on the definition of teachers' beliefs, there is no agreement on the definition of teacher emotions, considering they are highly elusive constructs (J. Chen, 2021). However, although it is challenging to define emotions scientifically and capture them empirically (Frenzel et al., 2021), some scholars have tried to establish conceptual frameworks for teacher emotions based on existing evidence. For example, the definition of emotions in the educational literature could be largely classified from four theoretical perspectives, which are psychological, social constructionist, and interactionist approaches summarised by Zembylas (2007), as well as the integrated perspective identified by J. Chen (2021) recently.

To be specific, the psychological perspective recognises emotions as primarily private, individual, and physiological, whereas the social constructionist perspective underscores the primary determinant role of context in emotions. In particular, the psychological perspective views emotions as individuals' internal feelings which are intrapsychic in essence (Parkinson, 2006). Put simply, the logic underlying psychological approaches is "I have feelings, and they are mine" (Zembylas, 2007, p. 59). Scholars following this view, for example, Pekrun et al. (2004), explored emotion in education as a personal experience, with a common assumption that emotions are first and foremost responses of those studied (Parkinson, 2006). In contrast, the social constructionist perspective considers emotions as essentially social experiences and, therefore, attends more to the sociocultural context where the meanings of emotions are shaped. Since the 1990s, many educational researchers have adopted social constructivist approaches to explore social and emotional relationships in teaching and learning (e.g., Hargreaves, 1998, 2000, 2001; van Veen & Lasky, 2005)

Transcending these two approaches, the interactionist perspective claims that it is overly simplistic to understand emotions in dichotomies (e.g., psychological and social), instead, emotions are essential to the processes where the psychological and the social are developed (Leavitt, 1996). The term "interactionist," as illustrated by Savage (2004), situates emotions beyond boundaries between psychodynamic and social constructivist approaches, and highlights the role of emotions in the formation of these boundaries which allow the subjects to interact. Furthermore, the interactionist perspective asserts that emotions are embodied and performative rather than merely language laden. In other words, how ones' sense of body. A performative view of emotion emphasises the role of the discursive actions and the materiality of the body in different kinds of actualisations of emotions. As such, the language itself can never be able to exclusively communicate emotions. Researchers influenced by this approach, for example, Holstein and Gubrium (2000), preferred ethnographic explorations which allow the voices of individual subjects and in-depth investigations of various fields of emotional interactions.

Finally, the integrated perspective defines emotions from various approaches (J. Chen, 2021). Drawing upon Vygotskian theory (Vygotsky & Cole, 1978) of the integrated perspective, J. Chen (2021) identified this approach from existing educational emotion research. By integrating the multiple approaches mentioned above, the integrated perspective regards emotions as 1) a part of a social-cognitive development process, 2) intrinsically associated with individuals' thoughts and behaviours, and 3) mediated by different contexts (e.g. Frenzel et al., 2016). In line with the majority of scholars in the emotion field (e.g., Frenzel et al., 2021), this thesis conceptualised teacher emotions as fitting this generalised definition. Following this view, I concur with Scherer's (2009) multicomponential conceptualisation, where emotions encompass cognitive, physiological, motivational, and expressive components. In this regard, an emotional episode does not simply refer to feelings, but is also accompanied by thoughts, bodily-physiological changes, action tendencies, and expressive behaviour. In explicating how emotions occur within HTs, this doctoral research is rooted in appraisal theory which is closely linked to the above definition of emotions (e.g., Frenzel, 2014; Scherer, 2004). Appraisal theory conceives that the "emotional process begins with some kind of judgement or appraisal that involves the interpretation of some transaction in terms of its significance or relevance for the individual's motives, goals, or concerns" (Sutton & Wheatley, 2003, p. 329). This proposition suggests that emotions are not merely motivated by an event itself, but rather by the individual's judgement of the event. In other words, various emotions could be elicited by the same event, which depends on the metaemotional process induced by personal judgement. Thus, appraisal theory, which highlights the role of individuals' judgement in the emotional process, provides some rationale for this doctoral research to explore the relations between TE (a kind of teacher belief) and teacher emotions. More details related to the appraisal process as antecedents of teacher emotions will be reviewed in Section 2.3.2.

The generalised definition of emotion and appraisal theory provides the theoretical background for the current research to explore how Chinese HTs expectations are associated with their emotions. Following Scherer's theory that emotions are related to personal interpretations of events, one could argue that the relations between TE and teacher emotions is dynamic and complex, for example, moving from ambivalent judgements of students' achievement at the start of the semester, to mixed emotions based on students' reactions and teachers' own reflections on their teaching practices. Meanwhile, the emotional appraisal process is suggested to be mediated by different variables, such as the person involved in the event, task characteristics, and the context (Efklides, 2006). This stance also explains why it was necessary to provide the background to the role of HTs and of the Chinese educational system in this thesis. As such, this doctoral research shifted attention from the individual perspective to a broader context where these HTs were situated.

Types of Teacher Emotions. A conceptual field worth mentioning related to emotions is how they are categorised and measured accordingly in educational research. There are two main classifications in existing emotion literature, namely, the state–trait continuum, and the dimensional–discrete entities.

Regarding the state-trait continuum, in particular, emotional states are short-lived and momentary experiences, whereas emotional traits refer to the habitual experiences of specific emotional states which are more ubiquitous and frequently occur. The trait view is more popular in quantitative research, where emotions are measured by self-report questionnaires, for example, the Teacher Emotions Scales (TES; Frenzel et al., 2016) and the Teacher Emotion Questionnaire (TEQ; Burić et al., 2018). This is also the approach adopted in Study 1 of this project. In contrast, the state view is widely adopted in qualitative research, where the researchers measure the emotions through daily diaries (e.g., de Ruiter et al., 2021), or interviews where teachers recall and describe certain emotional episodes with different stakeholders (e.g., Hargreaves, 2000), which is the approach adopted in Study 2 in this thesis.

Another major standpoint is to understand emotions as either dimensional or discrete entities. The dimensional perspective classifies emotions into unpleasant (negative) emotions and pleasant (positive) ones. Nevertheless, the discrete perspective emphasises the distinctions between each emotional component (e.g., happiness, anxiety, or boredom). The discrete approach underscores the importance of understanding these components separately because they are unique experiential states which suggest varying physiological and expressive responses and behavioural tendencies. Both dimensional and discrete emotion approaches are frequently adopted in teacher emotions research (Frenzel et al., 2021). As suggested by Frenzel et al. (2021), within the discrete approach, teacher enjoyment has been most frequently explored, whereas other discrete emotions, especially anxiety, have received less attention. This trend is somehow unexpected because anxiety is one of the most considered components in student emotion research (e.g. Pekrun et al., 2017). As explained by Frenzel et al. (2021), students' anxiety could be largely related to the assessments in schools, whereas teachers might be less concerned about their own failures given that their evaluations are less institutionalised. Therefore, the emotion of anxiety might not be that obvious among teachers compared to students, which may explain why it has received less attention in the existing teacher emotion literature. However, things can be different when exploring teacher emotions in some contexts with stricter evaluations in relation to standardised test scores (Frenzel et al., 2021), which is one of the reasons that this thesis evaluated teachers' anxiety as part of their emotions.

Effects of Teacher Emotions. As a critical role in both teaching and learning, teacher emotions have been found to be related to multiple outcomes. For example, J. Chen's (2021) review categorised these consequences into four themes: teachers, students, teaching, and learning. In addition, one conceptual framework was proposed by Frenzel et al. (2021), which was comprised of (a) *direct transmission effects* between teacher and student, (b) *mediated effects* on student outcomes, and (c) *the recursive effect* of student outcomes on teacher emotions.

In particular, on one hand, teacher emotions could exert an influence on student outcomes directly (teacher emotions \rightarrow student outcomes), or indirectly via teaching behaviours (teacher emotions \rightarrow teaching behaviours \rightarrow student outcomes). On the other hand, the recursive effects from student outcomes to teacher emotions were also found both directly (student outcomes \rightarrow teacher emotions) and indirectly, via teachers' appraisals (student outcomes \rightarrow [teachers' appraisals] \rightarrow teacher emotions) and teachers' differentiated behaviours (student outcomes \rightarrow [teaching behaviours] \rightarrow teacher emotions). Given that the overarching focus of the current project was TE and their relations with teacher emotions and student SC respectively, how teacher emotions exerted influences on student outcomes was not a particular focus of this thesis. However, among the effects of teacher emotions mentioned above, the recursive effects of student outcomes on teacher emotions are worth elaboration, because, as shown below, this process shares some similarities with the TE framework and might involve teachers' perceptions of specific students directly.

As illustrated by Frenzel et al. (2021) and also in their reciprocal model (Frenzel, 2014; Frenzel et al., 2020), teacher emotions are shaped by their appraisals of classroom conditions and student performance, which in turn could predict student outcomes. As such, students' individual or class-level characteristics (e.g., students' emotions and abilities), and their behaviours in the classroom might influence teachers' appraisals and perceptions of

them, which could further exert influences back on teachers (Frenzel et al., 2021). This mechanism is aligned with the patterns of TE effects introduced above. As shown in Section 2.3 and Figure 2.2, the current thesis was rooted in such a reciprocal conceptual framework and integrated it within the TE framework. Furthermore, another pattern worth mentioning in this model is the association between teacher emotions and their teaching behaviours, which is also underscored in recursive effects. Transcending the predictive role of teacher emotions on their instruction behaviours, Frenzel et al.'s (2021) model showed that teachers' instructional decisions and their subsequent behaviours might also feed back on their emotional experiences. To be specific, when teachers provide high-quality teaching practices, they are more likely to experience positive emotions (e.g., satisfaction), whereas negative emotions (e.g., boredom) might be aroused if teachers keep delivering low-quality instruction. As mentioned above and reviewed below, given that high TE are more likely to predict high-quality instructional behaviours (S. Wang et al., 2018), it is arguable that there might be some associations between TE and teacher emotions following Frenzel et al.'s (2021) model.

2.2.3 Theorising Students' Self-Concept

Different from the ongoing controversies over the definitions of teachers' beliefs and their emotions, there is a relatively clear picture in the context of students' SC (Mynott, 2018). This research adopts the definition of Shavelson et al. (1976), whereby SC refers to the way that individuals perceive themselves, which is shaped by experiences with surroundings and personal interpretations of the environment. According to Shavelson et al., the reinforcement by others and self-attributions of an individual's behaviours also contribute to the formation and the development of ones' SC, which could further explain and predict individuals' behaviours. Based on this definition, this doctoral research follows the Marsh/Shavelson model (Marsh & Shavelson, 1985), as will be shown below, to explore the relations of TE with students' academic and nonacademic SC.

Through the latter half of the 20th century, a domain-specific hierarchical SC structure (Shavelson et al., 1976) became popular instead of the early global perspective. The structure developed by Shavelson et al. (1976) involves an overarching global SC, below which are the academic and nonacademic levels (involving social, emotional, and physical aspects). These two levels break down further into more specific domains, for example, a subject-specific academic SC. However, this model has since been challenged by some subsequent studies. Drawing upon a confirmatory factor analysis (CFA), the study of Marsh

and Shavelson (1985) showed that verbal SC and mathematics SC could not be integrated to form a higher order academic SC (Marsh et al., 2006). Additionally, they also found that other components making up the SC could be highly differentiated. Therefore, a revised Marsh/Shavelson model was introduced in 1985 (Marsh & Craven, 1997), which comprised at least two second-order academic factors, the verbal-academic SC and maths-academic SC. This perception of SC as a multidimensional construct also resulted in Marsh and O'Neill's (1984) design of the Self-Description Questionnaire III (SDQ III), which was developed to measure various components of the SC specifically for the late adolescent population.

To establish the importance of exploring student SC along with TE, a brief summary of relevant research on student SC is provided here. As suggested by the Organisation for Economic Co-Operation and Development (OECD, 2003), SC is a critical construct in education as it "has important benefits for motivation and for the way in which students approach learning tasks" (p. 14). Combined with existing evidence on positive relations between SC and school achievement, quality of life, and mental health (Esnaola et al., 2020), investigating students' SC is of value. Specifically, in such studies where positive SC has been reported, higher levels of academic success and performance in subject-related activities (Susperreguy et al., 2018), effort and persistence (Marsh & Craven, 2006), approaches to learning (Mynott, 2018), and student engagement (Schnitzler et al., 2021) have also been reported. In contrast, in studies where lower levels of SC have been reported, students were more likely to experience greater declines in subject-learning interest (Denner et al., 2019) and at greater risk of mental health disorders (Mann et al., 2004).

Generally, most existing literature demonstrates relations between academic SC and academic achievement, but few studies have been found that focus on the nonacademic subdomains of the SC. To fill this gap, Ogle et al. (2016) examined longitudinal associations between nonacademic SC, academic SC, mental health symptoms, and academic functioning. Aligned with previous findings that children's SC in a particular area influences their performance in that area, more importantly, the study of Ogle et al. (2016) found that a change in the way a child perceived themselves in one domain led to subsequent changes in other domains. For example, if a student believed s/he could behave well, then s/he might begin to have faith that s/he can learn well. Their findings expanded on the hierarchical framework of Shavelson et al. (1976) and the one of Marsh/Shavelson (Marsh & Shavelson, 1985) by explaining that changes in one subdomain (e.g., social SC) could predict changes in another (e.g., verbal-academic SC). As such, the role of nonacademic SC in the development of a child could be further highlighted, which provides a rationale for this doctoral project to

explore both academic and nonacademic students' SC.

Together, the current evidence indicates that SC meaningfully promotes selfadjustment and works as a possible leverage for student development. Therefore, given the important links established above, it is somewhat unsurprising that several researchers have investigated ways in which student SC has been associated with other constructs of interest, especially its relations with teachers' psychological functioning (e.g., Arens & Morin, 2016). More details related to the predictors of student SC will be discussed in the following section. In addition to the key role of SC in students' development, this doctoral research pays special attention to both academic and nonacademic SC constructs given their close links to the particularity of HTs. As described previously, HTs play a crucial role in possibly contributing to students' overall SC by interacting with students more intensively and comprehensively than other teachers. Thus, there is reason to assume that the differences, if any, between any relations with students' academic and nonacademic SC will be more prominent when examined in the context of HTs than among regular teachers in the Chinese context. Moreover, given their unique roles, it is particularly important for HTs to realise how students' SC is related to teacher behaviours, especially inadvertent behaviours.

2.3 Relations Between Teachers' Beliefs, Their Emotions, and Students' Outcomes

The section above has presented the main and widely used theoretical frameworks referred to in the research fields of teacher beliefs, teacher emotions, and student SC. Following these theories, within the last 50 years, a number of studies have aimed to describe how teacher emotions and teachers' beliefs are respectively related to the process of learning and student outcomes (H. S. Kim, 2015). Recent studies of these concepts have moved away from investigating them in isolation, and have focused more on examining the combination of these factors alongside student cognitive and affective outcomes (e.g., Burić & Macuka, 2018). Some of these studies (e.g., Burić et al., 2017) have alluded to the interactions between some constructs of teacher beliefs (e.g., self-efficacy) and their emotions, with little regard for TE. However, given the importance of TE, teacher emotions, and student SC shown in this literature review, and their potential links, it is of value to explore how and to what degree these constructs are interrelated, which is one of the gaps that this doctoral project aims to address. Therefore, this section reviews the theoretical and empirical literature on the relations between these three fields, based on which the researcher has synthesised their relations into a six-staged flow diagram (see Figure 2.2). It is worth noting here that whereas these relations are theorised to be an inseparable process, only Stages A, B, C, and E

are relevant to this doctoral project. Hence, the following section will only review these pertinent components in detail.

Figure 2.2

Overall Patterns of Teachers' Beliefs, Emotions, and Students' Outcomes



2.3.1 Stage A: The Interactions Between Teachers' Beliefs with Teachers' Emotions

Previous research has suggested that the relations between emotions and beliefs are interconnected and interdependent (Frijda & Mesquita, 2000). As detailed below, the literature shows that, on the one hand, teachers' beliefs not only serve as a determinant of specific emotions but are also associated with the intensity and quality of the emotional responses. On the other hand, emotions have been found, in turn, to mediate beliefs and contribute to translating thoughts into action.

Predictive and Mediation Role of Teachers' Beliefs in Emotions. Existing literature has identified the predictive role of teachers' beliefs in certain emotions (J. Chen, 2021). Burić and Macuka (2018), for example, conducted a cross-lagged analysis with 941 teachers from Croatia and found that teachers with higher perceived self-efficacy experienced higher levels of positive emotions (joy, pride, and love), and lower levels of unpleasant emotions (anger, fatigue, and hopelessness), towards their students. Similarly, by exploring

teachers' experiences of reform-oriented practices, Cross and Hong (2009) suggested that teachers experienced negative emotions (such as stress or frustration) when their existing beliefs were not matched to what the reform required. Meanwhile, by identifying four general paths leading to pleasant or unpleasant emotions, Cross and Hong also stressed that induced emotions in turn worked as powerful predictors for confirming or challenging teachers' existing beliefs, as will be further discussed below.

Paralleling the predictive role of beliefs in emotions, teachers' beliefs might also intensify emotions that have already occurred. Based on 492 responses collected through teachers in the US Midwest, Chang (2013) claimed that teachers' intensity of emotions might increase when feeling less confident in solving a problem. Similar patterns have been shown to occur in the relations between desire, expectations, and emotional responses (Price & Barrell, 1984). Another example could be the work by de Ruiter et al. (2019, 2021), and de Ruiter et al. (2020), which showed interesting findings on the relations between teacher emotions and teachers' perceptions of students. In their studies, Dutch teachers of Grades 3-6were asked to recall and describe their emotional experiences of classroom events that were specific to individual students, by using either daily diaries (de Ruiter et al., 2021) or questionnaires (de Ruiter et al., 2020; de Ruiter et al., 2019). One important finding from their research was that teachers responded more emotionally negatively to students who were regarded initially as more disruptive compared to students regarded as less disruptive in the past (de Ruiter et al., 2020). Although the teacher perceptions of students measured in their studies were not exactly the same concept as the TE explored in this thesis, the findings that emerged from their work are illuminative for this research as possible relations between teacher emotions and their expectations of students are explored. In addition, teachers' selfefficacy, as an important teacher belief, has been found to be related to their emotions. Based on a survey with 1,187 preservice teachers from Canada, England, Hong Kong, and Thailand, Klassen et al. (2013) found that teachers' stress from student behaviour was largely mediated through their self-efficacy in three of four contexts, with no mediation effect observable in the Hong Kong sample. This result indicated a likelihood that teachers might alleviate their insecure emotions, or at least respond to emotions more effectively, when they had higher levels of self-efficacy. Additionally, Klassen et al.'s study also showed that teachers' beliefs and their interactions with emotions were influenced by contextual factors. This finding is particularly relevant to this doctoral research because, as a complement to existing conclusions in a Western context, the current project in a Chinese context was designed with consideration of shifting attention from the level of individual emotions to the contexts where
teachers are situated. However, most existing findings in this field (e.g., Klassen et al., 2013) are limited by the use of self-report data collected at one time point. It will be interesting and useful, for instance, to use longitudinal research designs to investigate the causal direction of the relations among these variables.

Although previous literature has alluded to teachers' beliefs being associated with teacher emotions, it is inconclusive as to how and to what degree. This could be partly because teachers' beliefs and teacher emotions are both dynamic, situated, and contextual, which has resulted in intrinsic and complex relations (Barcelos & Ruohotie-Lyhty, 2018). A recent study by Chang (2020) transcended previous findings on the direct relations between teachers' beliefs and their emotions; instead, she tried to test a hypothesis on the underlying mechanism between teachers' beliefs about emotional display rules, emotional regulatory approaches (e.g., cognitive reappraisals), and teachers' burnout. Based on survey data from 561 full-time teachers from a Midwestern state in the US, Chang confirmed the relations between these three factors (teachers' beliefs, emotional regulation, and emotional outcomes) using structural equation modelling (SEM). The results showed that teachers' beliefs about the display rules were highly related to teachers' suppression of their emotions in the classroom, which in turn mediated their emotional experiences and may have exerted detrimental effects on teachers' well-being. Chang's research provided insightful understanding of the relations between teachers' beliefs and emotions by showing the underlying mechanism for teacher's emotion regulatory approaches being shaped by their beliefs in the classroom. In particular, Chang found that teachers' beliefs (beliefs about classroom norms and culture, and self-perceptions as teachers) mediated their reactions to the emotions they experienced in their daily encounters with students, which further mediated their burnout as an outcome. More details related to emotional regulatory strategies and corresponding outcomes will be further reviewed in the following sections.

Predictive and Mediation Role of Teacher Emotions in Teachers' Beliefs. Bearing in mind the aforementioned predictive and mediation role of teacher beliefs in their emotions, emotions have been found, in turn, to predict beliefs in two ways: (1) beliefs that were previously nonexistent could be generated, or (2) existing beliefs might be changed. These modifications may also involve a mediation or increase in the strength of the belief (Frijda & Mesquita, 2000). Despite Frijda et al.'s (2000) concerns over the scarcity of empirical research confirming the emotions-beliefs effects, several relevant studies have been identified recently. For example, the interviews conducted by Timoštšuk and Ugaste (2012) demonstrated the influence of emotions on the development of the professional identity of student teachers. By interviewing 45 student teachers in Tallinn University, the researchers found the critical role of emotions in their teaching experiences, which further related to the development of the student teachers' professional identity. For instance, some physics teachers suggested that they felt confident and happy when they received positive reactions from others, which contributed to their self-perceptions as a teacher. In addition, the interview data also revealed that the influences associated with their negative emotions appeared to be more intensive than that related to positive emotions, which underscored the importance of teacher education to deal with teachers' negative experiences. In addition, the case study conducted by Zembylas (2005) with a teacher participant (Catherine Myers) helped illustrate how specific emotions shaped teachers' beliefs in more depth. For example, Catherine described how she experienced devastating emotions when she had goal conflicts between school rules (teaching-to-the-test) and her own teaching philosophy and values (drill and practice), and how these negative emotions in turn compromised her SC. As Catherine described, "During those days, I felt like crying all the time, feeling guilty of what I was doing. I thought I was a lousy teacher for not doing what everyone else was doing. I constantly questioned my approach ... and I still do" (p. 475). As shown in Catherine's descriptions, the feeling of being alienated and dismissed by others destroyed her selfconfidence and self-esteem and further threatened her own pedagogical beliefs. These findings are in line with those of Frijda et al. (2000) who proposed that beliefs can be awoken, interrupted, and shaped by emotions, during which the emotions might amplify, alter, or make the beliefs resistant to change.

Furthermore, in addition to the interrelations between teacher emotions and beliefs, philosophers (e.g., Frijda et al., 2000) have indicated that emotions are also integral to the manifestation of translating thoughts into action. In other words, whereas beliefs are precursors or guides for individuals' actions, without emotions, these actions would never be actualised (Frijda & Mesquita, 2000). Therefore, given that TE and teacher emotions are interrelated and are each associated with instructional practices, further exploration of them in combination is warranted, especially since this has not been empirically studied in previous literature (Frijda & Mesquita, 2000). This stance constitutes an important reason for this doctoral research to explore TE and teacher emotions in combination and then approach teachers' perspectives on corresponding patterns and how such interactions are related to their teaching practices.

Overall, despite a substantial body of literature exploring how teacher beliefs interact with emotions, most previous research has studied teachers' beliefs as a whole, with little attention paid specifically to TE. However, controversies remain over the interrelations between different constructs of teacher beliefs (Rubie-Davies et al., 2012), which suggests that different components might have various patterns. Thus, the findings on the relations of general beliefs and emotions cannot be simply transferred to TE. However, previous preliminary findings or speculations on the relations between teachers' beliefs and teachers' emotions provide some rationale for the doctoral project to explore this field in a more specific way.

2.3.2 Stage B: Teachers' Emotional Process

Appraisal Processes as Antecedents of Emotions. Following the view that emotions are elicited by evaluations (or appraisals) of events or situations depending on individuals' beliefs (Lazarus, 1991; Scherer, 2001), this doctoral research builds upon Lazarus's (1991) appraisal theory to explore teachers' emotional processes. This approach has been shown earlier to be helpful in analysing teachers' emotions (e.g., Howard & Gigliotti, 2016) by allowing the researchers to capture differences that characterise positive or negative emotional responses (C. A. Smith & Kirby, 2009).

Specifically, as shown in Figure 2.3, Lazarus (1991) divided the factors triggering and mediating emotions into two classes, namely, primary appraisal and secondary appraisal. To be more particular, primary appraisals "concern the stakes one has in the outcomes of an encounter" (p. 827). Primary appraisals involve evaluations of 1) goal relevance or importance, which concerns the importance or strength of the goal; and 2) goal congruence or incongruence, which relates to whether the encounter is considered favourable or not. Hence, primary appraisals are perceived to be the impetus in emotional activities (Lazarus, 2001), influencing how emotions are triggered. Secondary appraisals involve the coping options and prospects of the situation, which concerns agency for the event, coping potential, and future expectations (Lazarus, 1991). For instance, when faced with students' low achievement, teachers who believe that students themselves should take responsibility for their learning may feel angry, whereas teachers who attribute the failure to themselves may feel guilty or anxious. As reflected in Lazarus' theory of appraisal, both primary and secondary appraisal are closely related to personal beliefs. It is interesting to transfer this theory to the field of TE where a similar trend has been found: some teachers appear to believe that little can be done to improve the learning of low-ability students and seem to shrink from their

responsibility for student learning (Diamond et al., 2004; Rubie-Davies et al., 2006) and blame the students when they do not achieve well. Nevertheless, other teachers take personal responsibility for student learning and examine their own pedagogy when low-achieving students are making little academic progress (Diamond et al., 2004). Based on this stance, this research inquired into whether teacher emotions were different for teachers with high TE from those with low TE. In addition to the potential relations between teachers' emotions and their TE, this assumption also indicates a possibility that Chinese HTs' emotions, which have been typically described in the literature as insecure and negative, might be improved through an intervention related to their TE. Hence, it is valuable to first investigate whether there are any links between Chinese HTs' emotions and their TE. By showing the relations between these two factors, this exploratory research might provide future researchers with ideas for practical strategies to help raise TE and enhance their positive emotions.





Note. This figure is designed based on the theory of Lazarus (1991)

Emotion Regulation as Mediating Factors. In addition to the above appraisal processes, the other part of this stage (Stage B: emotional process) is emotion regulation, where a person determines which emotions to allow or contain and when, and how, to express them (Gross, 1998). Based on a literature review of teacher emotion regulation, Chang (2013) concluded that individuals might regulate their emotions in one of two ways,

namely, cognitive reappraisal and expressive suppression (Gross & John, 2003).

Cognitive reappraisal is the process where individuals rethink, reevaluate and reinterpret an emotion-eliciting situation to alleviate negative emotions, while increasing pleasant emotional experiences. For example, when a disruptive student messes up the classroom, a teacher might initially be angry; however, she can cognitively reappraise the emotional stimulus by thinking that the classroom was already quite dirty and in need of a good clean. In doing so, the teacher might mediate the destructive emotions that ensue from dealing with student misbehaviour. There is consistent evidence that the use of cognitive reinterpretation is related to many positive outcomes, for example, less stress (Yamasaki et al., 2006), less social anxiety and depression (Dryman & Heimberg, 2018), lower levels of emotional exhaustion (Donker et al., 2020), and improved mental health (T. Hu et al., 2014). As previously reviewed in Stage A (Section 2.3.1), Chang (2020) conducted a study exploring the relations between teachers' beliefs about emotional display rules, emotional regulatory approaches (e.g., cognitive reappraisals), and teachers' burnout. In addition to the relations between teachers' beliefs and their emotional regulation, the model also confirmed that the cognitive reappraisals predicted lower teacher burnout in all three dimensions (emotional exhaustion, depersonalisation, and low personal accomplishment). The opposite pattern was observed related to expressive suppression which increased teachers' burnout in all three dimensions, as shown below. The findings showing the relations between emotional strategies and outcomes indicate that emotions, especially negative ones, are not simply a condition to be felt or endured, rather, they imply a likelihood to be mediated by personal beliefs and interventions. From this perspective, it is important to be aware that whereas many inevitable encounters might trigger HTs' negative emotions, it is possible for individuals to regulate their emotions effectively and minimise corresponding negative results. Following this argument, the findings emerging from this doctoral research could be important for future studies on suggestions for Chinese teachers' responses to their emotions.

Expressive suppression, however, influences individual's behaviours by "shuttingdown" emotions which might threaten one's emotional equilibrium (Gross, 1998). In this regard, individuals might hide their true emotions to prevent negative outcomes and increase the possibilities of desired outcomes. Expressive suppression has been found to predict more teacher burnout, including emotional exhaustion and depersonalisation, as well as diminished personal accomplishment (Chang, 2020), and to hamper teachers' well-being (J. Han et al., 2020). However, there is an exception related to the expression of anger, which has been found to predict poorer social outcomes and physical indices among Americans, although this association was not evident for Japanese teachers (Kitayama et al., 2015). By examining 48 studies which comprised 21,150 participants, the meta-analysis of T. Hu et al. (2014) suggested that improved mental health was positively and significantly predicted by cognitive reappraisal, whereas expressive suppression was related to decreased mental health. More importantly, the researchers confirmed the moderator role of culture in this relation. In particular, the association between expressive suppression and negative mental health has been found to be stronger in Western culture than that for Eastern contexts. Echoed by the findings in Kitayama et al. (2015), T. Hu et al. (2014) emphasised that the associations between emotional regulations and outcomes might vary across contexts and thus further investigation was warranted. Following this view, it is interesting to explore teacher emotions further in the Chinese context, where teachers are more inclined to accumulate or hide unpleasant emotions than express them genuinely (Yin & Lee, 2012) due to high public expectations (T.-J. Wu et al., 2020), the hierarchical system, and Confucianism (Lee et al., 2013). Combining the promising outcomes found in cognitive reappraisal, exploring whether TE could mediate teachers' emotions and in what ways, is of special value for Chinese HTs. Bearing this in mind, this doctoral project functions as an initial step by exploring the relations between these two factors so future researchers may use the findings to develop practical strategies to help raise teachers' expectations and enhance their positive emotions.

2.3.3 Stage C: Teachers' Differential Behaviour Related to Their Expectations and Emotions

As mentioned in Stages A and B, TE and teacher emotions are likely to be interdependent, both of which are integrated in a cyclical emotion generation sequence. Subsequently, in Stage C, the combination of teacher emotions and their expectations is proposed to be related to different behaviours of teachers in class. This section reviews the literature on the associations of teaching practices with TE and teacher emotions.

Teachers' Differential Behaviour Related to TE. Many rigorous studies have verified that TE can be manifested and transmitted to students through teachers' different verbal and nonverbal behaviours during instruction (e.g., Babad & Taylor, 1992). The work of Brophy and Good (e.g., Brophy & Good, 1970) indicated that teachers' varying behaviours towards different students worked as the mediating mechanism for TE effects, in which TE cues were communicated to students both directly and subtly. Later, Brophy (1983) summarised 17 teacher–student interaction behaviours based on relevant studies, for example, teachers might wait less time for low-expectation students to answer and praise them less frequently compared to high-expectation students. Furthermore, a useful four-factor theory was proposed by Rosenthal (1974) which identified four ways that teachers could communicate their TE to students, namely, climate (socioemotional climate in classroom), feedback (teachers' praise, criticism, reward, and punishment for students), input (the time and attentions that teachers give to students, and teaching materials used), and output (students' opportunities provided by teachers to respond, ask, and answer questions). This typology was further supported by Harris and Rosenthal's (1985) meta-analysis of 136 investigations. Harris and Rosenthal's analysis highlighted the crucial role of distal behaviours in TE effects, especially the socioemotional environment created by teachers in their classroom. The teachers' behaviours and corresponding influences mentioned in Harris and Rosenthal's research were arguably at the class level, for instance, the class climate shared by all students, rather than teachers' behaviours towards certain students.

Combining Harris and Rosenthal's study and the TE classifications that teachers not only develop TE for individual students but also for their classes as a whole (Brophy, 1983), it appeared that class-level TE were worthy of exploration. Nevertheless, as mentioned earlier, most existing research has explored TE only from the perspective of individual-level TE, with relatively limited attention to class-level TE. In recent years, however, Rubie-Davies and her colleagues have initiated a series of studies investigating class-level TE and their underlying working mechanisms of functioning as self-fulfilling prophecies or influencing student achievement (Rubie-Davies, 2006, 2007; Rubie-Davies et al., 2014). Her work suggested that different levels of class-level TE generally led to differing teachers' behaviours in their teaching and interactions with students, which resulted in varying instructional and psychosocial environments in classrooms (Rubie-Davies, 2007). In one study, Rubie-Davies (2008) interviewed teachers with different levels of TE and found two distinct patterns in their instructional beliefs and self-reported behaviours, which complemented and confirmed previous findings from classroom observations (Rubie-Davies, 2007). In particular, it seemed that HiExTs provided a more careful learning framework for their students, including organising students into mixed-ability groups, enhancing students' autonomy, providing careful explanations of new concepts and clear feedback, reacting to student behaviour positively, and asking mostly open questions, whereas LoExTs often behaved in a directly opposing fashion. As the first work to explore the communication of TE to the class as a whole, rather than behaviours towards individual students, Rubie-Davies' research has contributed to TE research by showing more generalised TE effects. Additionally, Rubie-Davies' work advanced TE research by best elaborating Rosenthal's (1974) four factors (Z.

Li, 2014). However, this work was conducted at the elementary and middle school level and left unanswered questions about whether the findings would generalise to higher levels of schooling. Nevertheless, evidence since then has been consistent; for example, HiExTs have been shown to perform more proactively than LoExTs in providing oral feedback, and in engaging and responding to students (e.g., S. Wang et al., 2019), which in turn has been shown to predict larger academic improvement for students (e.g., Szumski & Karwowski, 2019).

However, despite a long history of TE research in Western academia, there are a much smaller number of studies in the Chinese context. Regarding individual-level TE research in the Chinese context, Jia's (2012) case study, for example, investigated the associations of TE with teacher-student interactions by conducting classroom observations of four English teachers based on the revised Flanders' Interaction Analysis Categories. By calculating the amount of time that each teacher spent with individual students regarding 12 classroom verbal behaviours, Jia found that teachers' behaviours and reactions to students were related to student gender, achievement level, and class leader status. Briefly, girls were provided with more interaction opportunities than boys and class representatives received more emotional support than other students. Additionally, teachers interacted with high achievers the most, followed by low achievers, whereas average achievers were somehow neglected. Regarding the contents of the interactions, high achievers were mostly praised and received referential questions which required higher order thinking, whereas low achievers were mostly criticised and asked questions that usually required "Yes" or "No" answers. The pattern that emerged from Jia's study was consistent with that of previous researchers (e.g., Babad et al., 1989a, 1989b; Brophy, 1985) who found that teachers might compensate lowexpectation students by spending more time and providing more instructions for them due to teachers' increasing awareness of their varying reactions towards specific students. Nevertheless, compared to the increased quantity of interactions, the quality of the teacherstudent interactions remained unchanged (Babad, 1998). For example, teachers tended to show more positive facial expressions and body language (e.g., smiling more) towards highexpectation students than to low-expectation students (Babad et al., 1989b).

In addition to individual-level TE research in the Chinese context, several recent studies exploring class-level TE research have been identified. For example, Z. Li (2014) explored both perceptions of teachers and students of the classroom climate in two Chinese universities. The results of teacher interviews and student focus groups indicated that HiExTs developed a more caring, personal relationship with their students, allowed more student autonomy, and promoted more cooperation between students compared to LoExTs. A similar trend was found in the study of S. Wang et al. (2019) by observing 32 lessons of eight junior high school teachers. Although these results are not robust enough to argue that TE effects in the Chinese context share similar patterns with earlier results in Western countries, the existing studies indeed shed light on TE effects in Chinese classrooms and provide a starting point for further exploration. However, the existing Chinese studies exploring the TE phenomenon are mainly focused on teachers in university (e.g., Z. Li & Rubie-Davies, 2017, 2018) and junior high school (e.g., Ding & Rubie-Davies, 2019; S. Wang et al., 2021; S. Wang et al., 2019, 2020). As shown above, there is limited evidence of TE effects in the context of the Chinese high school, despite its key role in students' transition from fundamental education to the tertiary level. Additionally, with intensifying academic competition for university places (X. Zhao et al., 2015), the Chinese teachers' role in influencing students' future success has been especially emphasised in high schools. Teachers are therefore subject to ever-closer public scrutiny and more external demands (X. Gao, 2008), which may impact how they experience dilemmas emerging from the relations between social change and cultural tradition (Fong, 2004; Huang & Gove, 2015). Taking these considerations together, this doctoral project paid special attention to the high school context.

Generally, teachers' differing TE of students can be reflected in their differentiated behaviours which in turn mediate the TE effects by generating the communication of TE cues. Based on individual- and class-level TE, relevant variances are found in teachers' instruction, the learning opportunities and materials they provide, and the socioemotional climate they create in the class. Given that students directly participate in these activities and are immersed in the class climate, how they perceive and react to these TE cues could be the next key part of the mediation chain. Studies have shown that students are sensitive to and capable of perceiving and interpreting teachers' beliefs and emotions beyond teachers' behaviours (Rubie-Davies, 2006), which can further predict students' academic and nonacademic development (Jennings & Greenberg, 2009). As shown in Stage D in Figure 2.2 and explained below, only when students are aware of and react to their teachers' expectations for them, can the influence of TE on student learning and achievement outcomes be finally exerted and realised.

Teachers' Differential Behaviour Related to Their Emotions. According to a recent review by J. Chen (2021), teacher emotions have been identified as a crucial factor in

varying teaching practices and teacher–student interactions. By examining 812 articles from 1985 to 2019, J. Chen identified two major teaching realms that were associated with teacher emotions, namely, teaching methods and classroom management. Briefly, positive emotions were found to predict student-centred teaching approaches and management strategies, whereas negative emotions were related to teacher-centred styles.

In particular, studies with teachers from elementary education to tertiary settings, from various contexts (e.g., Germany, China, and Australia), have shown consistently that teachers with positive emotions interact with their students more effectively and perform more proactively than those with negative emotions (e.g., Arens & Morin, 2016; J. Chen, 2019; Trigwell, 2012). In contrast, more emotionally exhausted teachers have been reported to have lower quality interactions with children (Ansari et al., 2020), less satisfying teaching practices (Arens & Morin, 2016), lower levels of emotional support, and poorer classroom organisation (S. S. Braun et al., 2019). For example, based on the self-report questionnaires of 175 Australian higher education teachers, Trigwell (2012) suggested that teachers with positive emotions tended to choose student-focused teaching approaches whereas teachers with negative emotions would favour transmission approaches. Using similar methods, J. Chen (2019) identified consistent patterns in 1,830 Chinese primary teachers. These patterns appeared to have things in common with the TE effects mentioned above.

In addition to these studies relying on self-report questionnaires, there is also confirming qualitative evidence which has illustrated the reasons underlying why teaching practices related to teacher emotions in certain ways. For example, Saunders (2013) collected interview data from 27 teachers who participated in a 4-year professional development programme. The programme was designed to refine and extend teachers' instructional practice. When examined longitudinally and qualitatively, a generalised and more in-depth pattern of why teacher emotions predicted specific behaviours emerged from the data. For example, as explained by one participant, negative emotions (anxiety and insecurity) prevented her from trying new approaches and instead she adhered to her original practices. As explained by Saunders, influenced by negative emotional experiences, teachers were more likely to implement things in a safe but ineffective way to avoid any potential risks. Similar explanations can be found in recent work by Frenzel et al. (2021), which indicated that negative emotions might stimulate shallow and narrow information processing and immediate actions. Consequently, teachers with negative emotions might be more rigid and less prepared to adopt novel and complex instructional strategies.

Despite these consistent findings, one more thing worth noting here is that it is overly

simplistic to classify the potential effects of teachers' emotions into a binary mode of "good" and "bad" (Frenzel et al., 2021). Again, as mentioned above, teacher emotions should be perceived as a condition that could be nudged, and how teachers respond to their emotions can also be mediated by contextual and personal factors, such as their experiences and teaching beliefs. Mevarech and Maskit (2015), for instance, provided evidence for this argument. Their survey data from 180 Israeli pre- and in-service teachers showed that experienced teachers might navigate their negative emotions more effectively than novice teachers and led to different coping strategies. It seemed that although teachers' behaviours were related to their emotions, how the emotions were finally actualised could vary from individuals. It is therefore important for further explorations to identify the specific factors contributing to teachers' emotional resilience, which will help improve not only teachers' own well-being but also teacher education and professional learning development.

In sum, there are promising findings showing the relation of teachers' behaviours to TE and teacher emotions. The patterns which seemed to share some similarities formed a rationale for this doctoral project to explore these fields further in association.

2.3.4 Stage E: Students' Outcomes Associated with Teacher Beliefs

As mentioned above, teachers' beliefs and emotions have been found to interact with each other (Stage A and Stage B) and predict their behaviours (Stage C). Furthermore, as shown in Stage D in the flow diagram above (Figure 2.2), earlier examinations have revealed that students' perceptions and reactions to teachers' differential behaviours is an integral part of how teachers influence students. Specifically, students are aware of teachers' differing behaviours toward low- or high achievers from a very young age (e.g., Babad & Taylor, 1992). Given that students' perceptions of TE and emotions are beyond the scope of this research, literature related to Stage D is excluded here. This is not to say that this stage is unimportant, rather, it is essential for the effects that follow. Bearing this in mind, this part will move on to the next stage (Stage E) which concerns the role of teacher emotions and teachers' beliefs in student outcomes. In studies where student outcomes have been examined across various educational levels, statistically significant relations have been found and attributed to a number of factors including teachers' beliefs and their emotions. Overall, these two factors are influential in students' development, including their academic achievement (e.g., Agirdag et al., 2013; H. S. Kim, 2015), psychosocial outcomes (e.g., Pesu et al., 2016), and behaviours (e.g., Demanet & Van Houtte, 2012). However, since this doctoral project specifically focused on the relation of students' SC to their HT's TE, only findings related to

this field will be reviewed here. As discussed in earlier sections, it is important to investigate student SC given the existing literature that has established strong ties between that construct and student outcomes during and beyond the schooling years.

Compared to TE effects on students' achievement, how TE influence students' SC has been somewhat neglected. As suggested in S. Wang et al.'s (2018) recent review, only four empirical studies were identified in the field related to TE effects on student SC. Considering the important function of SC in students' further development, this doctoral project particularly focused on the relations between TE and student SC outcomes. Additionally, most of the current studies on TE effects have attended to academic SC (e.g., Pesu et al., 2016; Upadyaya & Eccles, 2015), with little regard to nonacademic SC. Given the important role of nonacademic SC in students' development, such as their social identification (Tarrant et al., 2006), this doctoral research aimed to fill this gap by tracking both academic and nonacademic SC of students whose HTs had different levels of TE. Furthermore, since relations of TE to different components of SC might vary, as shown in the theoretical background in this thesis, this doctoral research explored how, if at all, TE predicted students' academic and nonacademic SC differently.

As described in the theoretical background (Section 2.2), how students' SC is associated with TE cannot be discussed separately from the self-fulfilling prophecy, which became prominent in the education context through the experimental study of Rosenthal and Jacobson (1968). According to the self-fulfilling prophecy, students tend to develop in line with TE (Rosenthal, 1991) regarding their intelligence, achievement, and motivation (Jussim, 1989; Jussim & Harber, 2005). For example, students expected to be high performers have greater improvement in their SC of ability than expected low performers (Jussim, 1989). Following the theory of the self-fulfilling prophecy, some studies have found that TE influence the formation and alteration of student SC (M. Zhu & Urhahne, 2015). However, as the examples below show, most of these studies were conducted within earlier educational levels (e.g., elementary school contexts; Gentrup et al., 2020). This trend can be partly understood in relation to the hypothesis that older students might be more resistant to biased TE than children in lower year levels (Y.-H. Chen et al., 2011; Jussim, 2012). For example, by tracking and comparing the changes in the self-perception of 256 elementary students in different TE classes across a year, Rubie-Davies (2006) found that students' self-perceptions changed over the year in line with their TE, with no statistically significant differences in student baseline self-perceptions. Consistent findings have also emerged in studies conducted in other contexts. Using latent growth curve models, Upadyaya and Eccles (2015) followed

849 elementary students in south-eastern Michigan and their teachers for 4 years, and found that TE predicted both students' concurrent and subsequent SC in reading and mathematics, controlling for students' achievement and general verbal intelligence. Similar results have also been observed in the Chinese foreign language-learning context (M. Zhu & Urhahne, 2015). Based on data from 505 fifth-grade students and their English teachers in Beijing, M. Zhu and Urhahne (2015) found a lower English SC in students who were underestimated by their teachers, despite the same levels of achievement. Furthermore, there are other researchers who have confirmed the existence of TE effects on student SC, even though they have underscored individual variations in such associations. For example, the Finnish work by Pesu et al. (2016) explored positive associations between TE and student SC of ability in reading and mathematics for high achievers but not for low achievers.

Further, when examining the literature of TE effects on student outcomes in more depth, the magnitude of TE effects has been found to vary, which, for instance, may be related to students' background or the particular knowledge domain or student achievement levels (S. Wang et al., 2018). Briefly, despite a small number of inconsistent findings (e.g., de Boer et al., 2010; Speybroeck et al., 2012), previous studies have been broadly in line with the conclusion that stigmatised groups are more susceptible to negative TE effects. As explained by Madon et al. (1997), children from stigmatised groups (such as low achievers) might have lower abilities and less faith in their potential. Alongside these factors, teachers' negative feedback might undermine these students' motivation and hence make them find school life even more difficult and unpleasant. More importantly, compared to those who firmly believe in their abilities, children from stigmatised groups may be more unsure about their own performances, which may lead them to more easily internalise the TE. As such, it is arguable that these students might be more vulnerable to negative TE effects. For example, students with lower SES background, and minority ethnicity could be more susceptible to TE effects and thus there is an increasing possibility for them to conform to the typical lower TE (Z. Li, 2014). In particular, regarding the moderators of TE effects, the review by Jussim et al. (1996) found that TE effects were stronger for girls, students with lower SES background, and African American students. Similarly, based on TE data from 30 teachers (10 each in Grades 1, 3, and 5) in 12 San Francisco Bay Area urban elementary schools and achievement data from 561 children, McKown and Weinstein (2002) found that two groups (African American students generally, and girls in mathematics) were more frequently subjected to unfavourable TE. This finding reveals an important message that the stigmatised groups referred to above should not only be explored in relation to student characteristics (e.g.,

gender, ethnicity, SES), but also be considered in terms of specific subjects (e.g., girls in mathematics and boys in reading). This finding was further supported by the study of Hinnant et al. (2009). Based on a longitudinal study which followed children at 10 different geographic sites from birth to fifth grade, Hinnant et al. reported an important three-way interaction between student gender, ethnicity, and TE in specific knowledge domains. Briefly, in terms of reading ability, they found that TE effects on ethnic-minority boys were stronger than those for students from the ethnic-majority or -minority girls, whereas students from low-income backgrounds were more susceptible to TE effects in the math domain. Following these findings, S. Wang et al. (2018) proposed an interesting point in their review: in addition to being an outcome of TE, students' SC should also be considered to, in turn, mediate TE effects on student outcomes. However, as suggested by S. Wang et al., no study could be identified which had investigated the entire mediating process of TE effects regarding SC. Although the mediating factors influencing the magnitude of TE are not explored in this thesis, the possibility of the variance of TE effects in different components of student SC is a particular focus.

Despite several longitudinal studies (e.g., Hinnant et al., 2009) having been identified, as suggested by S. Wang et al. (2020), there are relatively few studies tracking the TE effects in the context of the same teachers and students for a long time, because teachers and classrooms change every semester or year (e.g., students usually have different teachers every year) and multiple teachers teach different subjects in a classroom. This doctoral project addressed these concerns and contributed to longitudinal studies in this field by exploring HTs who account for students' overall performance in all subjects and with whom the students spend most of their school time for several years; hence, students could be more easily and deeply affected by HTs' expectations. Additionally, given HTs' basic commitments, they may play a more crucial role in students' nonacademic development than subject teachers. Thus, this group could serve as a lens to uncover the differences, if any, in the associations between TE and students' academic and nonacademic SC. This will contribute to an advancement in the field.

According to Ogle et al. (2016), every learning opportunity in schooling contributes to students' knowledge not only about the outside world, but about themselves and their associations with it. To foster student achievement, it is important for them to develop a positive SC and build corresponding skills, attitudes, and characters (such as the ability to navigate through the challenges and keep growing in the face of adversity). Existing literature indicates that these abilities need not be learnt solely from previous achievement in school,

instead, they can be learnt whenever and wherever a child has opportunities to succeed and learn that they can do so. As such, it is undeniable that, compared to subject-related activities with regular teachers, these fields could be more easily impacted by the intimate interactions between students and their HTs, a group highlighted in this research.

2.4 Literature Gaps and the Current Research

2.4.1 Literature Gaps

A number of gaps in the literature have been signalled throughout the review above. These gaps constituted the basis for this doctoral project. This section closes with deliberations about methodological challenges and other controversial issues as well as their implications for the directions and scope in the research.

Firstly, previous research into teachers' experiences has mostly been concerned with the entire group of teachers. The particularity of HTs has been rarely mentioned in earlier studies, let alone in the TE field. Some HTs' characteristics could be missed by the data collected from entire teaching staff, particularly in the context of the Chinese high school, where differences between HTs and regular teachers are especially prominent regarding their roles and responsibilities. Thus, further investigations need to be conducted to promote understanding of Chinese HTs' beliefs and their experiences in their lives and work.

Secondly, compared to research exploring individual student-level TE, studies attending to class-level TE are relatively sparse. Additionally, even the existing research on TE effects show inconsistent conclusions in both Western (e.g., Archambault et al., 2012; Friedrich et al., 2015) and Chinese contexts (e.g., Z. Li & Rubie-Davies, 2017). Meanwhile, empirical research on the role of TE as being related to teaching practices is especially scarce in the Chinese context. Bearing in mind that TE and their emotions and how individuals react to these two factors are mediated by the sociocultural context, there is a need for further studies to be conducted in the Chinese context to better understand the effects of teacher emotions and class-level TE.

Thirdly, there is an underrepresentation of a theoretical and empirical body of work that distinguishes how TE interact with emotions. Although previous literature implies that these two factors are related inextricably to students' development and teachers' own professional experiences, there is a dearth of research investigating these two factors in combination. Therefore, this doctoral project is of great value for teasing out a deeper and more holistic understanding of how class-level TE interact with teachers' emotions and how this interaction impacts their teaching practices and student outcomes.

2.4.2 Current Research

As a result of, and in response to, the gaps identified above, this doctoral project extended the current research in five ways: (1) focusing on the relations between teachers' emotions (joy, pride, love, anger, fatigue, hopelessness, and anxiety) and TE; (2) investigating the relations of TE with students' academic and nonacademic SC; (3) shifting attention from TE effects only on students to both students and teachers; (4) conducting a project in the Chinese high school context; and (5) attending to the group of Chinese HTs, who are connected intensely with students, parents, and other subject teachers.

This project, therefore, employed a mixed-method sequential explanatory design underpinned by five interconnected research questions. Three studies, as guided by the research questions below, were carried out as part of this doctoral project. The research questions were:

RQ1: Do HTs' expectations predict their emotions and, if yes, how?

RQ2A: In what ways, if any, do HTs with different levels of TE show different patterns of emotions?

RQ2B: In what ways, if any, do HTs with different levels of TE and corresponding emotions respond to scenarios differently?

RQ3A: Are there any relations between the SC outcomes of students and their HTs' expectations of them?

RQ3B: If students' SC is associated with their HTs' expectations of them, are there any differences in the associations with different domains of SC?

The research context of the project was senior high schools in a city in Southern China. A typical Chinese senior high school class normally comprises 40 to 50 students (approximately 16–19 years of age) who are taught by the same group of teachers usually for at least 1 academic year (one teacher teaches one subject). In most cases, one of these teachers is the HT of the class. Students usually stay in the same class with the same HT for at least 1 academic year in their senior high school period but often stay with the same HT throughout senior high school.

Built on the review of the literature, the following three chapters will present the three studies designed to address the corresponding research questions above. Chapter 3 details the first study, which used questionnaires to explore the relations between TE and emotions. Teacher participants were categorised with respect to their TE into groups of high-expectation teachers (HiExTs), medium-expectation teachers (MeExTs), or low-expectation teachers (LoExTs). Chapter 4 presents the second study, which elaborated the findings of Study 1 by interviewing HTs about the interactions between their TE, emotions, and their teaching practices. Following the second study, Chapter 5 presents the third and final study, which tracked the academic and nonacademic SC outcomes of students in the homeroom classes of earlier HT participants. A discussion of the limitations and future research directions is included at the end of each study as well as in the final chapter. Lastly, a general discussion of the findings from all three studies is presented in Chapter 6, including a discussion of the significant contributions that this doctoral research adds to the theoretical and empirical body of the existing literature.

Chapter 3: Study 1

How Does it Feel to be a Homeroom Teacher? Relations Between Teacher Emotions and Expectations

This study was designed to explore the relations between TE and teacher emotions in the Chinese senior high school setting. Controlling for students' prior achievement, HT's expectations of their homeroom students in three major subjects were measured. Additionally, HT's emotions were explored to see if they were related to TE. The overarching research question pertaining to the current study was: do HTs' expectations predict their emotions and, if yes, how?

3.1 Method

3.1.1 Participants

HTs of high school students were invited to complete a questionnaire on their emotions and expectations. A total of 142 HTs (a response rate of 44.24%) from 16 schools agreed to participate, ranging from 1 to 31 teachers per school. Among them, 135 teachers completed both questionnaires on their emotions and expectations whereas 7 teachers participated in the emotion questionnaire only. Specifically, no data on TE were reported from two of 16 schools, and, considering that TE was an essential component of this study, the data from these teachers were not included. Therefore, the final sample size was 135 HT participants from the remaining 14 schools. Additionally, students' achievement scores were collected as part of the measure of TE. With the permission of the schools and the students, the researcher collected students' scores in those 135 teachers' homeroom classes.

The sample consisted of 42 HTs from Grade 10 (31.1%), 60 HTs from Grade 11 (44.4%), and 33 from Grade 12 (24.4%). Of these 135 participants, 77.8 % were female teachers (n = 105), reflecting the gender imbalance among HTs at the high school level (Z. Zhao, 2014). Additionally, the sample echoed the trend that the age of HTs in high schools was younger rather than older as a response to the shortage of HTs (X. Zhu & Liu, 2004), that is, most participants (n = 102, 75.6%) were less than 40 years old, among whom 71 participants (52.6%) were less than 36 years. Finally, consistent with the literature that most HTs teach major courses (T. Wang & Yang, 2021), 24.4% of HT participants were teaching Chinese (n = 33), 23.0% were teaching mathematics (n = 31), and 28.9% were teaching English (n = 39).

3.1.2 Measures

The questionnaire used in the current research was developed based on studying the literature and existing, reliable, and valid instruments, which included: (1) a demographic section designed to collect data about the teachers' basic information (e.g., age, gender, and teaching subjects); (2) the Teacher Emotion Questionnaire (TEQ; Burić et al., 2018) and (3) a teacher expectation survey (Rubie-Davies et al., 2006). All details related to these measures is presented below.

Teacher Demographic Characteristics. In addition to basic information (class number) for the purpose of pairing the data of HTs and their students, other demographic characteristics were also collected, including teacher gender, educational background, age, teaching experience, teaching subjects, and the number of students in their homeroom classes.

Teacher Emotions. To measure teacher emotions, the instrument for this study was an adaptation of the TEQ (Burić et al., 2018), a self-report multidimensional instrument. Although self-report has been criticised for its reliance on the participants' ability and willingness to participate and respond to the questionnaire (Pekrun & Linnenbrink-Garcia, 2014), it is one of the most helpful, efficient, and accurate measure of cognitive and subjective components of emotion. As explained by Burić et al. (2018), in addition to the precise assessment of these two components of emotion, self-report is also useful in measuring other components (i.e., motivational, physiological, and expressive) that can be available to human consciousness.

As a tool to measure teachers' discrete trait emotions, TEQ consisted of six scales, namely Joy, Pride, Love, Anger, Fatigue, and Hopelessness (see Table 3.1 for sample items and see Appendix A for the whole questionnaire). The TEQ's validity and reliability have been demonstrated through a series of five studies that relied on both qualitative and quantitative methods, as well as exploratory and confirmatory factor analytical approaches (Burić et al., 2018). Additionally, as echoed by Lazarus (1991), and Sutton and Wheatley (2003), anxiety is often associated with the complexity of tasks and uncertainty of achieving goals. Given the unique commitment to care and open-ended nature of HTs, anxiety was regarded as a necessary emotion to measure in this study. Therefore, the "Anxiety" scale from the Teacher Emotions Scales (TES; Frenzel et al., 2016) was added to the questionnaire used in this research.

Respondents in this study were asked to indicate their level of agreement with each item of the emotion scales on a 7-point Likert scale (1 = strongly disagree, 2 = disagree, 3 =

slightly disagree, 4 = neither agree nor disagree, 5 = slightly agree, 6 = agree, and 7 = strongly agree). It is important to note that, in order to tap emotions as a more stable trait-like construct, HTs were asked to rate how they usually or typically felt when teaching and interacting with their students.

To be adapted for use in the current context, the original English version was translated into Chinese so that it was better understood by the Chinese participants. To guarantee appropriate translation, a separate translator was invited to back-translate the questionnaire from Chinese to English, and appropriate changes were then made before the questionnaire was distributed to the Chinese participants (Alreck & Settle, 1995).

Table 3.1

Scales	Number of	Cronbach alpha	Sample items
	items	reliabilities across	
		studies	
Joy	5	.80–.87 (Burić et al.,	I am glad when I achieve teaching
		2018)	goals that are set.
Pride	6	.84–.87 (Burić et al.,	I feel like a winner when my students
		2018)	succeed.
Love	6	.90 (Burić et al.,	I feel warmth when I just think about
		2018)	my students.
Anger	5	.7782 (Burić et al.,	The reactions of some students
		2018)	frustrate me so much that I would
			rather just quit the job.
Fatigue	7	.89–.91 (Burić et al.,	My job sometimes makes me so tired
		2018)	that all I want to do is "switch off."
Hopelessness	6	.8488 (Burić et al.,	I feel I cannot do anything more to
		2018)	correct the behaviour of some
			students.
Anxiety	4	.7081 (Frenzel et	I feel tense and nervous while
		al., 2016)	teaching my students.

All Scales and Sample Items for the Teacher Emotion Questionnaire

Teacher Expectations. In addition to the emotion questionnaire, at the beginning of the academic year (October), HT participants were also invited to rate their expectations for each student's academic achievement at the end of the academic year (June), without referring to school records. The measure of TE was conducted around 8 weeks after the school year started, which allowed time for HTs to get to know their students and form expectations of them (see Figure 3.1). This decision was made following the suggestions of

Raudenbush (1984) that teachers tend to form their expectations within the first 2 weeks of school and the TE are then fairly stable after the initial formation.

Figure 3.1

Time Nodes for The Teacher Expectations and Student Achievement Data Collection



Note. TE = teacher expectations, SA = student achievement.

Given that HTs were responsible for students' performance in all subjects rather than just the one they were teaching, the TEs assessed here were the ones for students' academic achievement in three major subjects (mathematics, English, and Chinese). The HT participants were asked to rate their expectations for their students' achievement in the school-year final examination based on a teacher expectation survey scale adapted from Rubie-Davies et al. (2006). The choices for the expected score were divided into 10 levels, covering the range of scores from 60 to 150 (the total score), namely, Level 1 (below 60), Level 2 (60–69), Level 3 (70–79), Level 4 (80–89), Level 5 (90–99), Level 6 (100–109), Level 7 (110–119), Level 8 (120–129), Level 9 (130–139), and Level 10 (140–150). Teacher participants were invited to choose the level that they believed each student would achieve at the end of June when the academic year ended (see Appendix A). Next, teachers' ratings were compared with students' actual achievement at the beginning of the semester (their scores in the latest final examination), as shown in Figure 3.1. More details related to the measure of student achievement are included below. Next, the researcher regressed TE onto actual achievement to obtain the standardised residuals. This method has been widely used in previous teacher expectation research (e.g., S. Wang et al., 2020). In doing so, the data indicated the degree to which teachers were over- or underestimating their students relative to actual achievement.

Further, as shown in the results section below, by using cluster analysis, the statistics program (SPSS) helped categorise the teachers into groups of HiExTs, MeExTs, and LoExTs. Instead of relying on a cut-off (e.g., 0.5 SD above or below) to group teachers, a cluster analysis was adopted because it is a statistically robust method of grouping individuals with

similar characteristics (e.g., teachers with similar class-level TE in this thesis), which has been used in previous TE research (Z. Li & Rubie-Davies, 2017). Specifically, HiExTs had expectations that were significantly above student actual-baseline achievement, whereas LoExTs' expectations were well below student actual achievement. This categorisation not only helped the researcher to uncover the relations between teacher emotions and different levels of TE, but also served to inform the interview questions for the second study and became a foundation for the third study.

Student Achievement. As shown in Figure 3.1, each academic year consists of two semesters in participating schools: Semester 1 from September to January, and Semester 2 from February to the end of June. There are two examinations in each school term: one in the middle and one at the end of the school term. The scores used as student baseline achievement were from the final examination conducted in the last semester of the previous academic year (see "SA" in Figure 3.1). For Grade 10, the scores were from their senior high school entrance examination (*zhong kao*), which is citywide and uses uniform examination papers and uniform marking. For Grades 11 and 12, the scores were from their school-based examinations, which were uniform in each school.

3.1.3 Procedures

Ethical approval was first gained for this study (Ref. 024436) from the University of Auckland Human Participants Ethics Committee (UAHPEC). Then, an email invitation was sent out to 41 randomly selected principals in the city. These invitations included participant information sheets (PISs) that described the overall aims of the doctoral project, as well as consent forms (CFs) to be signed and returned if principals consented for the research to take place with HTs and students enrolled in their schools (see Appendix B). In total, 16 (39%) principals responded agreeing to participate.

After receiving the contact list of HTs from the principals, the researcher approached the HTs by sending them an invitation letter along with the PISs and the Qualtrics link to the questionnaire (see Appendix B). Of the HTs (n = 321) who were contacted, 44.24% (n = 142) consented to participate. However, as suggested earlier, only 135 HTs of these 142 participants finished the complete questionnaire (all three components: demographic information, emotions questionnaire, and TE) whereas 7 of them did not complete the teacher expectation survey, and, therefore, were not included.

After confirming participation with the HT, the researcher asked them to send the PISs to their students or their parents (if the student was less than 16 years old) in their class.

The information sheets for students included a brief account of the research project, its aims and duration, and what was expected of participants (see Appendix B). This meant that students needed to self-select to be involved in the study. Specifically, students' participation in the current study (Study 1) only involved sharing their scores (mathematics, Chinese, and English) in the latest exam. Due to the impact of COVID-19, it was impossible for the researcher to collect students' data in their classes. Hence, students were asked to complete their student number and scores on a Qualtrics link. The student number was used to pair achievement data and teachers' data as a measure of TE. If students were unwilling to share their scores, they could just ignore the link and then the researchers received no access to their scores. As a result, the researchers were able to acquire student scores from all 135 teacher participants who completed the whole questionnaire.

3.1.4 The Conceptual Model

Given the literature review and research question related to this study (see Chapter 2), three conceptual models were proposed and empirically tested in the current study. As shown in Chapter 2, TE, as a component of teaching beliefs, are believed to be involved in both appraisal processes of situations and are further related to teachers' decisions on which emotions to follow (see details in Chapter 2). Following this view, three conceptual models below posit that TE may be related to their positive and negative emotions respectively, and as a whole (Figure 3.2). The researcher recognised that the relations shown in the models below could be bi-directional, however, given that TE was the overarching construct of interest in the current thesis, the purpose of this study was to explore the degree to which TE predicted teacher emotions.

Figure 3.2

Conceptual Models Showing the Hypothesised Relations Between the Variables Investigated in This Study



3.2 Data Analysis Plan

3.2.1 Analysis Plan of The Teacher Expectation Survey

TE were calculated by regressing teachers' ratings onto students' standardised achievement data. Then, to identify possible numbers of teacher groups with different TEs, cluster analysis was conducted in SPSS 23 to aggregate the 135 HTs according to the expectations each HT held for the students in their homeroom class. Ward's method and K- means cluster analysis were used to classify teachers into different expectation groups that had high internal (within-cluster) homogeneity and high external (between-cluster) heterogeneity.

3.2.2 Analysis Plan of The Teacher Emotion Questionnaire

Using AMOS, data analysis of the TEQ was carried out by using 1) CFA consisting of first-order and second-order factors, and 2) SEM techniques based on the maximum-likelihood estimation method.

A first-order CFA was first conducted to confirm the structure of each scale and their variables. Then, a second-order CFA was conducted respectively for the *positive emotions model* and *negative emotions model*. The following advantages led to the decision on the second-order model. Firstly, it allows assessments of the patterns between the first-order factors and the second-order factors. Secondly, it offers a theoretically error-free estimate of particular factors (F. F. Chen et al., 2005). Furthermore, while maintaining measurement accuracy, the second-order model meaningfully decreases the quantity of variables which are required to be estimated (Koufteros et al., 2009).

Conducting a CFA first was important to verify that each measurement model, leading to the final structural equation model, showed good model fit and represented the data well. Based on the CFA results, means and standard deviations for the questionnaire factors were calculated, along with McDonald's ω reliability estimates and Pearson's bivariate correlations. These preliminary descriptive analyses were then followed by examining the relations between the variables investigated in this study using SEM techniques. The results of these tests are presented in the results section which follows. All analyses in this study were conducted using the statistical software IBM SPSS Statistics 23 and AMOS 27.

Additionally, this study followed the guidelines suggested in the literature (e.g., Schreiber et al., 2006) on how to carry out and report results from CFA or SEM techniques. These include (1) clear and succinct research questions that dictate the use of CFA and SEM; (2) a clear theoretical rationale that drives the empirical model; (3) sufficient descriptive statistics that allow future researchers to reproduce the analyses; (4) graphical representation of the hypothesised and final models used in the SEM; (5) parameter estimates (e.g., standardised regression weights and squared multiple correlations), including a range of goodness-of-fit indices to assess model specification and the extent to which the hypothesised model fits the data collected; (6) information on any modification procedures made before proceeding with the analyses; and (7) a discussion of the implications or inferences made from the CFA or SEM results.

Despite the controversy on the best fit indices to describe model specification, there is a consensus to use a range of indices so that the limitations or biases of one index can be covered by another (Gallagher & Brown, 2013). For example, the root mean square error of approximation (RMSEA) is influenced by model parsimony (i.e., higher RMSEA values are found for more complex models), whereas the comparative fit index (CFI) and Tucker-Lewis index (TLI) are affected by sample size and degrees of freedom (i.e., lower CFI and TLI values are found in smaller sample sizes; Meade et al., 2008). Further, chi-square values and significance of normed chi-squared (χ^2 /df) values are found to be sensitive to sample size (e.g., Hooper et al., 2008; Kyriazos, 2018). In this regard, a range of fit indices (χ^2 /df, TLI, CFI, SRMR, and RMSEA) will be reported in the results section of this chapter. As shown in Table 3.2, the acceptable levels of each of those indices were adopted from a number of reviews as listed below on goodness-of-fit indices and were used to determine the fit of the measurement and structural equation models presented in the results section.

Table 3.2

Indices	Perfect fit	Good fit	Rationale
X^2/df	$X^2/df \le 2$	$X^2/df \le 3$	R. B. Kline, 2015
RMSEA	RMSEA $\leq .05$	$RMSEA \leq .08$	Hooper et al., 2008
SRMR	$SRMR \le .05$	$SRMR \le .08$	Brown, 2015
			L. Hu & Bentler, 1999
NNFI (TLI)	$NNFI \ge .95$	$NNFI \ge .90$	Tabachnick & Fidell, 2007
			B. Thompson, 2004
CFI	$CFI \ge .95$	$CFI \ge .90$	L. Hu & Bentler, 1999
			Tabachnick & Fidell, 2007

Acceptable Levels of Fit Indices Regarding CFA

3.3 Results

3.3.1 Identifying Teachers with Different Levels of Teacher Expectations

Standardised Scores for the Previous Final Examinations in Chinese,

Mathematics, and English. Given that achievement data involved three subjects and they were collected from different schools, student scores were standardised by calculating *Z*-scores to make achievement data from different examinations comparable. In doing so, all the achievement scores had a mean of 0 and a standard deviation of 1 regardless of the subjects and examinations. Consistent with the suggestions of H.-Y. Kim (2013) and T. Kline (2005), the distribution analysis indicated that all the standardised achievement data in this research

were well within the guidelines for acceptable skewness (< 2) and kurtosis (< 5) ranging from -1.95 to 0.78 for skewness and -2.56 to 4.60 for kurtosis.

Class-Level Teacher Expectations (Relative to Achievement). Before calculating class-level TE, the researcher first investigated the levels of expectations held by each HT for the individual students in their homeroom class, regarding their performance in each subject (mathematics, Chinese, and English). Single-level regression was conducted where TE (their ratings in the survey) were the dependent variable and student baseline achievement (standardised previous final examination scores) was the independent variable. Each resulting standardised residual was regarded as the extent to which each HT under- or overestimated each of their students' performance in specific subjects.

Based on the standardised residuals mentioned above, class-level TEs were then calculated by averaging the residuals for each HT (see Table 3.5 below for the class-level residuals for all participant HTs). For example, after the regression, one HT (No. 2) yielded 28 residuals for her students in Chinese, 28 for mathematics and 28 for English. Then, the class-level TE of this HT was the average of these 84 residuals. This method has been used in earlier research as a measure of class-level TE (Z. Li & Rubie-Davies, 2017). Since one HT was responsible for one homeroom class in the Chinese context, the mean standardised residuals for each HT were the same as the class-level TE (relative to achievement).

Grouping Teachers of Similar Expectation Levels. This study adopted Ward's method and K-means cluster analysis to identify HiExTs, MeExTs, and LoExTs. Firstly, to identify the optimal number of clusters, the researcher adopted Ward's method which is an efficient and the most widely-used method (Hair et al., 2013). The agglomeration schedule provided a solution for every possible number of clusters for these 135 teachers. The changes in the agglomeration coefficient between a particular stage and the next combination were calculated (see the last column in Table 3.3). As suggested by Hair et al. (2013, 456-462), when larger increases in heterogeneity (the agglomeration coefficient) emerged in moving from one stage to the next, the prior clusters. Following this view, as shown in Table 3.3, there was a rather large increase in heterogeneity in moving from Stage 132 to Stage 133, with coefficient differences of 17.798. Hence, the possible final solution for this study was to cluster HTs by their expectations into three clusters.

Agglomeration Coefficients of Clustering the Average Residuals Using Ward Method (n = 135)

Stage	Number of	Agglomeration	Agglomeration	Difference
	clusters	coefficient next stage	coefficient this stage	
125	10	2.138	1.728	0.410
126	9	2.873	2.138	0.735
127	8	3.695	2.873	0.822
128	7	5.100	3.695	1.405
129	6	7.650	5.100	2.550
130	5	10.659	7.650	3.010
131	4	17.100	10.659	6.441
132	3	34.898	17.100	17.798
133	2	112.174	34.898	77.276
134	1	-	112.174	-

Next, to determine which particular HTs belonged to each cluster, the researcher reran the clustering using the K-means clustering method (K = 3). The K-means algorithm yielded three clusters of HTs with distinguishable expectations (*Cluster 1* \ge 1.06, *Cluster 2* = [-0.10, 0.97] and *Cluster 3* \le -0.20). Table 3.4 shows the number of HTs in each group, where the mean, median, and standard deviation for each group were also provided.

Finally, a one-way ANOVA and post hoc tests were conducted with TE as the dependent variables. The results of the ANOVA test revealed statistically significant group differences in expectations, F = (2,132) = 397.47, p < .001. The post-hoc Scheffé test showed that all three groups were statistically significantly different from each other (all p < .001).

Thus, as suggested by the cluster analysis, three teacher groups could be identified that were then called the Overestimation group, Near-accurate-estimation group, and Underestimation group. As shown in Table 3.4, according to the HTs' different levels of overall TE for students in their homeroom classes, HTs from the Overestimation group, the Near-accurate estimation group, and the Underestimation group were respectively identified as HiExTs (n = 22), MEExTs (n = 62), and LoExTs (n = 51).

			Teacher Group	
		Cluster 1	Cluster 2	Cluster 3
Clustering	All	Over	Near-accurate	Under
result				
Ν	135	22	62	51
Mean	.17	1.61	.45	78
SD	.91	.42	.29	.38

Cluster Analysis Results of Teacher Expectation Groups Based on Average Residuals (n = 135)

Note. Over = overestimation teacher group; near-accurate = near-accurate-estimation teacher group; under = underestimation teacher group.

Mean Teacher-Level Teacher Standardised Residual Values (Relative to Achievement) (n = 135)

Teacher ID	TE	Teacher ID	TE	Teacher ID	TE	Teacher ID	TE
1	98	41	.16	81	70	121	.97
2	60	42	1.30	82	.12	122	.64
3	-1.60	43	10	83	.51	123	.54
4	48	44	.44	84	51	124	.54
5	-1.61	45	93	85	57	125	.57
6	-1.03	46	63	86	.31	126	.40
7	63	47	-1.36	87	01	127	.51
8	.23	48	94	88	.49	128	.59
9	30	49	78	89	69	129	.58
10	.77	50	85	90	.58	130	.66
11	.70	51	-1.75	91	73	131	45
12	.30	52	-1.34	92	76	132	.78
13	.38	53	.18	93	70	133	.68
14	1.15	54	.71	94	37	134	.70
15	.76	55	28	95	.56	135	65
16	.00	56	76	96	.69		
17	.58	57	.22	97	55		
18	.24	58	-1.22	98	05		
19	1.45	59	23	99	10		
20	-1.41	60	.33	100	-1.22		
21	-1.10	61	.43	101	26		
22	-1.22	62	1.11	102	93		
23	2.19	63	58	103	10		
24	.76	64	1.69	104	-1.17		
25	1.21	65	1.16	105	03		
26	1.52	66	.37	106	.78		
27	1.21	67	.92	107	.81		
28	1.89	68	.45	108	56		
29	1.93	69	1.76	109	.64		

Teacher ID	TE	Teacher ID	TE	Teacher ID	TE	Teacher ID	TE
30	.53	70	09	110	47		
31	2.05	71	31	111	.45		
32	1.65	72	53	112	.87		
33	.51	73	76	113	20		
34	1.12	74	-1.04	114	.54		
35	.48	75	73	115	41		
36	1.62	76	94	116	.38		
37	2.27	77	10	117	.54		
38	1.06	78	2.51	118	52		
39	1.58	79	.31	119	44		
40	1.89	80	.35	120	.85		

Note. TE = teacher expectation. Overestimations are presented in red (based on the cluster analysis). Underestimations are shown in green.

3.3.2 The Measurement Model on Teacher Emotions (Confirmatory Factor Analysis) First-Order CFA: Validating Each Subscale in The Teacher Emotion

Questionnaire. To establish the factor structure of the teacher emotion questionnaire, each subscale (namely, Joy, Pride, Love, Anger, Fatigue, Hopelessness, and Anxiety) was examined first using a CFA technique. In first-order CFA, a latent variable is measured based on several indicators that can be measured directly (see Figure 3.3 for an example of the CFA model of Joy). In doing so, items were dropped if: (1) the item's factor loadings were less than 0.40, or (2) one (redundant) item in a pair of items had high residual covariance (mostly due to similar wording or content). This procedure resulted in omitting six items from the original 39 items (resulting in the following items removed for each subscale: Joy 1, Pride 4, Pride 6, Love 4, Love 5, and Fatigue 3). Table 3.6 below describes the fit indices for these seven models, each of which had at least three items and showed acceptable fit to the data. Therefore, these seven factors were retained for further analyses (see Table 3.7 for a list of all remaining items measuring each scale).

Figure 3.3

CFA Model of Joy Showing the Standardised Regression Weights for Each Item



Table 3.6

Goodness-of-Fit Indices for Each Subscale in Teacher Emotion Questionnaire

	χ2 (df)	χ2/df	TLI	CFI	SRMR	RMSEA
Acceptable Values	<i>p</i> > .05	< 3	>.90	>.90	<.08	<.08
Joy	2.798 (2)	1.399	0.997	0.999	0.0049	0.055
	<i>p</i> > .05					
Pride	3.327 (2)	1.663	0.993	0.998	0.0094	0.070
	<i>p</i> > .05					
Love	5.336 (2)	2.668	0.987	0.996	0.0060	0.112
	<i>p</i> > .05					
Anger	8.274 (3)	2.758	0.964	0.989	0.0277	0.062
	<i>p</i> > .05					
Fatigue	17.473(9)	1.941	0.978	0.987	0.0269	0.084
	<i>p</i> > .05					
Hopelessness	11.509(9)	1.279	0.992	0.995	0.0202	0.046
	<i>p</i> > .05					
Anxiety	1.179(1)	1.179	0.998	1.000	0.0073	0.037
	<i>p</i> > .05					

Scales	Iten	ıs	
Joy	1	Joy 2	I am joyful when the class atmosphere is positive.
	2	Joy 3	I am happy when I manage to motivate students to learn.
	3	Joy 4	I am happy when students understand the material.
	4	Joy 5	Exerting a positive influence on my students makes me
			happy.
Pride	5	Pride 1	I feel like a winner when my students succeed.
	6	Pride 2	Due to my students' achievements, I feel as if I am
			"growing."
	7	Pride 3	I am filled with pride when I make a student interested
			in my subject.
	8	Pride 5	When I am proud of my students, I feel that my
			confidence is growing.
Love	9	Love 1	I feel warmth when I just think about my students.
	10	Love 2	I love my students.
	11	Love 3	My students evoke feelings of love inside me.
	12	Love 6	I honestly care about each of my students.
Anger	13	Anger 1	I sweat from frustration when the class does not operate
			in the way that I want it to.
	14	Anger 2	The reactions of some students frustrate me so much that
			I would rather just quit the job.
	15	Anger 3	The frustration I feel while working with students
			undermines my job motivation.
	16	Anger 4	Some students make me so angry that my face goes red.
	17	Anger 5	I get an anger-caused headache from the behaviour of
			some students.
Fatigue	18	Fatigue 1	At the end of my working day, I just want to rest.
	19	Fatigue 2	When I finish classes, I feel numb.
	20	Fatigue 4	Due to the speedy pace of work, at the end of the day I
			feel as if I am going to fall down.

Teacher Emotion Subscales and Items Following the CFA Results

Scales	Iten	ns	
	21	Fatigue 5	Sometimes I am so exhausted at work that I only think
			about how to endure.
	22	Fatigue 6	When I finish my work, I feel drained.
	23	Fatigue 7	Sometimes working with children makes me so tired
			that I can barely move.
Hopelessness	24	HPL1	I feel I cannot do anything more to correct the behaviour
			of some students.
	25	HPL2	While working with completely unmotivated students, I
			feel there is no way out.
	26	HPL3	Because of the behaviour of some students, I feel
			completely helpless.
	27	HPL4	I feel hopeless when I think about the achievement of
			some students.
	28	HPL5	It seems to me that I cannot do anything to get through
			to some students.
	29	HPL6	I feel defenceless because I cannot help some of my
			students.
Anxiety	30	Anxiety 1	I feel tense and nervous while teaching my students.
	31	Anxiety 2	I am often worried that my teaching isn't going so well
			with my students.
	32	Anxiety 3	Preparing to teach my students often causes me to
			worry.
	33	Anxiety 4	I feel uneasy when I think about teaching my students.

Second-Order CFA: Validating the Positive Emotions Model and Negative

Emotions Model. Based on the items obtained in each dimension in the first-order analysis, a second-order CFA analysis was conducted. The use of second-order CFA in this study was to examine 1) the positive emotion variable domain consisting of 3 indicators, Joy, Pride, and Love; and 2) the negative emotion variable domain consisting of 4 indicators, Anger, Fatigue, Hopelessness, and Anxiety. The results of the second-order CFA of these two variables are shown below.

Validating the Positive Emotions Model. The factors that relate to Joy, Pride, and Love were set to load onto a second-order factor labelled positive emotions (Figure 3.4). Two items (Pride 3 and Love 6) were removed in the CFA procedure from the positive emotions model because the items had relatively high proposed cross-loadings. The 10-item model with three scales demonstrated an acceptable model fit, that is, CFI = .98; TLI = .98; SRMR = .016; RMSEA = .093; and $\chi 2 = 69.350$; df = 32; $\chi 2/df = 2.167$; *p* >.338 (see Figure 3.4 and Table 3.8). Compared with the indices shown in Table 3.2, all fit indices had acceptable fit values. Thus, it can be stated that the model determined to have three factors was confirmed.

Figure 3.4

Second-Order CFA Model for Positive Emotions



Note. All path coefficients are significant (p < .001).

Scales	Iten	IS	
Joy	1	Joy 2	I am joyful when the class atmosphere is positive.
	2	Joy 3	I am happy when I manage to motivate students to learn.
	3	Joy 4	I am happy when students understand the material.
	4	Joy 5	Exerting a positive influence on my students makes me
			happy.
Pride	5	Pride 1	I feel like a winner when my students succeed.
	6	Pride 2	Due to my students' achievements, I feel as if I am
			'growing.'
	7	Pride 5	When I am proud of my students, I feel that my confidence
			is growing.
Love	8	Love 1	I feel warmth when I just think about my students.
	9	Love 2	I love my students.
	10	Love 3	My students evoke feelings of love inside me.

Positive Emotions Subscales and Items Following the CFA Results

Validating the Negative Emotions Model. A similar process was applied to the negative emotions' subscales. Based on the criteria mentioned above, seven items (Anger 5, Fatigue 1, Fatigue 2, Hopelessness 1, Hopelessness 4, Hopelessness 6, and Anxiety 4) were removed in the CFA procedure from the negative emotions model because the items had relatively high proposed cross-loadings. The 14-item model with four scales demonstrated an acceptable model fit, that is, CFI = .963; TLI = .954; SRMR = .0369; RMSEA = .079; and χ^2 = 134.604; df = 73; χ^2/df = 1.844; p >.397 (see Figure 3.5 and Table 3.9). Compared with the indices shown in Table 3.2, all fit indices had acceptable fit values. Thus, it can be stated that the model determined to have four factors was confirmed.
Figure 3.5





Note. All path coefficients are significant (p < .001).

Table 3.9

Scales	Iten	ıs					
Anger	1	Anger 1	I sweat from frustration when the class does not operate in				
			the way that I want it to.				
	2	Anger 2	The reactions of some students frustrate me so much that I				
			would rather just quit the job.				
	3	Anger 3	The frustration I feel while working with students				
			undermines my job motivation.				
	4	Anger 4	Some students make me so angry that my face goes red.				
Fatigue	5	Fatigue 4	Due to the speedy pace of work, at the end of the day I feel				
			as if I am going to fall down.				
	6	Fatigue 5	Sometimes I am so exhausted at work that I only think				
			about how to endure.				
	7	Fatigue 6	When I finish my work, I feel drained.				
	8	Fatigue 7	Sometimes working with children makes me so tired that				
			can barely move.				
Hopelessness	9	HPL2	While working with completely unmotivated students, I				
			feel there is no way out.				
	10	HPL3	Because of the behaviour of some students, I feel				
			completely helpless.				
	11	HPL5	It seems to me that I cannot do anything to get through to				
			some students.				
Anxiety	12	Anxiety 1	I feel tense and nervous while teaching my students.				
	13	Anxiety 2	I am often worried that my teaching isn't going so well with				
			my students.				
	14	Anxiety 3	Preparing to teach my students often causes me to worry.				

Negative Emotions Subscales and Items Following the CFA Results

3.3.3 Descriptive Statistics

Following the CFA results, means and standard deviations were calculated for all factors identified earlier and are presented in Table 3.10 below. Additionally, the internal consistency was evaluated through McDonald's ω (McDonald, 1978), since it is considered a reliability index with a lower risk of over- or underestimation of reliability than Cronbach's alpha (Hayes & Coutts, 2020). As shown in Table 3.10, all values were greater than .80, which suggested good internal consistency between items within each factor (Hermsen et al., 2013). That is, these items could be meaningfully used in further analysis.

Also, Pearson's bivariate correlations among these factors were calculated. As can be seen in Table 3.11, significant correlations were found between each factor (all p < .05). Using descriptive statistics in SPSS, all positive emotions were ranked higher than all negative emotions. Specifically, Joy was ranked as highest (M = 6.26, SD = 1.10), while Anxiety was rated lowest (M = 3.5, SD = 1.64). Among the four negative emotions, Hopelessness was reported highest (M = 4.31, SD = 1.46).

Table 3.10

Scale	M (SD)	ω	Skew	Kurtosis	
Positive Emotions	5.95 (1.17) 0.98	-1.68	2.19	
Joy	6.26 (1.10) 0.97	-1.61	1.73	
Pride	5.84 (1.31) 0.95	-1.67	2.58	
Love	5.75 (1.24) 0.97	-1.39	1.55	
Negative Emotions	4.00 (1.37) 0.96	0.48	-0.81	
Anger	4.13 (1.51) 0.90	0.33	-1.09	
Fatigue	4.06 (1.58) 0.93	0.11	-1.31	
Hopelessness	4.31 (1.46) 0.86	0.22	-0.75	
Anxiety	3.50 (1.64) 0.93	0.58	-0.67	

Descriptive Statistics and McDonald's ω by Factor

Note. N = 135 for all factors.

Table 3.11

Pearson's Bivariate Correlations Among the Seven Emotion Factors and Teacher

Expectations

Scale	Posi	itive Emot	ions		Negative Emotions			
	Joy	Pride	Love	Anger	Fatigue	Hopelessness	Anxiety	
Positive								.260**
Emotions								
Joy	-							
Pride	.899***	-						
Love	.884***	.906***	-					
Negative								293***
Emotions								
Anger	558***	422***	504**	-				
Fatigue	564***	446***	503***	.743***	-			
Hopelessness	573***	507***	598***	.749***	.756***	-		
Anxiety	639***	577***	645***	.718***	.650***	.720***	-	
ТЕ	.284***	.228**	.247**	175*	243**	271**	346***	-
<i>Note</i> . *** $p < .001$, ** $p < .01$, * $p < .05$.								

3.3.4 Exploring Relations: A Structural Equation Model

After the CFA, the SEM technique was employed to build the empirical models for this study. Models of how TE are related to teacher positive emotions and negative emotions were explored separately and will be presented under subsections below as follows: (1) TE predicting positive emotions; (2) TE predicting negative emotions; and (3) TE predicting both positive emotions and negative emotions. As a measure of effect size, Cohen's (1988) conventional guidelines for determining the strength of relations between two variables (i.e., 0.1 for "small" or "weak," 0.3 for "medium" or "moderate," and 0.5 for "large" or "strong") were used when interpreting the strength of the standardised regression coefficient paths found in this study (Nakagawa & Cuthill, 2007).

Teacher Expectations Predicting Positive Emotions. In accordance with the assumption that TE would predict teacher emotions, a structural model in which all paths from TE to the positive emotion factor were tested. Figure 3.6 is a diagram indicating the standardised regression weights for each path between the positive emotions' variables and TE. The fit indices indicated the model met acceptable thresholds ($\chi 2 /(df) = 1.870$, p > .392; TLI = .978; CFI = .984; SRMR = .019; RMSEA = .081), with numerous statistically

significant paths between the variables. Overall, this model showed that higher levels of reported positive emotions were predicted by higher TE ($\beta = .26$, p < .01, $R^2 = .07$). The size of the standardised beta values reported indicated that TE were a weak-to-medium predictor of teacher positive emotions. Therefore, significant relations between TE and their emotions were found.

Figure 3.6

Structural Equation Model Predicting Paths Between Positive Emotions Variables and Teacher Expectations



Note. ***p < .001, **p < .01, *p < .05. All paths in this diagram are statistically significant.

Teacher Expectations Predicting Negative Emotions. Similarly, a structural model in which all paths from TE to the negative emotion factor was tested using standardised estimates. Figure 3.7 is a diagram indicating the standardised regression weights for each path between the negative emotions' variables and TE. The fit indices indicated that the model met the acceptable thresholds ($\chi 2 /(df) = 1.754$, p > .416; TLI =.953; CFI = .962; SRMR = .0413; RMSEA = .075), with numerous statistically significant paths between the variables. The results showed that higher levels of TE predicted lower levels of negative emotions ($\beta = -.30$, p < .001, $R^2 = .09$). The size of the standardised beta values reported indicated that TE were a medium negative predictor of teacher negative emotions.

Figure 3.7

Structural Equation Model Predicting Paths Between Negative Emotions Variables and Teacher Expectations



Note. ***p < .001, **p < .01. All paths in this diagram are statistically significant.

Teacher Expectations Predicting Both Positive Emotions and Negative Emotions. Figure 3.8 shows a diagram of the empirical model built in this study, with fit indices indicating that the model met acceptable thresholds ($\chi 2 /(df) = 1.672$, p > .433; TLI = .950; CFI = .956; SRMR = .0550; RMSEA = .071), with numerous statistically significant paths between the variables.

As shown in Figure 3.8, there was a negative correlation between positive emotions and negative emotions (r = -.64, p < .001). The path between positive emotions and TE was found to be statistically significantly positive ($\beta = .27$, p < .01, $R^2 = .07$), whereas the path between negative emotions and TE was found to be statistically significantly negative ($\beta = -$.30, p < .001, $R^2 = .09$). The positive path between TE and positive emotions meant that higher TE predicted higher positive emotions; or lower TE predicted lower positive emotions. In contrast, the negative path between TE and negative emotions means that higher TE predicted lower negative emotions; or lower TE predicted higher negative emotions.

Figure 3.8

Structural Equation Model Predicting Paths Between Both Emotions and Teacher Expectations



Note. ***p < .001, **p < .01, *p < .05. All paths in this diagram are statistically significant.

In sum, the results from the CFAs and SEMs above showed that 1) TE positively predicted positive teachers' emotions of joy, pride, and love; and 2) TE negatively predicted negative teachers' emotions of anger, fatigue, hopelessness, and anxiety. That is, positive emotions increased as TE increased and negative emotions decreased when TE was higher.

3.4 Discussion

The purpose of this study was to explore the relations between HT's expectations and their emotions. To be specific, the research question pertaining to the current study was: do HTs' expectations predict their emotions and, if yes, how? Through CFA and SEM, the findings revealed that teacher emotions and TE were associated with each other. Specifically, HTs who reported higher levels of TE also reported an experience of higher levels of pleasant emotions of joy, pride, and love. The opposite pattern of relations was explored for devastating and debilitating emotions, which meant that HTs who had lower levels of TE had higher levels of anger, fatigue, hopelessness, and anxiety. After showing the overall patterns of emotions in Chinese HTs, this section will respond to the research question by discussing the structural model identified in the current study.

3.4.1 Understanding Emotions in Chinese HTs' Context

As explained in the literature review, differently from regular teachers, HTs in the Chinese context are responsible for students' overall development. In addition to subject teaching, HTs' commitments are multidimensional and they are expected to work as "doctor," "policeman," and "psychologist" to students (S. Liu & Hallinger, 2018). As such, HTs are believed to experience more negative emotions than regular teachers, which could be largely attributed to their excessive workloads, especially noninstructional work (S. Liu & Onwuegbuzie, 2012; Z. Zhao, 2014). However, the results of this study showed that, although the rating of Fatigue caused by heavy workloads was relatively high, the Hopelessness related to student's misbehaviours and attitudes was the most common negative emotion of HTs. This result is not unexpected when considering the more intimate relationships between students and their HTs than with regular teachers. Additionally, it is worth noting here that HTs are responsible for students' moral and character education, which means HTs are required to deal with students' misbehaviours (e.g., cheating, fighting) whereas other regular teachers are not. That is not saying that regular teachers do not care about these scenarios, but rather, HTs are more likely to be impacted by these encounters given that they are held accountable. This could partly explain why Hopelessness caused by students' misbehaviours and unmotivated attitudes (as indicated by the items measuring hopelessness) had the highest means in HT's negative emotions. Finally, findings on teacher emotions were indicative of the complicated professional conditions that HTs often face and suggested that their emotions were related to various factors rather than merely structural conditions. In other words, HTs' emotional experiences should be understood and promoted through a more comprehensive lens instead of structural factors such as workloads or the imbalance between their income and workload.

3.4.2 Understanding the Structural Relations Between HT's Emotions and TE

The overarching aim of the current study was to examine whether TE could predict HTs' discrete emotions (i.e., joy, love, anger, hopelessness, and anxiety) and in what ways. It is worth noting that the cross-sectional design of this study only allowed the relations to be explored rather than drawing any causal inferences. Based on the theoretical rationale, the models in the current study were indicative of the beneficial role of TE in developing HTs' emotional well-being by being positively related to pleasant emotions (i.e., love) and negatively to the debilitating emotions (i.e., hopelessness). The following two sections will discuss how these patterns can be understood in relation to the TE research and previous theory on the emotion appraisal and regulation process.

Teacher Expectancy Effects. The pattern between TE and emotions can be understood in relation to self-fulfilling-prophecy effects (Merton, 1948), whereby an individual tends to behave in a way to realise their prophecies and, therefore, is more likely to achieve their expectations. In other words, an HT with higher expectations might provide better instruction and contribute more to students' motivational, affective, and cognitive development (e.g., S. Wang et al., 2018). As such, higher TEs have a greater probability of resulting in improved student outcomes (Z. Li & Rubie-Davies, 2017), which may, in turn, be associated with teacher emotions. Specifically, those teachers who tend to achieve the classroom goals they set (for example, to engage students and to improve their academic performance), are also more likely to have positive emotions. This could be explained in relation to the reciprocal model on causes and effects in the context of teacher emotions (Frenzel, 2014). Additionally, higher levels of TE and consequent improvements in teaching performance and student outcomes might prevent the occurrence of negative emotions such as anxiety.

It is notable that, as shown in the TE effects literature, teachers with lower TE might sometimes absolve themselves from their responsibility for student learning (Diamond et al., 2004; Rubie-Davies et al., 2006) and instead, the students might be blamed for classroom problems when they did not achieve well (Diamond et al., 2004). From this viewpoint, it seems likely that teachers with lower TE might be less frustrated with undesirable teaching and students' unsatisfactory performance because they do not view themselves as culpable for poor student achievement. However, the model in this study showed that teachers with lower TE did not show reduced negative emotions as might be expected given the research cited above. Instead, the results showed that HTs' anxiety related to teaching practices decreased when they had higher TE. In this regard, it could be argued that teachers with higher TE benefitted more emotionally from providing proactive instructional support to students and their commitment to improving their teaching practices than teachers with lower TE.

Furthermore, although this study explored the relations between TE and emotions only, it should not be discussed separately from other teaching beliefs given their interrelations. Previous research has shown possible relations between TE and other teaching beliefs, such as teaching self-efficacy (Z. Li & Rubie-Davies, 2018), which has been found to mediate teacher emotions (Burić et al., 2020; Klassen et al., 2013). As indicated by Burić et al. (2020), higher levels of teaching self-efficacy predicted more positive emotions (e.g., joy) and less devastating emotions (e.g., anger). The results of the current study shared similar patterns with what has been found in studies of other teaching beliefs. Although the role of other teaching beliefs in teacher emotions and expectations is beyond the scope of this study, it is reasonable to believe that the association between TE and teacher emotions did not happen in isolation.

Appraisal Process and Emotion Regulation. This study found statistically significant relations between TE and teachers' positive and negative emotions, which confirmed the hypothesised conceptual model introduced earlier: teaching beliefs appeared to explain some of the variance in teachers' emotions (see Chapter 2 for details). As shown in the literature review, the degree to which teachers' beliefs matched with the encounters during instructions not only served as a determinant of specific emotions but also mediated the intensity and quality of the emotional responses (Frenzel, 2014). Furthermore, previous research suggested that individuals might cope with their emotions differently in one of two ways: cognitive reappraisal and expressive suppression (Chang, 2013). TE, as one important teaching belief, is theoretically involved in these processes and thus is likely to lead to different emotions. The results of this study corroborated these theories by showing that positive emotions increased as TE increased and negative emotions developed conversely. For example, a teacher with higher TE may have greater confidence in their students' ability, which consequently would be less impacted by a temporary frustrating scenario in which students did not achieve at the expected levels. However, the current quantitative study only attempted to show the relations between TE and their emotions, and it remains unknown how TE mediate teachers' appraisal of situations and their responses to them, and how this then leads to different emotions. Nevertheless, the findings in this study suggested that future research could potentially investigate how TE works in these emotional processes. This will also be explored further in Study 2 using semistructured interviews.

3.5 Limitations and Future Directions

Firstly, this study employed a cross-sectional design which does not enable causal inferences to be drawn. The findings suggested that TE predicted teacher emotions but the causal relations between the two constructs could not be inferred. Additionally, this study only explored the degree to which TE (the overarching factor in this thesis) predicted teacher emotions, whereas the relations between these two factors are probably bi-directional. Furthermore, despite the predictive role of TE on emotions found in this study, it remains unclear whether this relation was consistent across different discrete emotions and time. Future explorations might employ longitudinal or experimental designs to investigate the more dynamic relations between TE and teacher emotions. Specifically, it would be worthwhile exploring whether current levels of TE could predict future levels of teachers'

emotions, or vice versa.

Secondly, self-report data might risk a bias due to its high reliance on the participants' ability and willingness to answer the questions honestly (Pekrun & Linnenbrink-Garcia, 2014). With a view to reflecting on the nature of TE and teacher emotions, teacher interviews or observations could be considered in future investigations.

Thirdly, regarding the sample, approximately 44% of all approached teachers agreed to participate, which might raise questions regarding the characteristics of teachers who declined to enrol. Additionally, although all participants were informed that their responses would be treated confidentially, the possibility of giving socially desirable responses could not be excluded.

Fourthly, this research was conducted in 2020 in the Chinese educational context that was undergoing the impact of COVID-19, which might have influenced teachers' emotions and their expectations of students. Future research should aim at replicating the findings from the current study in diverse contexts. Additionally, since the results of this study have shown the predictive role of TE on emotions in the HT context, further research could replicate these findings with regular teachers. It would be interesting to see whether there are any differences between these two groups regarding the patterns of TE and emotions, and if the associations are similar, why that might be.

Finally, this study explored the relations between TE and teacher emotions, but it failed to explain how TE could be reflected in their emotions. Specifically, it would be worth further exploring how these associations emerged and whether there are any possible mediators between these factors. For instance, the role of TE in teacher's emotion regulation remains unknown, for example, whether and how TE mediates strategies (reappraisal and suppression) for dealing with emotionally stressful situations. In addition, it is possible that other confounding variables (e.g., teacher self-efficacy) could partly account for the observed relations between TE and teacher emotions. Future research could explore these relations further by measuring TE and teachers' emotional regulation strategies and/or other teaching beliefs.

3.6 Theoretical and Practical Implications

3.6.1 Theoretical Implications

This research was the first attempt, to our knowledge, to investigate TE as a potential predictor of teacher emotions. Despite the growing attention to teachers' emotions and abundance of studies on TE, researchers have rarely focused on the relations between these

two constructs. The findings of this study may increase our awareness of the interplay of TE and teacher emotions, both of which have important implications for teaching practices, student outcomes, and teachers' own well-being (e.g., Pesu et al., 2016). Additionally, this study shifted attention from teacher expectation effects of students to teachers themselves (HiExTs and LoExTs). As such, the findings from this research on the relations between TE and emotions of different qualities might contribute to the scarce base of knowledge on the role of TE in the teaching profession.

3.6.2 Implications for Teacher Education

This study has meaningful implications for teacher education and teacher professional development. A better understanding of emotions might contribute to teachers' personal wellbeing and their overall teaching effectiveness (J. Chen, 2020). Furthermore, this research provided empirical data to promote theoretical knowledge on TE and teacher emotions, which could serve as a solid basis for examining how teachers might regulate their emotions and developing relevant interventions. Through the involvement of "emotive" dimensions, the findings will also provide implications for global dialogues which concern teaching improvement.

Specifically, although teachers' negative emotions were a structural condition embedded in the profession (Kelchtermans, 1996), this research suggested that it might be possible to help HTs diminish their negative emotions to some extent. The finding of the role of TE in teachers' emotions can be involved in both in-service and preservice training and intervention programmes. For example, teachers might be trained to use proactive instructional strategies in a way that HiExTs usually do, which could prevent the experiences of negative emotions such as anxiety and fatigue. However, TE and its effects have been rarely involved in teacher education worldwide (Rubie-Davies et al., 2018). Therefore, it is of great significance for teacher education and professional development programmes to consider involving relevant content, which would increase teachers' understanding of the effects that their TE could exert—not only on their students but also on themselves. Furthermore, the retention of teachers should not only be considered in terms of short-term attraction but should also attend to the long-term agenda. A support system that is genuinely concerned about the perspective of teachers is needed. This is a message worth listening to in contexts where schools find it increasingly difficult to retain qualified HTs and teachers.

Reliving teachers' negative emotions and preparing teachers for responding to their emotions effectively can be much easier said than done, considering that this profession is an emotional endeavour (Hargreaves, 2000) and it is unlikely for teachers to exert total control over what happens in their practices, nor over the consequences of their actions (Kelchtermans, 1996). Teachers might find it even more challenging when they are working in a hierarchical culture where frontline teachers' voices are often marginalised or even ignored (Lee et al., 2013). The tension that "to teach is to be vulnerable; to be vulnerable is to be capable of being hurt" (Bullough, 2005, p.23) does not mean that nothing should be done to support teachers, rather, it reminds us of how important it is to prepare teachers to manage these emotions effectively, because "while vulnerability is part of teaching, teachers manage it differently, and these differences have profound importance for teachers and their development, students and their learning, and teacher educators and their practice" (Bullough, 2005, p. 23). Being aware of the interplay between TE and emotions might be the first step to involve emotive dimensions in a TE framework. Based on this, offering professional support to equip teachers with high TE, efficient instructional practices, and adaptive emotion regulation strategies can be another step forward.

3.7 Conclusions

The aim of the present study was to explore the predictive role of TE in teachers' emotions. The results confirmed the hypothesised conceptual model that TE predicted positive emotions and negative emotions in opposite directions: higher levels of TE predicted higher levels of positive emotions (joy, pride, and love) and lower levels of negative emotions (anger, fatigue, hopelessness, and anxiety); lower levels of TE predicted lower levels of positive emotions and higher levels of negative emotions.

In conclusion, this study has contributed to the current body of knowledge on TE and teacher emotions by showing that TE, as a teaching belief, provides a protective shield against negative emotions and is positively related to pleasant emotions. In addition to contributions to theoretical knowledge of TE and their significance in teacher emotions, the current study also has implications for initial teacher education and professional development programmes by illuminating the possibility of promoting teachers' positive emotional experiences using a teacher expectation framework.

Chapter 4: Study 2 Expecting More: From Mind Full to Mindful

In the previous chapter, teacher emotions and teachers' average expectations for students in their homeroom classes were examined. Clear associations were found between these two factors, which engendered questions about *how such relations emerge* and *how they would manifest in teaching and learning*. With the aim of exploring these questions in more depth, this study, the second of three in this doctoral research project, elaborated on the findings of Study 1 by utilising semistructured and scenario-based interviews with the identified high-, medium-, and low-expectation teachers. The research questions pertaining to the current study were:

- (1) In what ways, if any, do HTs with different levels of TE show different patterns of emotions?
- (2) In what ways, if any, do HTs with different levels of TE and corresponding emotions respond to scenarios differently?

This chapter (Chapter 4) consists of six sections. The first section presents the methods of this qualitative study, including the participants, materials, procedures, and data analysis plan. After that, themes that emerged from the interviews will be provided along with relevant quotations in the results sections. The third section will then answer the research questions and discuss the findings in more depth. Limitations of the research design are presented in the fourth section, followed by considerations of the implications for future research and a conclusion section.

4.1 Method

4.1.1 Participants

Individual interviews were conducted with 12 teachers from six different high schools who participated in the questionnaires in Study 1 (Table 4.1). As shown in Chapter 3, 135 high school HTs from 14 senior high schools participated in a questionnaire which evaluated teachers' emotions and TE levels. Based on their average TE (relative to student baseline achievement), a cluster analysis was conducted to identify HT participants as HiExTs, MeExTs, and LoExTs. Specifically, HiExTs were teachers whose TE were significantly above student actual-baseline achievement, whereas LoExTs were those who had TE that were significantly below student baseline achievement. Among the 135 teachers, HTs from the high-expectation (Overestimation) group, the medium-expectation (Near-accurate estimation)

group, and the low-expectation (Underestimation) group were respectively identified as HiExTs (n = 22), MeExTs (n = 62), and LoExTs (n = 51).

Among these 135 participants, the researcher invited the four highest, four lowest and four most accurate (expectations closest to achievement) of the 13 teachers who volunteered to take part in the interviews. As shown in Table 4.1, of the 12 participants in this study, 66.7% were female (n = 8), and 75% self-identified as postgraduate (n = 9) and 25% as undergraduate (n = 3). Teachers reported teaching six different subjects: Chinese (n = 5, 41.7%), mathematics (n = 2, 16.7%), English (n = 2, 16.7%), chemistry (n = 1, 8.3%), and history (n = 2, 16.7%). The participants had an average of 5.6 years of teaching experience (SD = 3.6) and 5.1 years of working as an HT (SD = 3.8). Of these teachers, 8.3% indicated that they had been working as a teacher for nearly 1 year (n = 1), 25% for 1–3 years (n = 3), 33.3% for 4–6 years (n = 4), 25% for 7–10 years (n = 3), and 8.3% for 11–15 years (n = 3), 25% for 4–6 years (n = 3), 25% for 7–10 years (n = 3), 8.3%

Interviewees' attributes							
Interviewee	Gender	Expectations	Subject	Age	Years of Teaching		
Code		Group			Experience		
					(Years of Experience as an		
					HT)		
H1	Male	High	Chinese	35	7 (7)		
H2	Female	High	History	30	4 (4)		
H3	Female	High	Chinese	30	5 (<1)		
H4	Female	High	English	31	5 (5)		
M1	Female	Medium	Chinese	29	3 (3)		
M2	Male	Medium	Mathematics	31	7 (7)		
M3	Male	Medium	English	35	10 (9)		
M4	Female	Medium	Chinese	26	3 (2)		
L1	Female	Low	Chemistry	25	<1 (<1)		
L2	Female	Low	Mathematics	40	15 (15)		
L3	Male	Low	Chinese	29	3 (3)		
L4	Female	Low	History	28	4 (4)		

Interviewees' Attributes

Note. The H in H1, H2... stands for teachers from the high-expectations group. M in M1, M2... stands for teachers from the medium-expectations group. L in L1, L2... stands for teachers from the low-expectations group.

4.1.2 Materials

Based on the questionnaire data in Study 1, individual interviews were carried out with teachers in different expectation groups. Each of these 12 interviews lasted for 30 to 60 minutes and were all audio-recorded and transcribed by the researcher. The purpose of these interviews was to clarify the findings of the questionnaire, to acquire deeper understandings, and to verify the results, which would contribute to the reliability of the conclusions. However, the interviews were also designed to obtain new information about TE and their emotions. The interviews provided the researcher with deeper insight into the relations between TE, their emotions, and how the interactions influenced their teaching practices.

In particular, semistructured interviews and scenario-based methods were used. The decision to conduct the interviews in a semistructured manner was because of the type of data

required to answer this study's research question. Given that how HTs' expectations relate to their emotions remains unknown, and there might be variation among individuals, semistructured interviews were suitable for this study because the order of the questions was alterable and additional questions could be asked (Gray, 2014). This method was useful to probe participants to elaborate on their answers, which contributed to a rich insight into teachers' experiences. As such, the interviews were guided by a set of key open-ended questions (Mutch, 2013) and key interview techniques such as prompting and probing were also used to obtain an insightful understanding of the answers provided by participants. Specifically, the interview started with an open-ended question about teachers' most frequent emotions and teachers' opinions of the relations between their emotions and their expectations of students. Subsequently, the teachers were asked about the influences of emotions and expectations on their teaching practices, how they dealt with emotions in class and how they reacted to different scenarios. Examples of interview questions are shown in Table 4.2. The full list of interview questions is provided in Appendix D. Prompts such as "Why?" or "What do you mean?" and probes such as "How do you deal with this issue?" were used to ensure that participants were able to explain and justify their responses, as well as expand on them. It has been found that in studies where prompting and probing techniques have been used during interviews, participants are more likely to provide critical and relevant insights to the questions, as these techniques of prompting and probing have been shown to facilitate higher order thinking skills (e.g., H. J. Smith et al., 2008).

When it became clear that teachers were aware of their emotions and expectations, they were asked to provide further information on four scenarios developed from the first two interviews with HT participants. The scenario interview was used to explore both *how HTs respond to encounters* (e.g., teacher–student interactions) during instructions and *their reasoning underlying these responses*. Participants were provided the opportunity to ask for background and contextual information (e.g., year level, what prompted the situations) that they wanted to know before answering. The contextual information was standardised to guarantee that all respondents were exposed to consistent knowledge of the context if it was requested. The interviewees were then asked to describe how they might be emotionally impacted, what they might do next and to explain the underlying reasons for their reactions. By providing teachers with a common set of teaching scenarios and core of common contextual information, the scenario interview allowed the researcher to directly compare the responses of HTs in different TE groups to the scenarios, which contributed to explorations of the research questions. As a result, the key interview questions can be found in Table 4.2:

Table 4.2

Interview Questions

Semistructured Questions

- 1. Can you please describe in five words the most frequent emotions you have experienced as an HT?
- 2. In what ways, if any, do you believe that your emotions are related to your expectations of students in your role as an HT?
- 3. How do you think your emotions influence your teaching practices in class?
- 4. How do you think your expectations of students influence your teaching practices in class?
- 5. In what ways, if any, do you believe that you can deal with your negative emotions effectively in class?

Scenario-Based Questions

1. Improper behaviours of students

Please imagine a scenario where you are informed that some of your students have cheated in their exams. But the subject they cheated on was not the one that you teach. This means that you need to address this issue together with your colleagues (e.g., the subject teachers, the school leader....). How do you think this will affect you? How will you respond to this issue?

2. Pressures from parents

Please imagine a scenario where some parents of the students in your homeroom class blame you for what they perceive as regressions in the learning of their children. This means that the parents doubt the quality of your instruction and attribute the failure of the students to you. How do you think this will affect you? How will you respond to this issue?

3. Progress of students

Please imagine a scenario where some students in your homeroom class have made impressive progress in the final exam, especially the subject taught by you. How do you think this will affect you? How will you respond to this issue?

4. Disengagement of students

Please imagine a scenario where students in your homeroom class are relatively less active during class time, especially in the subject taught by you. For example, they are not engaged in the group discussions or are unresponsive in your class. How do you think this will affect you? How will you respond to this issue?

4.1.3 Procedures

Data Collection. Ethical approval was gained for this study (Ref. 024436) from the UAHPEC. The participants in this study were selected from the sample of Study 1, therefore, the processes of recruiting schools and contacting potential teacher participants were the same as what has been described in Chapter 3. The sampling procedures for the interviewees are detailed below.

Because it was important to recruit HT participants at different levels of expectations, purposive sampling was employed to select 12 teachers from the HTs who completed Study 1. Teachers who were willing to be interviewed had already indicated that on their CFs. Study 1 divided the participants into three groups (LoExTs, MeExTs, and HiExTs). Four teachers from each group were selected to attend individual interviews. The researcher took the four highest, four lowest, and four most accurate (expectations closest to achievement) of those who volunteered to be interviewed. However, in order to avoid adverse effects, interviewees were not informed of their expectation levels during the project. Those HTs who agreed to participate in this study were provided with CFs indicating their agreement to be interviewed and audio-recorded. All interviews in the current study were conducted in Mandarin given the context and participants' preferences. Additionally, all interviews were conducted remotely via Zoom due to the constraints of COVID-19. Following data collection, all interviews were transcribed and coded. The data analysis then started, as detailed in the next section.

Translation Procedure. After all interviews had been conducted in Mandarin, the data were translated followed the procedure suggested by H.-Y. Chen and Boore (2010), which was developed from previous literature (e.g., Brislin, 1970; McDermott & Palchanes, 1994) and H.-Y. Chen's (2004) earlier study. The detailed translation procedure is shown in Table 4.3. After coding and analysing the interview data, the student researcher translated the codes and themes into English herself and a bilingual doctoral student who had no association with the study was invited to translate them as well to ensure accuracy. Then, another external doctoral student conducted a back-translation (a target language version, English in this study, was translated back into the source language version, Chinese), with the aim of contributing to the trustworthiness of data in qualitative research (Maneesriwongul & Dixon, 2004). Additionally, regarding the most appropriate translation, the current study adopted the combination of literal translation and "free translation" (Filep, 2009). To be specific, if the interviewee provided clear answers with coherent sentences, the researcher translated them literally. As suggested by Honig (1997, p. 17 cited in Birbili, 2000), "a literal translation (word-by-word) could perhaps be seen as doing more justice to what participants have said

and make one's readers understand the foreign mentality better." Nevertheless, this technique might hamper the readability of the text, which could in turn undermine readers' patience and their understanding of the results (Birbili, 2000). For example, when a sentence in Mandarin involved grammatical and syntactical structures which do not exist in English, translating it into English by simply applying the rules of English structure could compromise its meaning and sense (Filep, 2009). As such, a "free" translation (Filep, 2009) was applied when a readable quotation was required, by adjusting the structure and adding missing fragments.

Table 4.3

Phase	Description of the Phase	Language	By Whom
	1	Version	5
1	Interview data were transcribed verbatim	Chinese	The student researcher
	in Chinese.		
2	Interview data were analysed.	Chinese	The student researcher.
	The codes and themes emerged.		An independent rater
			(Chinese Doctoral
			Candidate 1) was asked
			to recode two randomly
			chosen transcripts.
3	The student researcher translated the codes	Chinese-	The student researcher
	and themes into English herself and also	English	and a bilingual
	invited a bilingual doctoral student to		colleague (Chinese
	translate them. Both translators reached		Doctoral Candidate 2).
	agreement in the final English version.		
4	The codes and themes were back-	English-	A bilingual colleague
	translated (from English to Chinese) by	Chinese	(Chinese Doctoral
	another bilingual doctoral student. Steps 3		Candidate 3).
	and 4 were repeated to decrease		
	inconsistencies between the original		
	version and the back-translation version.		

Translation Procedure

4.1.4 Data Analysis Plan

Thematic Analysis. In current study, the data collection and analysis proceeded

simultaneously, which allowed the emerging findings to inform further data collection (Merriam, 1998). Given the rich nature of data collected from the semistructured and scenario-based interviews, a thematic analysis approach was used with the assistance of Excel to analyse themes within the data. An inductive rather than theoretical approach was applied to allow themes to emerge from the interviews rather than being generated by the interview questions (V. Braun & Clarke, 2006).

Informed by the steps of the qualitative analysis proposed by V. Braun and Clarke (2006), the qualitative data analysis of this study included: (1): becoming familiar with the data by transcribing, reading, and rereading the data while noting down initial codes; (2) generating initial codes; (3) developing themes by systematically collating data under each code and into potential themes; (4) interlinking themes to ensure correspondence with the coded extracts and with the entire data-set; (5) analysing and refining themes; and (6) constructing a report by selecting and analysing examples relevant to the research questions in the findings and discussion.

To be specific, after all the individual interviews had been transcribed by the researcher, the researcher read through each one to make sense of the responses before adding initial codes. When all transcripts had been coded, the researcher examined commonalities and differences and tried to identify themes that emerged from the codes. For example, codes that related to teachers' anger, hopelessness, and fatigue were coded as "negative emotions." This was the process for other emerging themes until a final theme could be established. All transcripts were cross-referenced for commonalities and differences to see if more themes emerged.

The second part of the data analysis involved creating a codebook using a Microsoft Excel spreadsheet, where all the codes (i.e., the collated data of similar features) were grouped under the main themes and subthemes. Then, the frequency of occurrences of each subtheme and theme were recorded for each group of teachers with different TE levels. In doing so, the researcher was able to identify which themes and subthemes occurred more frequently in low-, medium- or high-expectation teachers. Lastly, percentages of codes related to a particular level of TE were calculated and tabulated in the results section to allow for comparisons across the types of HTs. The results section below will report responses across all participants and also any differences between the three TE groups, acknowledging that these groups are small.

Interrater Agreement. The peer debriefing strategy was used to improve the accuracy of the findings and the validity of the qualitative study (Creswell & Miller, 2000).

Before analysing the data, an independent rater (a Chinese doctoral candidate) was asked to recode two randomly chosen transcripts of the 12 interview transcripts to examine the reliability of the data coding process. She was blind to the expectation groups to which these two interviewees belonged. The coding of the independent rater was then compared against the researcher's initial coding, and both were used to calculate a Cohen's kappa statistic of interrater agreement as a measure of data reliability, and to ensure that the qualitative data coding process did not result in substantial measurement error. Landis and Koch (1977) encouraged the use of Cohen's kappa (Cohen, 1960, 1968) because it takes into account the probability of random agreement among raters. Calculating kappa can be done by SPSS or using the formula:

$$\kappa = 1 - \frac{1 - P_0}{1 - P_1}$$

where p_0 represents the observed proportion of agreement among raters, and p_1 represents the hypothetical probability of raters agreeing by chance.

As a result, two kappa values were calculated by SPSS for each of the two transcripts, as shown in Table 4.4 below. Following the guidelines suggested by Landis and Koch (1977), where a coefficient's value between .41 and .60 indicates moderate agreement, values between .61 and .80 suggest strong agreement, and values above .81 suggest almost perfect agreement, the kappa coefficients found in current study suggest high levels of data reliability ($\kappa > .81$).

Table 4.4

Cohen's Kappa Coefficient for the Two Transcripts Analysed for Interrater Reliability

	К	Standard Error	95% Confidence Interval
Transcript 1	.87***	.05	(.97, .77)
Transcript 2	.84***	.06	(.96, .72)

Note. ****p* < .001.

4.2 Results

When analysing the interview data in this study, the researcher did not superimpose categories, instead, themes emerged from the participants' answers. These themes largely indicated how the emotions of HiExTs, MeExTs, and LoExTs were related to their TE and how such interactions were reflected in their teaching practices. Accordingly, thematic analysis of the interview data gave rise to three central themes: 1) *Awareness of TE and Emotions*, 2) *Perceptions of Teacher and Student Roles in Learning*, and 3) *Capacity for*

Resilience in Practices.

More specifically, (1) Awareness of TE and Emotions referred to participants' awareness of their frequent emotions as an HT and their understandings of TE. This category also represented the degree to which HTs were aware of their expectations having to do with their teachings and corresponding emotions. (2) Perceptions of Teacher and Student Roles in Learning emerged from the ways in which teachers tended to respond to the scenarios and their attribution process. This category was strongly related to (3) Teachers' Capacity for Resilience in Practices, which was constituted by the way in which teachers regulated their emotions, the teaching approaches they usually followed and how they were able to mobilise existing resources.

All three themes and their subthemes are shown in Figure 4.1 below. The following subsections will explore each of these themes, subthemes, and codes, along with HT quotations that illustrate and represent their meaning.



Themes and Subthemes Developed Through Thematic Analysis of Interview Data

4.2.1 Theme 1: Awareness of TE and Emotions

Theme 1 was related to teachers' own awareness of their TE and emotions, which worked as a foundation for the following findings. This section compromises three subthemes: 1) awareness of emotions, 2) awareness of TE, and 3) perceived relations between TE and emotions. The subsections below will report these three parts respectively.

Awareness of Emotions.

Frequent Emotions. Various emotions were mentioned and discussed during the interviews. In total, 17 kinds of emotions were proposed by HT participants, among which five were positive emotions (happiness, gratification, sense of achievement, hopefulness, and interest) and 12 were negative emotions (anxiety, irritability, grievance, nervousness, anger, pressure, boredom, guilt, sensitivity, fatigue, disappointment, and helplessness). Examples of

quotations are shown in Table 4.5. Overall, as shown in Table 4.6, negative emotions (n = 76) were mentioned more frequently than positive emotions (n = 46). Regarding the whole group of participants, happiness (n = 22, 47.8%) and sense of achievement (n = 13, 28.3%) were the most common positive emotions. Anxiety (n = 17, 22.4%), irritability (n = 17, 22.4%), and anger (n = 11, 14.5%) were the most common negative emotions.

In particular, positive emotions were mentioned more frequently by the HiExT group (n = 19, 41.3% in HiExT group, n = 14, 30.4% in MeExT group, and n = 13, 28.3% in LoExT group) whereas negative emotions were more common in the LoExT group (n = 8, 10.5%) in HiExT group, n = 27, 35.5% in MeExT group, and n = 41, 53.9% in LoExT group). Specifically, in the LoExT group, irritability (n = 11) and anxiety (n = 9) were the most frequently mentioned emotions. In the MeExT group, happiness (n = 6), sense of achievement (n = 6), and irritability (n = 6) were the most frequently mentioned emotions, followed by anxiety (n = 4), aggrieved (n = 4), and anger (n = 4). In the HiExT group, happiness (n = 12) was the most common emotion, which was mentioned far more frequently than the remaining emotions.

Table 4.5

Codes	Quotations
Positive Emotions	
Happiness	"My students are very heart-warming, which makes me happy when I get along with them." (Teacher H4)
Gratification	"I'm gratified to find that my students love me very much, although sometimes I'm harsh on them." (Teacher L2)
Sense of	"When my students trust me and rely upon me, I feel a sense of achievement for being an HT." (Teacher M4)
Achievement	
Hopefulness	"I'm hopeful about the progress made by the students in my class." (Teacher H4)
Interest	"I can learn something exciting and fun from my students, which makes me feel this job is interesting." (Teacher H2)
Negative Emotions	
Anxiety	"My anxiety is mostly related to my students' score." (Teacher L4)
Irritability	"It is irritating to be an HT especially when you have so much on your plate and parents just sidestep their
	responsibility." (Teacher L3)
Grievance	"I feel aggrieved when parents complain or doubt the quality of my work." (Teacher L1)
Nervousness	"I feel nervous when my students are about to have examinations. I know everything has been well prepared, but I
	just feel that's not enough." (Teacher L2)
Anger	"I feel angry when my students behave improperly and when some parents make unreasonable demands." (Teacher
	M4)
Pressure	"The pressure is more self-driven. It's more from my own expectations of my work." (Teacher L4)

Codes and Quotations Relating to Their Emotions

Codes	Quotations
Boredom	"HTs are required to sit in on lots meetings that are long, boring, unproductive, and sometimes even self-indulgent
	on the part of the meeting leader. This is the most boring part of being an HT." (Teacher L4)
Guilt	"I feel guilty when I have blamed my students wrongly." (Teacher L1)
Sensitivity	"I'm very sensitive to students' mood swings." (Teacher H2)
Fatigue	"Classroom management can be said to be trivial, and to be very complicated and painstaking work. I'm pretty
	exhausted when dealing with these issues." (Teacher M2)
Disappointment	"I feel disappointed when my students perform badly." (Teacher L3)
Helplessness	"Sometimes I have no idea how to address tricky problems. I don't know what to do or who can help me." (Teacher
	L1)

Note. The H in H1, H2 ... stands for teachers from the high-expectations group. M in M1, M2... stands for teachers from the medium-expectations group. L in L1, L2... stands for teachers from the low-expectations group.

Table 4.6

Number of Teacher Utterances Containing Codes Relating to Their Emotions

	č	0		
Codes	HiExT	MeExT	LoExT	Total
	Group	Group	Group	
	(N = 4)	(N = 4)	(N = 4)	
Positive Emotions				
Happiness	12	6	4	22 (47.8%)
Gratification	2	2	4	8 (17.4%)
Sense of achievement	3	6	4	13 (28.3%)
Hopefulness	1	0	1	2 (4.3%)
Interest	1	0	0	1 (2.2%)
Total	19 (41.3%)	14 (30.4%)	13 (28.3%)	46
Negative Emotions				
Anxiety	4	4	9	17 (22.4%)
Irritability	0	6	11	17 (22.4%)
Grievance	0	4	1	5 (6.6%)
Nervousness	0	1	2	3 (3.9%)
Anger	1	4	6	11 (14.5%)
Pressure	0	3	3	6 (7.9%)
Boredom	0	0	2	2 (2.6%)
Guilt	0	0	1	1 (1.3%)
Sensitivity	1	0	0	1 (1.3%)
Fatigue	2	3	0	5 (6.6%)
Disappointment	0	0	2	2 (2.6%)
Helplessness	0	2	4	6 (7.9%)
Total	8 (10.5%)	27 (35.5%)	41 (53.9%)	76

Sources of Emotions. When asked for the sources of their emotions, all participants (n = 12) suggested that their positive emotions (e.g., happiness) came from their interactions with students in their homeroom class and 11 participants reported the positive influence of students' progress. For example, one teacher said that she felt happy when she was trusted by her students, especially when "They were even willing to share their secrets with me rather than their parents or friends" (Teacher H4). Similarly, "Meeting with former students brought

me a sense of achievement" (Teacher L2) and "Learning something new from students increases my interest in this job" (Teacher H2). However, as shown in the examples of quotations in Table 4.5 above, things became complicated when it came to negative emotions. By conducting word searches in Excel, the researcher retrieved relevant utterances and created codes by grouping similar sources (Table 4.7). In particular, six codes emerged, namely, student misbehaviour (n = 14, 23.7%), students' fluctuating academic performance (n = 10, 16.9%), tedious noninstructional practices (n = 10, 16.9%), unsupportive parents (n = 10, 16.9%), flaws in the current HT system (n = 7, 11.9%), and blame from colleagues/leaders (n = 8, 13.6%). When comparing the three TE groups, MeExT reported the highest number (n = 25, 42.4%) of negative sources, where students' misbehaviour was mentioned the most frequently. HiExT accounted for the lowest number (n = 12, 20.3%) of sources and the most influential factor reported was a decline in student's academic performance performance. For LoExT (n = 22, 37.3%), both student misbehaviour and parents' blame were mentioned most frequently.

In addition to student misbehaviour and their declining academic performance (two most popular codes), teachers from all three groups mentioned how their emotions were specifically impacted by the particularity of the HT position. For example, innumerable noninstructional practices required by other stakeholders indicated the ambiguous scope of HTs' commitments. Similarly, HTs' complaints about structural problems in the current HT system were related to the ambitious goals they were expected to achieve and the insufficient support that they received. As explained by Teacher M2, "HTs are required to work on all 'grey areas' where subject teachers and parents can't or are unwilling to do so." Following the overarching commitment to promote the overall development of students, all participants signalled the same information as Teacher M3 who said:

I'm aware I need to achieve this, achieve that, but when you ask about what is out of my commitment, I can't figure out anything. I just feel like everything can be counted in my work as long as they are related to students. But the problem is, nothing in school is essentially irrelevant to students. (Teacher M3)

Additionally, it is worth noting that, for those six HTs who spoke about the influences of student academic performance on their emotions, they all pointed out that they felt far more pressured and anxious about students' scores in all subjects rather than the one they taught. As such, it appeared that all HT participants, regardless of their TE, were under pressure from the goal they were expected to achieve in their position and were more vulnerable to external requirements than subject teachers.

Table 4.7

Number of Teacher Utterances Containing Descriptors Relating to the Sources of Negative Emotions

Descriptors	HiExT	MeExT	LoExT	Total
	Group	Group	Group	
	(N = 4)	(N = 4)	(N = 4)	
[student misbehaviour disruptive	3	6	5	14 (23.7%)
students uncooperative students]				
[student academic performance	5	1	4	10 (16.9%)
decline unstable scores]				
[noninstructional practices chores]	3	5	2	10 (16.9%)
[parents' blame	0	5	5	10 (16.9%)
difficult/unsupportive parents]				
[flaws in the HT system structural	1	4	2	7 (11.9%)
problems]				
[blames/misunderstandings from	0	4	4	8 (13.6%)
colleague				
blames/misunderstandings from				
leaders]				
Total	12 (20.3%)	25 (42.4%)	22 (37.3%)	59

Awareness of TE. Compared to the codes related to emotions, HT's awareness of their TE was relatively homogeneous, which means no salient difference was found in this theme across three TE groups. However, findings related to this theme are still reported below because they helped understand the implications of this research by showing what could be done to refine teachers' awareness of their own beliefs, and what was missing from teachers' voices in the current literature.

When asked whether they were aware of their expectations for students, all participants admitted that they had different expectations for different students. HT participants in the current study were more inclined to think of individual-level expectations and only four HTs mentioned class-level TE throughout the whole interviews. Ten participants revealed that they usually had higher expectations for high-ability students and lower expectations for low-ability students. For instance, one teacher said: I held fairly high expectations for the monitor in my homeroom class, including her scores and other fields such as her character and behaviour. But for another naughty boy in my class, regarding his scores, I just want to say, "leave it alone." All I wish is that he doesn't fight with others in the school and no more complaints from subject teachers. And I felt grateful if it is a normal day for him, although his scores remain poor, at least it is a normal day. (Teacher L2)

As a result, teachers provided dissimilar learning opportunities for students of high and low ability. Specifically, regardless of TE levels, participants shared a similar view that they would like to, "Provide more support for low-ability students than high-ability ones after class, which are usually based on basic/repeated knowledge, while allowing high-ability students more independent learning opportunities" (Teacher H2) and "advanced tasks" (Teacher L2). Two LoExTs reported that they felt "dissimilar learning opportunities is unfair to low-ability students, which might make children feel as though they are being looked down on" (Teacher L1), thus they were trying to eliminate these discrepancies.

Furthermore, half the teachers (n = 6) suggested that their TE constituted two parts: expectations for students' academic scores and ones for students' nonacademic performance. It is unsurprising that HTs did not usually hold the same level of expectations for their students' scores and their performance in other fields. However, it is hard to say which parts were more influential in the teacher–student relationships and had further impact on teachers' emotions. For example, one HT said, "Some students were naughty in school, which gave me a super headache. But most of them are affectionate and when they graduated, they come back to see me every year – which is the happiest moment in my profession" (Teacher L2). As will be further discussed below, this finding provided some rationale for future research to attend to the mechanism and influences of nonacademic TE, which could serve as complementary to the existing academic TE literature.

Perceived Relations Between TE and Emotions. Similar to the subtheme above, no salient difference was found in teachers' perceptions of the relations between their TE and emotions. Specifically, when asked whether teachers felt any relations between their TE and their emotions, nine of 12 participants (three LoExTs, four MeExTs, and two HiExTs) immediately confirmed the existence of the relations in their daily work. As indicated by the participants, these relations tended to become more salient when teachers' expectations of individual students were not being met. In particular, seven of 12 HT participants (two LoExTs, three MeExTs, and two HiExTs) said that they felt disappointed when high-ability students fell short of their expectations whereas they might feel more tolerant when low-

ability students did something wrong. This pattern could be understood in relation to HTs' higher expectations of high-ability students and lower expectations of low achievers, as shown above. With the lower expectations of low achievers, HTs tended to anticipate these students would make mistakes and behave improperly. Additionally, HTs were inclined to pay more attention to low-ability students' behaviour and less on their academic performance. As such, HTs seemed to perceive low achievers' academic failure and their disruptive behaviours as normal but felt disappointed and surprised when high-ability students fell short of their expectations. This could partly explain the differences in the sources of negative emotions found above, where both LoExT and MeExT reported students' misbehaviour as the most frequent factor but HiExT were mainly concerned with a decline in student's academic performance.

On the other hand, six participants (two LoExTs, two MeExTs, and two HiExTs) reported that they felt more positive emotions when interacting with high-ability students for whom they had higher expectations. As suggested by Teacher L4, "high-ability students usually react more actively. They are highly organised and they can take charge of their own study. I feel at ease when I'm teaching and interacting with them." However, as revealed above, HTs reported providing more advanced tasks for high-ability students whereas low-ability students were asked to focus more on repeating practices. Therefore, it was hard to distinguish whether HTs' positive experiences with high achievers were led merely by these students or from the differentiated learning opportunities they designed for each group of students, given teachers' reluctance to repeat basic knowledge by suggesting "they're boring" (Teacher H1).

Despite the pattern mentioned above, no participant in the current study could explain explicitly how their class-level TE were related to their emotions, which was the focus of the current research question. This was understandable because each HT was fully in charge of one homeroom class while being a subject teacher for other class(es). Therefore, HT participants were more inclined to attribute any perceived emotional differences between these classes to their role—whether they worked as an HT or merely as a subject teacher for the specific class. It is worth noting here that despite HT's ambiguous explanations on how they believed their class-level TE were related to their emotions, interesting results emerged from teachers' responses to different encounters. As will be revealed in the following subsections, data on *how teachers reacted to frustrating issues* and *how they were emotionally impacted by similar situations to different degrees* helped illuminate the patterns that teachers themselves had not noticed.

Summary of Theme 1. Whereas the first theme could not explain the relations of TE and emotions directly, it worked as a foundation for the following discussion because the research questions cannot be answered without teachers' own understanding of these two factors. From the results above, it appeared reasonable to conclude that HiExTs reported more positive emotions and less negative emotions than MeExTs and LoExTs. This result was consistent with the findings that emerged from Study 1 in the current doctoral project (see Chapter 3). However, when it came to the sources of negative emotions, regardless of TE levels, all HT participants were negatively influenced by the excessive workloads and external requirements rooted in this specific position. In other words, the particularity of the HT position constituted a potential threat to HTs' emotional experiences, whereas HiExTs seemed to manage to relieve the negative influences more effectively than their counterparts with low TE. As will be discussed further in the discussion section, this pattern could be understood in relation to the gap between what HTs were expected to achieve and what they believed their students could achieve. With the high standard that HTs were required to achieve, LoExTs seemed to experience more negative emotions related to their potential concerns about their students' abilities. Additionally, the results of the first theme constituted a basis for discussion of the following themes where the variations in HTs' responses to these negative sources could be explored.

4.2.2 Theme 2: Perceptions of Teacher and Student Roles in Learning

Based on the understanding of teachers' own perceptions of their emotions and TE, Theme 2 reported the differences emerging from the interviews in terms of teachers' role identifications of themselves and their students. Therefore, this theme compromises two subsections: perceptions of HT's roles and perceptions of students' roles.

Perceptions of HT's Roles. Teacher participants in different expectation groups showed divergent views of their role as an HT, including the degree to which they thought they should be involved in different scenarios. Regarding the definition of this position, two identifications emerged from the interview data, where one of them was similar to a traditional mother role and the other was more inclined towards the idea of a facilitator. In particular, two LoExTs and one MeExT identified themselves as the "baby-sitter" and "housekeeper" (Teachers L4 and M3) and "mother" (Teacher L3) of their students. As such, they tended to micromanage the students in their homeroom classes. In contrast, two HiExTs (Teachers H4 and H1) regarded themselves as a "listener/sister" or a "brother" in the interactions with their students and the other teacher from the HiExT group said, "Being an

HT is only part of my job and my job is only part of my life. Whatever students have messed up, all I can do is to teach them, but I don't think I need to be emotionally involved" (Teacher H3). It was unsurprising that HT participants perceived themselves more than merely a teacher in their students' daily life, given the particularity of this position. However, it was interesting to find that HT's high expectations of students were related to relatively lower emphasis on their own role in education and therefore contributed to a more equal teacher–student relationship. Additionally, as will be shown below, following these role identifications, HT participants reported divergent teaching practices and emotional strategies, which were, in turn, associated with their emotional experiences.

Participants' disparate understandings of their roles in education were also reflected in the attribution of students' failure or misbehaviour. When talking about the same scenario (Scenario 1 where students cheat in the exam), teachers showed completely different attitudes and emotions. For example, one teacher from the MeExT group said, "I would feel ashamed if my students cheated in the exam because this means that something goes wrong in my work" (Teacher M4), however, one HiExT said, "I won't feel guilty or ashamed as it's the student's mistake, not mine" (Teacher H4). Additionally, regarding the emotions related to students' progress, all participants indicated their happiness and increasing sense of achievement, despite the discrepancy in their attributions of these positive emotions. Specifically, three HiExTs (Teachers H2, H3, and H4) saw their student progress as related to some degree to student effort whereas two LoExTs (Teachers L3 and L4) and one MeExT (Teacher M1) appeared to see themselves as contributing to the progress, by thinking "my teaching methods turn out to be effective" (Teacher L4).

Perceptions of Students' Roles. By examining the practices of the HTs in the three expectation groups, discrepant perceptions of their students' role were discovered. The importance of student's active role in their learning was mentioned 11 times in total, all of which were raised by HiExTs (n = 7, 63.6%) and MeExT (n = 4, 36.4%) who put greater emphasis on students' ownership of their learning and believed that students should be encouraged and supported to actively pursue and take charge of their learning process. LoExTs, instead, largely highlighted the teachers' role and its impact in students' development. This pattern could be understood by the degree to which the teachers believed that their students could achieve. Specifically, with higher expectations of their students, HiExTs might have had more faith in their students' ability to achieve their goals and, therefore, expected the students to be more than a passive receiver in the learning process.

potential, LoExTs seemed to prefer to rely on themselves to take charge of every detail as a means of risk-aversion. As such, HiExTs appreciated students' autonomy in the teaching practices and, "Make them play a role as the assistant or even leaders" (Teacher H1) in some situations. However, teachers in the LoExT group believed that, "Most children are in the middle, which means how they perform in the test totally depends on how harsh their teachers are. ... most children need to be closely supervised except for those beyond redemption" (Teacher L4). A similar discrepancy was also observed in the learning opportunities provided for the students, which will be more fully investigated in a later part of this section (Theme 3). In addition to disciplined study, differences in the beliefs about the students' role also impacted children's engagement and their development in other fields, such as their leadership capacity. For instance, regarding the questions about carrying out extra-curriculum activities, Teacher H4 reported that "I usually let the class leaders deal with these things. They are highly organised and better at encouraging other children even than me," however, Teacher L4 argued, "I know some teachers leave these issues to their class leaders. But I feel like I couldn't find a reliable student to do these things instead of me." Teachers H4 and L4 were particularly interesting cases as they had worked in the same school and for similar numbers of years, which meant that they might have been faced with the same kinds of task requirements and class leaders. However, these two teachers (one HiExT and one LoExT) had completely different attitudes and actions related to their expectations. As shown in the current study, LoExTs' beliefs seemed to impose a burden on the role of teachers in student development, which might lead teachers to employ exhaustive instructional strategies and in turn increased HTs' stresses due to the feeling of excessive responsibilities.

Furthermore, it is interesting to note that, regarding the third scenario on students' progress, although all HT participants mentioned that they would like to praise their students in class, four teachers (two LoExTs and two MeExTs) stated that they would warn the student not to be overconfident at the same time, which showed their concern for their students' future performance and their doubts about their students' ability to react to their progress appropriately. Furthermore, despite finding that all teachers planned to praise their students who made improvements, as a role model for the class, only one HiExT stated that she would like to, "Ask the student's opinion first to see if she/he wants to be praised openly" (Teacher H2), which showed a strong awareness and respect for students' autonomy and feelings.

Summary of Theme 2. HT participants in the current study showed salient variations in their perceptions of teachers' roles and students' roles in education. These disparities were

further related to the framework of teaching practices HTs preferred and how they interacted with their students in and after class, which seemed, in turn, to mediate the outcomes and HT's corresponding emotions. On one hand, allowing students more autonomy and involving them actively in teaching practices indicated a possibility for HTs to relieve their workloads and pressures to a certain degree. On the other hand, the differences in the perceptions of the key role in education were also associated with teachers' emotional process which involved the appraisal of the outcomes and the emotional regulations. As shown above, and also below, by highlighting the role of the teacher themselves, LoExTs seemed to be more likely to get emotionally involved in frustrating situations and attribute students' failure to themselves (i.e., perceive students' cheating as a flaw in HT's work).

4.2.3 Theme 3: Capacity for Resilience in Practices

Theme 3 concerns divergence in HT's capacity for resilience, which might contribute to understanding why teachers with different levels of TE might have various emotional experiences. This section compromises three subthemes: 1) emotional regulations, 2) teaching approaches, and 3) resources mobilisation, which will be reported respectively in the following sections.

Emotional Regulation.

Preventive Actions. Teachers reported different reactions to the situations they encountered. When comparing the codes across the three TE groups, teachers' responses to scenarios could be largely classified into two categories: preventive work (n = 18) and reactive work (n = 7). As shown in the interview data, teachers in the high-expectation group were more inclined to take preventive actions (n = 12 in HiExT group, 5 in MeExT group, and 1 in LoExT group) to preclude unsatisfactory situations, whereas teachers in the low-expectations group preferred to persist with their original and often traditional plans (n = 0 in HiExT group, 2 in MeExT group, and 5 in LoExT group).

Regarding scenarios which might trigger negative emotions, participants showed varied responses and attitudes towards them, especially for the fourth scenario where children were not engaged in the class. The fourth scenario was related to teaching practices, which meant teachers had the chance to anticipate and perceive students' reactions and to modify their teaching plan interactively. Regarding this situation, all four HiExTs stated that it would not influence their emotions too much because "There is no reason to be upset. When you get upset, the class goes further into a wrong direction." (Teacher H3). Instead of negative emotions, HiExTs cared more about the potential causes: "Why should my class make
children silent? Is it because my class doesn't satisfy children's interests, or is it because it's beyond their ability?" (Teacher H4). It was interesting to find that one HiExT would even "appreciate this situation" and, "Take students' silence as the uniqueness of the class, which provides a chance for me to design something new or change my existing teaching plans" (Teacher H2). Despite different explanations of their emotional experiences, all HiExTs reported that they would like to, "Tailor the teaching plan accordingly to promote the class climate" (Teacher H1). HiExTs' desire to figure out the reasons behind students' unresponsiveness indicated their willingness to adjust to or prevent the situation. The positive attitudes underlying their preventive actions might be related to their high TE of the students, which seemed to encourage them to believe that the situation could be changed with joint effort. Additionally, although the teaching did not proceed as hoped, HiExTs appreciated students' characters rather than merely blaming them for behaving in a way that teachers could not control or might not like.

In contrast, the remaining eight teachers (four LoExT and four MeExT) explicitly stated that this scenario would place them "into a tight corner" (Teacher L1). When asked to elaborate how they would react to or address this issue, all four teachers from the LoExT group and two teachers from MeExT group said they "had no choice" (Teacher L4) but to continue the teaching by doing a "monologue" (Teachers M1), or "with volunteer students only, usually the high-ability students" (Teacher L4), or "picking students on the spot randomly" (Teachers L1 and M3), even "regardless of their willingness to answer the questions" (Teachers L3, L2, and M1). LoExTs' responses here echoed their perceptions of their students mentioned above, in which students were regarded as passive receivers in teaching and learning. Furthermore, as revealed in LoExTs' reactions, it could be detected that they perceived this frustrating situation as something they could not change. Highlighting the teachers' role as the master of the class, LoExTs seemed to regard students' unresponsiveness as an irreconcilable factor, which predicted increasing negative emotions such as anxiety and hopelessness. Additionally, their passive coping strategy might also be related to their teaching priority for finishing the class first and their lack of adaptive teaching methods, as will be shown below.

Attention Deployment. Another strong subtheme found as part of HT emotional responses was where the teachers reported to place their attention during work time. As shown above in Theme 1, there were no major differences in the sources of HTs' negative emotions, which means that regardless of TE levels, all HT participants were expected to deal with similar challenges. However, even under the same circumstance, some teachers

mentioned several times when they deployed their attentions on positive factors (n = 8) whereas others appeared to focus more on negative factors (n = 8), which could lead to various emotional experiences. Specifically, HiExT (n = 5, 62.5%) and MeExTs (n = 3, 37.5%) paid more attention to positive factors whereas MeExTs (n = 4, 50%) and LoExTs (n = 4, 50%) focused more on negative factors.

Six teachers (three MeExTs and three LoExTs) suggested that they easily became angry and frustrated about students' misbehaviour. By contrast, three HiExTs considered that they tended to ignore these issues, as long as they were minor enough, and appeared to focus more on positive sources instead. As explained by one HiExT:

I feel like most mistakes made by children in daily life are not that serious, I mean, they are not principle problems. For example, boys in my class tend to play basketball during lunch time – which is not allowed. But I don't think that's a big deal. These issues are all tiny things and as a human being, we all make mistakes. (Teacher H3)

In contrast, three LoExTs and two MeExTs suggested that they had at least one experience where they "lost composure because of students' small mistakes" (Teacher L1). For example, Teacher M4 (MeExT) relayed an experience during which students were arguing with her. This experience frustrated her to the point that she slammed some papers down on a table and broke down in tears. After sharing their stories, three teachers admitted that "when I look back, it shouldn't happen" (Teacher L3), which showed HT's additional guilt or regret related to these emotional outbursts triggered by minor mistakes.

The examples above illustrate the degree to which HTs with different expectations paid attention to students' misbehaviour—the most frustrating emotional source in the teaching profession identified in this study. Among most of the LoExTs, there was a clear struggle present as they focused "fixedly on students' every action and behaviour" (Teacher L4), which was largely led by their self-portrait as a mother. In other words, LoExTs searched for full control over what happened in their classroom based on their expectations. Once the lived experiences led to a drop in their expectations, the tension signalled that something was out of balance and attracted nearly all of their attention, which, in turn, resulted in emotional outbursts. In contrast, following the belief that children would finally "come back on the right track" (Teacher H2), HiExTs were inclined to allocate their attention to preventive actions (mentioned above) and were more tolerant of small mistakes made by students in their daily life.

Cognitive Changes. Cognitive changes might involve a reappraisal and reinterpretation of a situation. Interpreting the same situation from different perspectives led HTs to various conclusions on the causes and impacts, which in turn influenced their own emotions. As mentioned earlier, in contrast to one MeExT who perceived students' cheating as a "shame," Teacher H2 (HiExT) considered "a mistake as a pathway for children to grow." Reappraisal and reinterpretation strategies were mentioned 17 times in total whereas direct reactions (e.g., blame the student immediately) were mentioned eight times across all three TE groups. In particular, all four participants from the HiExT group frequently mentioned these reappraisal strategies (n = 10) whereas such techniques were less frequently mentioned in the other two expectation groups (n = 3 in MeExT group and n = 4 in LoExT group). For example, when talking about the misbehaviour of students, Teacher L3 (LoExT) perceived them as a kind of "aggression" and said, "I usually lost my temper and argued with disruptive male students in class." The challenge of student disruptive behaviours was no different for HiExTs. For instance, Teacher H1 (HiExT) replied, "That's annoying, especially for novice teachers like me several years ago." However, faced with this same challenge, Teacher H1 continued to say:

But now I have learnt to regulate my emotions by the use of self-talk. I usually think about the potential reasons for these behaviours. I mean, sometimes children just want to attract more attention from their teachers through this means. Attributing these misbehaviours to students' inhibiting sense of insecurity rather than their slights makes me calm down. And then I can really take a breather to figure out what's going on and how to address the problem. Being mad is never the key, fixing the issues is. Similarly, when asked how they felt when students were not engaged in their class, Teacher H4 (HiExT) responded:

Sometimes the class climate largely depends on children's temporary feelings. For example, when students have just finished a gruelling exam, they could be too exhausted to follow you, no matter how much they like your class. Also, I found children are usually sleepy in the first class of a day.

The examples above suggested HiExTs' curiosity about the causes behind the encounters and the degree to which they took different explanations into account. In comparison, participants in the low-expectation group tended to react to the situation directly and interpret the frustrating results in a negative way. It is worth noting that cognitive changes here do not mean to shrink responsibility or find excuses for the unsatisfactory scenarios. Rather, HiExTs showed a more active attitude to explore all potential reasons

(instead of merely blaming their students) and address the demands of a problem instead of withdrawing or focusing on its emotional consequences. This interpretive process, by its nature, was associated with HTs' beliefs of different stakeholders in education, for example, the degree to which the students should be blamed for the mistake and how the situation could be addressed and by whom. With high expectations of students, HiExTs seemed to have more trust, empathy, and understanding of their students and thus, they were more likely to look on the bright side when things went wrong. In contrast, LoExTs reported a more direct and simplistic thinking process, in which teachers were more inclined to take it for granted that anything out of their control was inexorably in conflict with them. Following this view, LoExTs seemed to more easily get angry and feel hopeless, which was an immediate and direct reaction to negative situations. Furthermore, as shown above, LoExTs also experienced emotional repercussions where subsequent negative emotions (i.e., guilt) might occur following their instant outburst (i.e., anger).

Emotional Boundaries. It is interesting to find that teachers treated the emotional boundaries differently, which appeared to be related to their TE levels. Three codes emerged from the interview which were mentioned several times across each group: 1) clear line between private life and work (n = 6 in total, 3 in HiExT group, 3 in MeExT group); 2) sharing emotions with students (n = 2 in total, 2 in HiExT group); and 3) hide or suppress negative emotions (n = 5 in total, 1 in HiExT group, 2 in MeExT group, 2 in LoExT group).

As revealed by the interviews, although all teachers in the three TE groups admitted their emotional burnout (except one HiExT), there were significant discrepancies in their reactions to it between HiExTs and LoExTs. Regarding HiExTs, even if they experienced emotional exhaustion, they knew how to get over it without serious struggle by maintaining emotional boundaries or distance. Emotional boundaries or distance here refer to the line between private life and work, and the extent to which teachers disclose themselves with their students which was mentioned by three HiExTs (Teacher H2, H4, and H3). As explained by HiExTs, separating the emotions caused by work from their personal life helped HTs find a way to calm down, which also implied a window (Teacher H4) for them to reinterpret the situation and figure out effective solutions. By comparison, teachers in the low-expectation group were more inclined to conceptualise their professional roles as involving intangible and boundless responsibility, which some reported was difficult to separate from their private lives. For example, Teacher L4 was fairly concerned about her student even during her sick leave and "checked the scores immediately once I opened my eyes after general anaesthesia surgery."

Additionally, HiExTs emphasised the importance of grasping the distance between teachers and students. Although teachers in all three groups mentioned that "it is unwise to be too close to students" (Teacher L1), HiExTs showed a more appropriate grasp instead of refusing all self-disclosure during teacher-student interactions. Whereas four teachers from low- and medium-expectation groups (2 MeExTs and 2 LoExTs) tended to hide or suppress their negative emotions from their students, thinking, "Teachers shouldn't let students know she is frustrated" (Teacher L1), HiExTs showed more openness to share their emotions with their students. As explained by some HiExTs, they perceived this as an alternative to signal to their students what was happening and they believed that, "Students are sensitive to teachers' reactions and most of them are nice and really care about how their teachers feel" (Teacher H1). For instance, Teacher H3 said: "sometimes I show my children that I'm unhappy—but actually I'm not that much, and then they quickly realise that something has gone wrong. This is even more effective than criticising them harshly, which usually in turn irritates yourself to death." By leveraging the interactions with students, HiExTs were possibly more able to prevent themselves from potential negative situations and, at the same time, contributed to a closer teacher-student relationship. Furthermore, expecting more of students appeared to encourage HiExTs to believe in students' ability to react to their emotions positively, whereas LoExTs reported crude methods by simply concealing their emotions. As shown in the literature review chapter, the suppression of feelings might further predict negative practices and outcomes. Similarly with other patterns in this theme, the differences in how teachers defined the emotional boundaries were captured and compared directly from teachers' answers. However, why HTs with different TEs showed such disparities could not be explained from one perspective. Given the emotional experiences were dynamic, interactive, and reciprocal, it was necessary to explore this field across all themes, as will be shown in the discussion section.

Teaching Approaches.

Priorities and Foci. By conducting word searches in Excel, the researcher retrieved relevant utterances on HTs' priority and foci during their teaching (Table 4.8). Teachers' priorities in terms of subject teaching comprised two main codes, finishing the plan on time (n = 8) or ensuring students' engagement (n = 6). An interesting difference emerging from the data was that the teachers from the LoExT and MeExT group focused more on finishing the class as planned (n = 3 in MeExT group and n = 5 in LoExT group), whereas HiExTs $(n = 1)^{10}$

6) paid more attention to "the quality of the teaching, including the class climate and the reactions of their students" (Teacher H1).

Table 4.8

Number of Teacher Utterances Containing Descriptors Related to Teachers' Priority or Foci in Teaching

Descriptors	HiExT Group	MeExT Group	LoExT Group
	(N = 4)	(N = 4)	(N = 4)
[achieve teaching objectives on time	0	3	5
finish each step finish plan not postpone]			
[students' engagement responses	6	0	0
reactions feelings]			

When asked what they would do when faced with students' silent unresponsiveness, Teacher H1 said, "I would rather stop for a while to talk with my student about something outside of the book" whereas Teacher L2 said, "I'll try to follow the original plan and finish as much as I can." The LoExTs appeared to focus more on whether the learning tasks could be sequentially organised for their students rather than whether their students were actually involved. When asked to elaborate, Teacher L2 responded that she perceived "finishing the class is one of the most essential parts of my work" and Teacher L4 stated that "there are always some students who don't engage whereas others do. I can't postpone every time as each class has its objectives." In other words, she was treating this instance as something out of her control and would rather complete the course with the few students (usually the highability students) who were always engaged. This priority was followed by her statement regarding how she saw herself as a teacher: "I see myself as a 'wage earner'-I get paid, teach and then go. So, I just want to finish the class on time. I prefer teaching those (active) students, which makes me happy. And I feel that wasting my class time with students who don't want to learn is usually futile." However, as mentioned previously, LoExTs such as Teacher L4 also showed a strong sense of responsibility who even "checked the scores immediately after general anaesthesia surgery" (Teacher L4). By combining the data here and those in previous themes, it could be argued that LoExTs' dilemma reflected or was rooted in several conflicts: 1) a mismatch in LoExTs' self-portrait as a teacher (they emphasised their key role in one way but felt reluctant to fulfil this role in another way), and 2) an imbalance between LoExTs' heightened alertness and their passive coping (they were sensitive and exposed to intense emotions but they responded to them by simply denying the existence of

these struggles), or rather, these LoExTs had no other effective choices but to prioritise essential tasks (e.g., finishing the class) as the outlet for their continuous frustrations.

Adaptive Teaching Methods. As shown previously in Theme 2, HiExT participants mentioned that emotional episodes could probably be a precious chance to improve the way they approached and thought about teaching and to make adjustments as necessary. By conducting word searches in Excel, the researcher retrieved relevant utterances on teaching methods. As shown in Table 4.9 below, nine teaching methods emerged from the interviews, which were mentioned 32 times in total across the three TE groups. In particular, HiExTs mentioned more frequently (n = 20, 62.5%) about the activities that they were providing for their students in and after class than teachers in the other two groups (n = 7, 21.9% in MeExT group and n = 5, 15.6% in LoExT group). For instance, Teacher H3 emphasised the importance of making the activities appealing to students by suggesting that: "only when I feel that my class is interesting, then the students will get motivated and engaged. So, the point is how I make the class fun and to improve my teaching ability." As indicated in the interviews, adaptive teaching not only helped foster intrinsic motivation within learners but also contributed to student performance and the class climate, which in turn prevented teachers from frustration related to unsatisfactory outcomes.

Table 4.9

Codes	HiExT Group	MeExT Group	LoExT Group
	(N = 4)	(N = 4)	(N = 4)
[Role Play]	2	1	0
[Scene Practice]	1	0	0
[Painting]	1	0	0
[Multimedia]	1	0	2
[Craftwork]	1	0	0
[Workshop]	5	1	0
[Humorous Language/jokes]	4	2	2
[Customised Homework]	4	3	1
[Supplementary Readings]	1	0	0
Total	20 (62.5%)	7(21.9%)	5(15.6%)

Number of Teacher Utterances Containing Codes Where Teachers Applied Adaptive Teaching Methods

Resource Mobilisation.

Social Networks. Relational resilience is developed through a web of supportive interpersonal relationships (e.g., HTs-students, HTs-parents, and HTs-subject teachers), and highlights the significance of mutual empowerment, growth, and support at the centre of the resilience process (Day & Gu, 2013). In an earlier section it was suggested that, on the one hand, HiExTs had clearer emotional boundaries between their job and their personal life, and on the other hand, they valued highly their students' role in education. However, this does not mean that HiExTs absolved themselves from their responsibility and let their students succeed or fail. Rather, HiExTs appeared from the evidence in their interviews to appreciate considerably all stakeholders' roles (e.g., colleagues and school leaders, parents, students, and administrators) in education and had higher reliance on them instead of working alone (Table 4.10). The researcher conducted word searches in Excel to find out who HTs had involved in their work and grouped similar utterances into one code in Table 4.10. Overall, five stakeholders were mentioned as a source of support across all three TE groups (n = 33). In particular, HiExTs seemed to appreciate these interpersonal supports more frequently (n =18, 54.5%) than teachers in the other two groups (n = 11, 33.3% in MeExT group and n = 4, 12.1% in LoExT group).

For instance, one essential component of an HT's job was to promote moral and character education for all students and to ensure a positive socioemotional climate in class, which constituted a relatively huge workload for HTs. Whereas all participants recognised the burden raised by this expectation, HTs in the current study showed varied strategies to achieve their goal. As mentioned before, all four HiExTs tended to give students autonomy to organise their activities whereas LoExTs took charge of each step. In addition to contributions to students' capacity, involving different stakeholders in the educational process sometimes also worked as a strategy of doing more with less. As explained by Teacher H4: "letting class leaders organise these activities also helps to improve the cohesion of the class at the same time." Similarly, a consistent pattern emerged from the responses across HiExTs where they appreciated the extensive support from administrators and leaders in terms of the major challenges they faced, such as in Scenarios 1 (students' cheating) and 2 (parents' complaints). HiExTs also showed a stronger awareness of involving students' parents in the learning process and "keep[ing] them informed of students' progress" (Teacher H4).

Table 4.10

Number of Teacher Utterances Containing Descriptors Where Teachers Utilise Their Social Networks

Descriptors	HiExT Group	MeExT Group	LoExT Group
	(N = 4)	(N = 4)	(<i>N</i> =4)
[students class leaders class	3	0	0
representatives]			
[parents]	4	1	1
[colleagues subject teachers	6	3	3
experienced teachers]			
[leaders principals]	1	4	0
[other school departments student	4	3	0
affairs office academic affair office]			
Total	18 (54.5%)	11 (33.3%)	4 (12.1%)

Materials. Data in this study suggested how HiExTs marshalled human and material resources in support of their learning and classroom management, while also capturing the dilemmas and tensions with which HTs must contend. Materials here refer to the resources that teachers designed to guide their instruction and the tools that teachers might use for their classroom management. Regarding high school curriculum materials, these often take the form of print-based teacher guides and student textbooks. However, regarding moral education and classroom management, participants reported confusion because there were no designed materials or resources and "it all relies on HTs" (Teacher M2). Despite general dilemmas, the researcher managed to capture several utterances containing the materials HTs used in their practices by conducting word searches in Excel and, it seemed, HiExTs (n = 4, 80%) were more active in mobilising and developing materials both for in-class teaching and after-class activities (Table 4.11). For example, Teacher H1 wove "magnolia in campus" into his Chinese literature class and Teacher H3 designed various "video or interactive, online materials" which were attractive to students in class. Additionally, teachers incorporated their own expertise into teaching, for instance, Teacher M2 was fluent in English so sometimes he taught mathematics in English and provided supplementary materials for his students, which not only contributed to richness and interest in his teaching but also his authority as an outstanding teacher in class. It appeared from the evidence in the interviews that HiExTs

were more willing to incorporate or develop external materials in their teaching, whereas LoExTs seemed to rely on what already existed. As shown in Table 4.11, although the additional materials were mentioned more frequently by HiExTs than teachers in the other two groups, they were still limited regarding the number and contents. This was understandable because involving external materials implied additional workload and proposed high demands on teachers' expertise, especially for HTs who already reported excessive workloads and overwhelming emotions. However, this trend at least revealed an awareness that, compared to LoExTs, HiExTs seemed to be more willing to go beyond the established system to support their students to achieve at higher levels.

Table 4.11

Number of Teacher Utterances Containing Descriptors Where Teachers Utilise Additional Materials to Support Their Work

Descriptors	HiExT Group	MeExT Group	D LoExT Group	
	(N = 4)	(N = 4)	(N = 4)	
[magnolia in campus]	1	0	0	
[self-developed materials online	3 1		0	
materials readings]				
Total	4 (80%)	1 (20%)	0 (0%)	

Summary of Theme 3. The results of Theme 3 appeared to indicate divergence in the teaching practices of HTs with different TEs and their varied emotional strategies. To be specific, HiExTs reported less emotional trauma by adopting more effective emotional regulation, teaching approaches, and resource mobilisation. This pattern could be partly explained by the relations between these strategies and the outcomes. For example, adaptive teaching methods might predict better student outcomes, which were more likely to prevent possible negative situations. At the same time, adopting effective emotional regulation seemed to be positively related to HiExTs' resilience, which helped them to be less impacted by a specific situation than their counterparts. Additionally, this theme should be discussed with the earlier themes on the sources of HTs' negative emotions and teachers' perceptions of their own role and their students' role in teaching and learning. The discussion section below will combine these themes together to answer the research questions in more depth.

4.3 Discussion

As shown in the previous chapter, the first study of this doctoral project found that HTs' emotions were associated with their expectations of students in their homeroom class.

That is, positive emotions increased as TE increased, and negative emotions decreased when TE were higher. The question of interest in Study 2 was to determine whether teachers in different expectation groups seemed to exhibit differential behaviours towards their students, which in turn, may have indirectly signified how their emotions and expectations were related. The interview results in this study indicated that teachers with different levels of TE seemed to vary in their perceptions of the teacher/student role in education and revealed variation in emotional strategies, teaching approaches, and their use of contextual resources. Along with discussing the results reported above, this section will answer the two research questions: 1) In what ways, if any, do HTs with different levels of TE and corresponding emotions respond to scenarios differently? After that, the limitations and implications of this study will be explained.

4.3.1 Answers to Research Question (1): In What Ways, If Any, Do HTs With Different Levels of Teacher Expectations Show Different Patterns of Emotions?

The model in Figure 4.2 below is designed based on the three themes reported above, with the aim of explaining the mismatches in HTs' experiences that emerged from the interview results. As shown in the model, the first gap (Gap 1) concerned the degree to which teachers' TE were aligned with the external requirements from different stakeholders (e.g., parents, colleagues, and school leaders). As reported earlier, and also below, one challenge that all teachers were facing was the excessive external demands and requirements. Under this circumstance, LoExTs who had lower faith in their students' ability which seemed to risk more negative emotions (e.g., anxiety). This seemed to be caused by the significant gap between what they were required to achieve and what they believed their students could achieve. Furthermore, as shown in Theme 2 above, TE levels seemed to interact with teachers' role identifications, that was, HiExTs highlighted students' active role whereas LoExTs emphasised more their own role. When combined this finding with Theme 3 that LoExTs appeared to have lower resilience and less effective strategies in practice, Gap 2 emerged here in which LoExTs highlighted their critical role on one hand but failed to fulfil their role on the other hand. The relatively unsatisfying outcomes related to Gap 2 then created a new mismatch (Gap 3) where LoExTs risked more negative emotions because they were more likely to fail to meet the external requirements.

The three gaps shown in the model reflected where and how those tensions seemed to occur, which could help HTs better understand how their emotions were triggered. From the

results, it appeared that the magnitude of these gaps seemed to be mediated by TE (more intense for LoExTs whereas less intense for HiExTs). Additionally, the model should be understood within appraisal theory (e.g., Scherer, 1999), where the emotional process is led by judgement or appraisal containing the interpretation of a transaction regarding its significance or relevance for the individual's motives, goals, or concerns. In other words, TE levels appeared to mediate the magnitude of gaps in the model, but more importantly, how teachers responded to these gaps also seemed to be related to their TE, which in turn had possible influences on the teachers' emotional experiences. The following sections will explain each gap in more detail.

Figure 4.2



Model of The Relations Between TE And Teachers' Emotions

Gap 1: Goals/External Requirements and Teacher Expectations of Students. Despite the ambiguous descriptions of HTs' commitments, most HT participants in the current study showed a strong sense of responsibility and an awareness of what they wanted to achieve in this position. This consensus underlying HTs' negative emotions related to their commitments and external expectations can be understood in three ways. One possible explanation could be HTs' intimate emotional bonds with their students, which encouraged them to be continuously devoted, which has been found both in the literature (Yin & Lee, 2012) and in this study. Secondly, the structural elements of the HT system should also be considered. All participants in this study confirmed that HTs were selected by school leaders, which meant that teachers with lower performance were more likely to have already been prevented from taking up this position. The third explanation could be the cultural traditions in China, where the teaching profession is deeply influenced by Confucianism (Lim & Thien, 2020) and the public's strict demands for teachers (X. Gao, 2011). As demonstrated in Chapter 2 (literature review) and later in Chapter 6 (general discussion), although the teacherreverence culture of Confucianism improves teachers' professional authority (J. Chen, 2016), it empowers the public to be demanding and requires teachers to be morally and ethically impeccable (X. Gao, 2011). Considering these factors together could explain why both LoExTs and HiExTs seemed to share similar standards of goals and reported being similarly affected by the external requirements.

Examined within the HT group, given that teachers were vulnerable to external scrutiny and requirements and had little chance to alter or adjust them, HTs' different expectations for their students worked as a key variation in their way of fulfilling these goals. HiExTs believed in their students' ability to achieve their best and contribute to teachers' fulfilment of existing goals, whereas LoExTs had less confidence in their students' performance that led to several concerns. Thus, compared to HiExTs, there appeared to be a more obvious gap between LoExTs' teaching beliefs and the existing goals and requirements, which seemed to result in more intense emotions, such as hopelessness, frustration, and anxiety.

As presented in the results section, regardless of HTs' TE levels, all participants had a clear awareness on the ambitious goals they were expected to achieve (despite minor variations in priorities) as well as corresponding workloads and intricate personal relationships, which constituted a major source of their negative emotions. This finding aligns with HTs' overall portrait in previous literature, where this group has long been perceived as struggling with the emotional dilemma by playing multiple roles beyond the normal

classroom experience (S. Liu & Hallinger, 2018; Y. Wang et al., 2015). As such, when interpreting the results of this study, it is worth remembering that, although HiExTs seemed to experience more positive emotions than LoExTs, it did not mean that they managed to completely prevent themselves from negative emotions. Rather, this study illuminated the necessity and rationale of attending to HTs' emotions by indicating the possible different psychological mechanisms of LoExTs and HiExTs. This is especially of significance because existing research has focused more on deficits or problems (e.g., what is going wrong), instead of the strategies that encourage teachers to stay (e.g., what is going right, Hong, 2012).

Gap 2: Emphasis on Teachers' Role and Ability to Fulfil the Role. As shown in the results section, HiExTs and LoExTs seemed to show divergence in their perceptions of the teacher and student role in education. By emphasising the key role of HTs, LoExTs appeared to be more likely to struggle with excessive workloads (e.g., exhaustive instructional strategies) and corresponding overwhelming emotions, such as fatigue. On the other hand, as revealed in this study, LoExTs reported relatively weak capacity for resilience related to their practices. These two factors, hence, formed a possible gap between teachers' beliefs in their irreplaceable role and their insufficient ability to fulfil this role, which might intensify a sense of losing control over the processes and tasks they felt responsible for as teachers.

Specifically, HiExTs' student-centred beliefs underlying their inquiry-based approach seemed to promote positive appraisals and thus elicit pleasant emotions in response to the same scenarios (Jiang et al., 2021). By contrast, it was more possible for LoExTs to set themselves up for failure when they seemed to be less likely to fulfil their ambitious teaching goals. Consequently, teachers who were less confident about a specific activity might have considered it as a threat and, therefore, tried to avoid it (Bandura, 1977). This could partly explain the contradictions in LoExTs' behaviours and their beliefs. By avoiding challenging situations or merely enduring the frustrations, a teacher who was in this dilemma might find it even harder to develop into an effective and resilient teacher. This seemed to, in turn, undermine teachers' capacity shown in Gap 2 and might indicate a possible vicious circle.

Additionally, the mismatch between LoExTs' understanding of their own role and their insufficient capacity to fulfil this role indicated a cognitive dissonance, which has been found to be related to negative emotions (M. Zheng, 2020). As shown in earlier research drawing on experiences within teacher narratives, pleasant emotions were found to signify cognitive congruence, a sense of balance between teachers' beliefs and the interpretation of

the instances; in comparison, devastating and debilitating emotions were found to signify cognitive dissonance (Ruohotie-Lyhty, 2016). The increasing gap between LoExTs' self-portraits as a parent and the uncontrollable situations seemed to intensify such cognitive dissonance, leading to more possible negative emotions.

Gap 3: Outcomes and Goals. Accompanied by the gaps mentioned above, it is unsurprising that LoExTs' lack of effective teaching practices and emotional regulation appeared to predict less promising outcomes. Consequently, a direct gap between the relatively strict goal and discouraging outcomes emerged where LoExTs might risk blame or questions from other stakeholders (e.g., other subject teachers) in education. These negative triggers may partly explain why LoExTs were more likely to experience intense emotions frequently. However, as mentioned above, teachers' experiences seemed to be related not only to the event itself, but also might be associated with individual's appraisals of the situations and their responses.

In addition to the variation in the direct gap between outcomes and goals, LoExTs and HiExTs also seemed to show different emotional responses to the tensions that occurred. Although the sources of variation in teachers' appraisals or reinterpretations of the situations is beyond the scope of this research, it is important to note that data in the current study seemed to have illuminated the possible role of TE in this process. That is, the appraisals of unsatisfactory situations (e.g., three gaps in the model) and corresponding emotions were mediated by, and would mediate, TE. Specifically, as shown in the result section, HTs with different TE reported varied judgements of the situation (e.g., how serious it was), divergent attribution process (e.g., who was to be blamed), and different solutions (e.g., how to fix the situation). Teachers' different responses and reactions will be discussed further below in the responses to Research Question 2. The point here is that, as a response to the trigger (e.g., the gap between outcomes and goals), these different interpretations and reactions seemed to, in turn, mediate teachers' emotions and beliefs. From the data in this study, it appeared reasonable to conclude that LoExTs, who were struggling with how to navigate the pressures of teaching and frustrating situations, might need to be facilitated and supported by engaging in mindful attribution work surrounding the perceived outcomes of their teaching practices.

Referring back to the model in Figure 4.2, all three gaps contributed to a possible explanation on why LoExTs reported more negative emotions, by showing the mismatches between the goals/roles they would have liked to fulfil and their failures to do so. As discussed above, the lens of appraisal theory can be used to interpret how these tensions could ultimately lead to negative emotions. However, the difference here is that, whereas the

110

first gap in the model reflected LoExTs' potential concerns related to their students (LoExTs seemed to be less likely to believe that their students could behave in a way to approach their goals), the second and third gaps were more associated with teachers' emotional strategies and their professional skills. Thus, it is worth remembering the distinction revealed here between *I don't believe this could be achieved* and *I don't know how to achieve this*. Only when the potential patterns have been understood, can HTs be better supported by equipping them with effective approaches in practices, or knowledge / adjustments of teaching beliefs and emotions, or both.

Furthermore, given the cyclical nature of the emotional process (Corcoran & Tormey, 2012), it is necessary to perceive the model depicted in Figure 4.2 as a dynamic experience where continual interactions exist between each component. In addition, it is important to bear in mind that there are interactions moving both ways. For example, teachers' emotional resilience in Gap 2 seemed to mediate how they reacted to the outcomes and the corresponding gaps, with the reverse also true. Additionally, contextual factors are constantly influencing each component, such as personal goals, teaching beliefs, and individual's expressions of emotion.

4.3.2 Answers to Research Question (2): In What Ways, If Any, Do HTs With Different Levels of Teacher Expectations and Corresponding Emotions Respond to Scenarios Differently?

This study's findings corroborated the view that emotions and beliefs might be related to teachers' daily experiences and to the ways that they interpreted these encounters to inform future behaviours (Barcelos & Ruohotie-Lyhty, 2018). HTs with different levels of TEs in this study reported salient variations in practices. However, given the interrelations between TE and emotions mentioned above, it is hard to distinguish whether these divergences in practices were led by TE or emotions, or both. Further, it should be remembered that these were small groups of teachers (only four in each TE group) and therefore drawing any firm conclusions related to specific groups based on this qualitative data is unwarranted. Nevertheless, considering the relations of beliefs with emotions are dynamic, interactive and reciprocal, where cognitions influence emotions and vice versa (Barcelos & Ruohotie-Lyhty, 2018), it is reasonable to discuss these factors as a whole, as shown below.

As indicated in the results section, differences in teachers' practices were found mainly in three fields regarding their capacity for resilience, namely, emotional regulation, teaching approaches, and the mobilisation of resources. Acknowledging the small sample, findings from this study were consistent with previous literature on teacher expectancy effects, where HiExTs have been found to behave in a more inspiring and proactive way than their counterparts with low expectations (S. Wang et al., 2019). Similarly, results shown in this study were aligned with previous research on emotions, which has shown that positive emotions predict more student-centred approaches to teaching (J. Chen, 2019) and inquiry-based instruction (Jiang et al., 2021), whereas negative emotions relate to knowledge transmission (J. Chen, 2019) and direct instruction (Jiang et al., 2021). Additionally, this study also echoed earlier findings on emotional boundaries between students and teachers, which indicated that an appropriate emotional distance helped teachers to achieve a balance between a sense of professionalism and a helpful level of involvement (Aultman et al., 2009).

In addition to the consistent findings mentioned above, several interesting results could be observed through the lens of LoExTs' experiences. As suggested in previous TE effects research (e.g., S. Wang et al., 2019), HiExTs provided a more careful framework for students' learning than LoExTs. For example, compared to LoExTs, HiExTs tended to provide more class-level feedback, more orientation statements, and more statements involving students' previous knowledge and learning experiences. It appeared that these approaches implied larger devotion and workload than the passive coping strategies that LoExTs might prefer. However, although confirming earlier findings related to HiExTs, this study contributed to this field by showing the patterns underlying the traits of LoExTs. In particular, the current study revealed that LoExTs seemed to respond to challenging situations in a less careful way, for instance, by merely ignoring low-ability students or completely complying with the lesson plan, regardless of student needs. However, the results also indicated that LoExTs seemed to be more likely to suffer from the potential workload caused by these less effective approaches and, more importantly, the corresponding discouraging outcomes. For example, when LoExTs prioritised finishing the class regardless of students' involvement, the question remained on the degree to which their teaching could be effective. As a result, it seemed possible that LoExTs might have had to compromise more after-class time to train the low-ability students, which implied larger workloads and could be in turn negatively related to their emotional experiences. As such, behaving in a lowexpectation way seemed to fail to benefit teachers with less workload, rather, LoExTs might have had to work even harder to achieve their goal and risked more negative emotions.

From another perspective, the LoExTs' approaches shown in the current study and in previous literature could also be understood in relation to their teaching beliefs. As discussed above, LoExTs emphasised their own role and thus seemed to be more inclined to adopt

exhaustive pedagogical strategies and micromanage their students. In other words, being a LoExT and behaving in a passive way was not always teachers' subjective choice to absolve their responsibility, rather, it could happen unintentionally or even reluctantly. Thus, it is worth thinking about how teachers could be better supported and what could be improved in the current system to help prevent them from experiencing such a dilemma. If the LoExTs could have been provided with professional opportunities for sharing their emotional experiences and finding more effective strategies to navigate emotionally charged situations, then they might have formed more positive teaching beliefs. Furthermore, bearing in mind that resilience is something that can be developed through a protective and supportive environment (Castro et al., 2010), it is possible for teachers to learn, train, and nurture their resilience. As such, by showing the differences between HiExTs and LoExTs, the current study sheds some light on the importance of developing an environment that allows teachers to form a more resilient and committed identity, which will not only contribute to student outcomes but also to teachers themselves. Nevertheless, all HT respondents in the current study suggested that they had rarely received professional development on how to regulate emotions in their teaching practices or how to prevent and recover from emotional trauma. This is a message worth listening to, especially for the contexts where schools are facing the challenge of supporting and retaining qualified HTs/teachers. A more supportive system should be established which genuinely cares about teachers' perspective and facilitates preservice teachers.

In sum, the above discussion indicated a possibility that the relations between TE and emotions might be partly attributed to the tensions embedded in teaching beliefs and how teachers performed in their profession. This study found that, regardless of the level of TE, all Chinese HT participants had relatively high professional goals and self-expectations, which might be driven by strict scrutiny from the public (X. Gao, 2011) and the heavy societal expectations embedded in the teacher-reverence culture (X. Gao, 2011) in the Chinese context. However, as revealed in the current study, there was an obvious mismatch in LoExTs' beliefs about *what they want/need to achieve, what they believed their students could achieve* and *what they believed they were able to do*. Additionally, the lack of effective strategies strengthened LoExTs' dilemma in which they seemed to be aware of their vulnerability and frustrations but might have been less likely to respond to or regulate their emotions effectively. Generally, the findings in this study contributed to an insightful understanding about teachers' various emotional patterns (e.g., why HiExTs seemed to experience more positive emotions and less negative emotions than others) through the

psychological lenses of TE and emotions. Despite the exploratory and descriptive nature of this study, as shown below, the results that emerged from the interviews did provide some possible implications for teacher educators and school leaders in response to the current challenge in supporting and retaining qualified HTs.

4.4 Limitations

The current study has several limitations which also suggest directions for further exploration. Firstly, for a qualitative study with a small sample size (four teachers in each TE group), the findings of the current study can only be tentative and should be generalised with caution. Variations in geographic context, education level, school types, and teacher types might all lead to divergent findings in similar studies. Nevertheless, this study is the first-ever attempt to look at the emotions of HTs with different class-level TE in the Chinese high school context. This provides some directions for future explorations in this field.

Another limitation is related to conducting the interviews online. Compared with the traditional face-to-face approach, using Zoom software made it less possible to capture the interviewees' body language (Janghorban et al., 2014). In addition, it was possible that interviewees may have abbreviated their answers and acted less naturally than they may have in an onsite interview. Unfortunately, face-to-face interviews were not possible due to the COVID-19 pandemic.

Finally, the self-reported data in the current study were just HTs' anticipated answers. As suggested in earlier studies (e.g., Van Zoest et al., 2002), concerns cannot be eliminated that how teachers describe they would respond to an instance in an interview setting could be different from what they would actually do in their daily professional lives. Therefore, classroom-based studies would be necessary to address these concerns. This would provide an opportunity to compare teachers' responses in reality and in interviews. Such research has the potential to further enhance understandings of teacher responses and the potential relations of their expectations and emotions, which could support the design of subsequent relevant professional learning and development.

4.5 Implications for Future Research and Practice

Although this study was qualitative, the findings did provide insights into the dynamics and complexities of the relations between TE, teacher emotions and teaching practices, which can be incorporated into theory. Specifically, previous literature on class-level TE effects has indicated that high TE positively predicted more proactive teaching practices and higher student achievement after student prior achievement was controlled (S.

Wang et al., 2018). However, little has been known about what high TE means to teachers themselves. From this viewpoint, this study has theoretical implications by illuminating and explaining the potential of TE as related to teachers' own professional experiences, which might work as a starting point for further research to explore these factors in more depth. However, TE measured in this study was only related to student's academic performance. As revealed by participants, there were variations in HTs' expectations of their students' academic development and nonacademic development. Thus, it is important for future research to explore how these two fields are related and how such relations are associated with teachers' practices and their emotions.

Furthermore, although the research on the associations between teachers' beliefs and emotions is still in its infancy, it has continually become a prominent research area in informing teacher education and professional development (Barcelos & Ruohotie-Lyhty, 2018). A comprehensive understanding of TE is of significance for researchers to make use of the rich findings in this area to reinforce teacher education. However, there is a remarkable lack of such content in current teacher education programmes around the world (Rubie-Davies et al., 2018), let alone instruction on how TE could be used to regulate teachers' emotions. Until recent years, teacher education has counted on facilitating teachers with advanced information and strategies to reflect and develop their beliefs without paying substantial attention to teacher emotions (Ruohotie-Lyhty, 2013). However, echoed by previous literature on the key role of emotions in education (Arens & Morin, 2016), this study showed that how HTs interacted with their negative emotions was not a given, but that it could be regulated and influenced. Additionally, the development of teaching beliefs could not occur without emotional involvement. Thus, it is essential for teacher education programmes and/or professional learning development to introduce these two fields together into their courses, especially for HTs whose overall portrait, emerging from the literature and this study, seems insecure and pressured. In addition to advanced skills of class management and pedagogical practices, it is also necessary to help HTs understand how to respond to their emotions effectively and how to prevent themselves from potential emotional trauma. By doing this, Chinese HTs might be facilitated to perform in a way which indeed improves their teaching effectiveness and also contributes to teachers' own experiences and lives.

Finally, as shown in the literature review (Chapter 2), most existing research on teachers' experiences is explored through the perspective of the whole teacher group, which might compromise the particularity of HTs. This study filled in this gap by paying specific attention to this group in Chinese high schools. This study helped promote understandings of

the HT group and called for more empathetic support for them, which could help relieve teachers' reluctance to take up this position and help explain the shortage of experienced HTs in schools (Z. Zhao, 2014).

4.6 Conclusions

This study is the second study of this doctoral project, and elaborated on the findings from Study 1 (a quantitative study). With reliance on individual interviews, the current study has investigated how HTs' expectations for their students were related to their emotions and how such interactions influenced their teaching practices.

HTs' responses in this study, on the one hand, were consistent with previous literature on HTs' vulnerability that seems to be a consequence of the particularity of this position. By exploring HTs' emotional episodes, the study showed how HTs struggled with various triggers of negative emotions. On the other hand, data in this study illuminated that HTs with higher expectations were more inclined to emphasise students' active role in education and thus adopted adaptive teaching approaches. Additionally, HTs with higher expectations showed stronger capacity for resilience, by using various emotional strategies and mobilising different contextual resources. These variations, in turn, predicted HTs' experiences of emotional labour in a positive way.

In sum, the study revealed the dilemma that HTs were experiencing; more significantly, however, it indicated an opportunity for HTs that they could diminish their consequential negative emotions to some degree by expecting more of their students. By disclosing how and why teachers with higher TE experienced more positive emotions, this study could serve as a concise starting point to foster the professional learning development of teachers by equipping them with effective pedagogical approaches and emotional strategies. Additionally, given the key role of teachers' beliefs and emotions in teaching practices and in student development (Arens & Morin, 2016), future studies could work on these factors further based on the findings of the current study, in order to support both teachers and their students to achieve their potential.

Chapter 5: Study 3 Expecting the Best of Students: Teacher Expectations and Student Self-Concept

The previous two empirical studies presented in Chapters 3 and 4 respectively provided evidence for the relations between class-level TE and teacher emotions in the context of HTs in Chinese high schools. It was also found that TE may not only be related to teacher emotions but also associated with HTs' classroom instructional behaviours and their ways of interacting with and responding to students. These findings of TE effects, however, focused only on the teacher level. A remaining question was whether the expectations teachers formed at the beginning of the school year would predict their students' development, in particular, students' SC. To explore how students' SC was related to TE, the current study tracked the SC outcomes of students whose HTs had different TE. Students' SC was measured twice in the same setting—initially in the first month of the academic year (September, 2020) and the later in the last month of the academic year (June, 2021). Detailed procedures and instruments that correspond to this study will be presented below.

The research questions pertaining to the current study were:

(1) Are there any relations between the SC outcomes of students and their HTs' expectations of them?

(2) If students' SC is associated with their HTs' expectations of them, are there any differences in the associations with different domains of SC?

This chapter (Chapter 5) was therefore designed to explore HTs' expectation effects on their students' SC. To present the third of the three studies in this doctoral research project, this chapter begins with a brief description of the current study and then presents the method employed by the researcher. Following that, the results and discussion pertaining to this study will be provided before consideration of the limitations and implications.

5.1 Method

5.1.1 Participants

This third study in the doctoral project involved 30 HTs from the teacher samples taking part in Study 1. HT participants in this study comprised five Chinese teachers (16.7%), 13 English teachers (43.3%), six mathematics teachers (20%), and six teachers in other subjects (20%, chemistry, physics, and politics). With these teachers' assistance, we recruited 482 high school students in their classrooms. As shown in Figure 5.1, during the course of the

study, 134 (27.8%) students (from different schools and classrooms and evenly distributed across grades and genders) completed only the first questionnaire but not the second one. Imputing the missing data risks underestimating variability and adding random bias to the data (Patrician, 2002). Therefore, these students' data were not included in the analyses and the final student sample in this study involved 348 students and their HTs (n = 30). Table 5.1 presents the attributes of the respondents at these two time points and the final sample. Of the 348 remaining student participants, 50.3% were female (n = 175), and 54.3% self-identified as Grade 10 (n = 189) and 45.7% as Grade 11 (n = 159). No Grade 12 students were recruited because they have the national college entrance examination (Gaokao) at the beginning of June and students usually leave school to prepare for the examination the week preceding. The advantage of having the students of HTs in Studies 1 and 2 was that the data from these three studies could be linked.

Figure 5.1





Table 5.1

Variables	SC1	SC1 only	SC1 and SC2
	(<i>n</i> = 482)	(<i>n</i> = 134)	(n = 348)
Gender			
Female	232 (48.1%)	57 (42.5%)	175 (50.3%)
Male	250 (51.9%)	77 (57.5%)	173 (49.7%)
Grade			
Grade 10	244 (50.6%)	55 (41.0%)	189 (54.3%)
Grade 11	238 (49.4%)	79 (59.0%)	159 (45.7%)
Grade 12	0 (0%)	0 (0%)	0 (0%)

Demographics of Student Participants

Note. SC1 = student self-concept measured at the beginning of the school year (Time 1). SC2 = student self-concept measured at the end of the school year (Time 2).

5.1.2 Measures

Teacher Expectations. Data collected from Study 1 were used here as a measure of TE. As shown in Chapter 3, in October 2020, the HT participants rated their expectations for their students' achievement (mathematics, English, and Chinese) using a teacher expectation survey scale adapted from Rubie-Davies et al. (2006). The choices for the expected score were divided into 10 levels, covering the range of scores from 60 to 150 (the total score). Teacher participants were invited to choose the level that they believed each student would achieve at the end of June 2021 when the academic year ended. Next, teachers' ratings were compared with student actual achievement at the beginning of the semester (their scores in the latest final examination). The researcher then regressed teachers' ratings onto actual achievement to obtain the standardised residuals, which indicated the degree to which teachers were over- or underestimating their students relative to actual achievement. Based on the standardised residuals for each HT. Further details have been provided in Study 1 (Chapter 3).

Student Self-concept. The current study explored how students' SC was related to TE across time. Students' SC was measured at two time points (the beginning of September 2020 and the end of June 2021) to create a time lag of approximately 10 months between the two measurement points. This also enabled the researcher to get a baseline SC so that any changes over the year could be associated with TE. The instrument for this study was a

shortened version of the Self-Description Questionnaire III (SDQIII; Marsh & O'Neill, 1984) which was designed to measure multidimensional academic and nonacademic SC for late adolescents. As mentioned in the literature review, the SDQ is theoretically related to the hierarchical model of SC (Shavelson et al., 1976). The development of the SDQIII was based on two previous versions of the SDQ for preadolescents, and the SDQII for early adolescents. Earlier studies have reported that the internal consistency reliability coefficients of the SDQIII ranged from 0.79 to 0.95 (mean $\alpha = 0.90$; Byrne, 1988), and test–retest reliability coefficients ranged from 0.66 to 0.94 (mean r = 0.86; Marsh et al., 1986). The psychometric properties of the SDQIII are considered exceptionally superior (Byrne, 1988). Further studies have employed the instrument in different contexts and found it to be a reliable and valid measure of the SC of late adolescents (e.g., Faria, 1996; Gujare & Tiwari, 2016).

The original SDQIII is a 136-item questionnaire which measures four academic scales (Verbal, Mathematics, General School, and Problem Solving/Creativity), eight nonacademic scales, and one Overall Global SC scale. Given the HTs' commitments, six irrelevant subscales (Physical Abilities, Physical Appearance, Relations with Same Sex Peers, Relations with Opposite Sex Peers, Relations with Parents, and Spirituality) were excluded from the current project. Given that one aim of the current study was to compare the differences in students' specific aspects of SC, the researcher removed the Overall Global SC scale as well as the General School subscale so that the academic ones were specific to particular curriculum areas. Thus, the version of the SDQIII used in this study only included the following nonacademic SC subscales: Honesty/Truthfulness (12 items) and Emotional Stability (10 items), and academic ones: Problem Solving/Creativity (10 items), Mathematics (10 items), and Verbal (10 items; see Table 5.2 for sample items and Appendix C for more details). The original 8-point Likert scale was modified for the present study to a 7-point scale (1 = strongly disagree, 2 = disagree, 3 = slightly disagree, 4 = neither agree nordisagree, 5 = slightly agree, 6 = agree, and 7 = strongly agree) to ensure consistency across all measurement tools employed. The questionnaire was also professionally translated into Mandarin and cross-checked by experienced researchers who were fluent in both English and Mandarin. To guarantee appropriate translation, the questionnaires were then back-translated from Chinese to English by a separate translator, and adjustments were then made appropriately before the questionnaire was distributed to the Chinese participants (Alreck & Settle, 1995).

Table 5.2

Scales	Subscales	Item	Sample Item
		Number	
Nonacademic	Honesty/Truthfulness	12	I often tell small lies to avoid
SC			embarrassing situations
	Emotional Stability	10	I am often depressed
Academic SC	Problem Solving/Creativity	10	I enjoy working out new ways of
			solving problems
	Mathematics	10	I find many mathematical
			problems interesting and
			challenging
	Verbal	10	I have trouble expressing myself
			when trying to write something

All Scales and Sample Items for the SC Questionnaire Measures

5.1.3 Procedures

Ethical approval was first gained for this study (Ref. 024436) from the UAHPEC. The data collection procedures for this study were similar to those described in Study 1, as both studies involved teacher participants from the same HT sample and students from their homeroom classes. However, apart from data related to TE evaluated in Study 1, HTs did not directly participate in Study 3. The participating teachers helped send the PISs and CFs to students and parents/caregivers of students under 16 years old in their homeroom class. The online link for the student questionnaire was shown on student PISs and students who were willing to participate could complete it online privately. As such, teachers would not have known whether their students took part in this research. As shown in Figure 5.2, in addition to the TE collected in October 2020, students' SC was measured through a Qualtrics link twice in the same setting—initially in the first month of the academic year (September, 2020) and the later in the last month of the academic year (June, 2021). Tracking students' SC aimed to compare whether the SC of students in the classes of different HTs changed over time and whether these changes were associated with the TE. This was designed to enable the prediction of TE effects on students' SC, controlling for baseline SC.

Figure 5.2



Time Nodes for The TE and Student SC Data Collection

Note. SA = student achievement, TE = teacher expectations, SC = student self-concept.

5.1.4 The Conceptual Model

Given the literature review (see Chapter 2) and research questions related to this study, three conceptual models were proposed and empirically tested in the current study. As shown in Chapter 2, TE are predicted by various factors including students' behaviour and performance (S. Wang et al., 2018), which are all intrinsically related to student SC. Therefore, it was hypothesised that the TE measured in this study may be predicted by students' initial SC measured at Time 1. Additionally, it was hypothesised that TE were likely to be involved in the development of student SC (see details in Chapter 2), which led to the possible paths from TE to SC measured at Time 2. Following this view, three conceptual models below posit that TE may be related to student academic and nonacademic SC respectively, and as a whole (Figure 5.3).

Figure 5.3

Conceptual Model Showing the Hypothesised Relations Between the Variables Investigated in This Study



5.1.5 Data Analysis

Analysis of class-level TE in this study was the same as in Study 1, and was explained in Chapter 3. Regarding SC data collected at the two time points, CFA was conducted respectively. This step was to verify that each measurement model, leading to the final structural equation model, showed good model fit and represented the data well. Based on the CFA results, means and standard deviations for the questionnaire factors were calculated, along with McDonald's ω reliability estimates and Pearson's bivariate correlations. These preliminary descriptive analyses were then followed by SEM techniques based on the maximum-likelihood estimation method, which aimed to explore the relations between TE and student SC variables. The results of these tests are presented in the results section which follows. All analyses in this study were conducted using the statistical software IBM SPSS Statistics 27 and AMOS 27. Similar to Study 1, a range of fit indices (χ 2/df, TLI, CFI, SRMR, and RMSEA) will be reported in the results section of this chapter. As shown in Table 5.3, the acceptable levels of each of those indices were adopted from a number of reviews on goodness-of-fit indices and were used to determine the fit of the measurement and structural equation models presented in the results section.

Further, because the student number in each class was small (Median = 5) in the current study, it was not considered appropriate to nest the data and complete two-level analyses. This is because the power in multi-level models depends on the sample size of both level-2 units (which in this case is the teachers) and level-1 units (which in this case is the students).

Table 5.3

		8	
Indices	Perfect fit	Good fit	Rationale
χ2/df	$\chi 2/df \le 2$	$\chi 2/df \le 3$	R. B. Kline, 2015
RMSEA	$RMSEA \leq .05$	$RMSEA \leq .08$	Hooper et al., 2008
SRMR	$\text{SRMR} \le .05$	$\text{SRMR} \le .08$	Brown, 2015
			L. Hu & Bentler, 1999
NNFI (TLI)	$NNFI \ge .95$	$NNFI \ge .90$	Tabachnick & Fidell, 2007
			B. Thompson, 2004
CFI	$CFI \ge .95$	$CFI \ge .90$	L. Hu & Bentler, 1999
			Tabachnick & Fidell, 2007

Acceptable Levels of Fit Indices Regarding CFA

5.2 Results

5.2.1 Missing Data Analysis

As mentioned above, 482 students initially took part in the survey. However, data for 134 (27.8%) students were removed from the current study during the data-cleaning process because they did not complete the second survey. Before removing the data, independent sample *t* tests were conducted to compare the Time 1 SC of the students who dropped out halfway (n = 134) and all remaining student participants (n = 348). Table 5.4 provides the means and standard deviations for the data-set. As shown in Table 5.4, no statistically significant difference was found in the baseline student SC who withdrew from the study and the remaining students (p-value > 0.05 in all cases). Given that these data were missing at random, the research adopted the deletion method rather than imputation, because imputation methods risk underestimating variability and introducing bias to the data (Patrician, 2002).

Table 5.4

Means and Standard Deviations for Variables for Different Sample Groups

Variables	Participants in Time 1		Particip	Participants in	
	only		bo	both	
	(<i>n</i> = 134)		(<i>n</i> =	(n = 348)	
	М	SD	М	SD	-
Honesty/Truthfulness (Time 1)	5.740	.844	5.688	.806	.534
Emotional Stability (Time 1)	4.924	1.141	4.922	1.137	.985
Problem Solving (Time 1)	4.860	.763	4.722	.773	.083
Mathematics (Time 1)	4.884	1.232	4.805	1.216	.528
Verbal (Time 1)	4.719	1.080	4.617	1.112	.368

5.2.2 The Measurement Model on Student Self-Concept (CFA)

To establish the factor structure of the student SC questionnaire, each subscale (namely, Honesty/Truthfulness, Emotional Stability, Problem Solving, Mathematics, and Verbal) was examined first using a CFA technique. In CFA, a latent variable is measured based on several indicators which can be measured directly. In doing so, items were dropped if: (1) the item's factor loadings were less than 0.40, or (2) one (redundant) item in a pair of items had high residual covariance (mostly due to similar wording or content). Table 5.5 describes the fit indices for the 10 models, each of which had at least three items and showed acceptable fit to the data. Therefore, these 10 factors were retained for further analyses.

	Time	χ2 (df)	χ2/df	TLI	CFI	SRMR	RMSEA
	point						
Acceptable Values		<i>p</i> > .05	< 3	>.90	>.90	< .08	< .08
Honesty	T1	9.097(5)	1.819	.980	.990	.0267	.049
		<i>p</i> > .05					
	T2	7.374(5)	1.475	.987	.994	.0228	.037
		<i>p</i> > .05					
Emotional Stability	T1	52.211 (20)	2.611	.959	.971	.0368	.068
		<i>p</i> > .05					
	T2	54.415 (20)	2.721	.961	.972	.0357	.070
		<i>p</i> > .05					
Problem Solving	T1	5.775 (5)	1.155	.993	.997	.0237	.021
		<i>p</i> > .05					
	T2	7.283 (5)	1.457	.986	.993	.0256	.036
		<i>p</i> > .05					
Mathematics	T1	11.682(9)	1.298	.995	.997	.0205	.029
		<i>p</i> > .05					
	T2	26.674(9)	2.964	.977	.986	.0270	.075
		<i>p</i> > .05					
Verbal	T1	7.568(5)	1.514	.986	.993	.0236	.038
		<i>p</i> > .05					
	T2	11.460(5)	2.292	.973	.986	.0282	.061
		<i>p</i> > .05					

Goodness-of-Fit Indices for Each Subscale in Student SC Questionnaire

5.2.3 Descriptive Statistics

Table 5.5

Following the CFA results, means, standard deviations, and the McDonald's ω of internal consistency were calculated for all factors identified earlier (Table 5.6). Additionally, as shown in Table 5.7, Pearson's bivariate correlations among these factors and TE were calculated. Statistically significant correlations were found between TE and all SC factors at Time 2 (p < .05).

Table 5.6

Descriptive Statistics and McDonald's $\boldsymbol{\omega}$ by Factor

Scale	M (SD)	ω	Skew	Kurtosis
Nonacademic self-concept				
T1 Honesty	5.28 (1.13)	.74	39	54
T1 Emotional Stability	4.77 (1.17)	.87	30	07
T2 Honesty	5.27 (1.07)	.75	36	62
T2 Emotional Stability	4.76 (1.20)	.88	10	32
Academic self-concept				
T1 Problem Solving	4.86 (1.00)	.67	28	15
T1 Mathematics	4.53 (1.25)	.86	04	61
T1 Verbal	4.66 (1.13)	.76	12	17
T2 Problem Solving	4.89 (1.08)	.74	13	40
T2 Mathematics	4.37 (1.41)	.90	.09	74
T2 Verbal	4.76 (1.19)	.80	02	43

 $\overline{Note. N = 348}$ for all factors.

Table 5.7

Scale		Nonacademi	c self-conce	pt		A	cademic s	elf-concept			TE
	T1	T1	T2	T2	T1	T1	T1	T2	T2	T2	
	Honesty	Emotional Stability	Honesty	Emotional Stability	Problem Solving	Maths	Verbal	Problem Solving	Maths	Verbal	
Nonacademic self-concept T1 Honesty	-										
T1 Emotional Stability	.363***	-									
T2 Honesty	.543***	.220***	-								
T2 Emotional Stability	.321***	.639***	.386***	-							
Academic sell-											
T1 Problem	.321***	.511***	.207***	.358***	-						
T1 Mathematics	.254***	.381***	.164**	.286***	.532***	-					
T1Verbal	.315***	.381***	.203**	.281***	.519***	.150**	-				
T2 Problem Solving	.313***	.326***	.415***	.541***	.560***	.390***	.355***	-			
T2 Mathematics	.191***	.238***	.280***	.367***	.386***	.674***	$.108^{*}$.508***	-		
T2Verbal	.181***	.223***	.282***	.313***	.327***	.006	.616***	.464***	.088	-	
ТЕ	.204***	.147**	.276***	.251***	.072	.098	.110*	.169**	.184***	.120*	-

Pearson's Bivariate Correlations among the SC Factors and TE

Note. ****p* < .001, ***p* < .01, **p* < .05

5.2.4 Exploring Relations: A Structural Equation Model

Following the CFA, the SEM technique was employed to build the empirical models for this study. Models on how TE are related to student academic and nonacademic SC were explored separately and will be presented under subsections below as follows: (1) the relations between TE with students' nonacademic SC at Times 1 and 2; (2) the relations between TE with students' academic SC at Times 1 and 2; and (3) the relations between TE with both students' nonacademic SC at Times 1 and 2. As a measure of effect size, Cohen's (1988) conventional guidelines for determining the strength of relations between two variables (i.e., 0.1 for "small" or "weak," 0.3 for "medium" or "moderate," and 0.5 for "large" or "strong") were used when interpreting the strength of the standardised regression coefficient paths found in this study (Nakagawa & Cuthill, 2007).

Relations Between TE and Nonacademic Student SC. A structural equation model in which all paths from 1) nonacademic SC (Time 1) to TE, and 2) TE to the nonacademic SC (Time 2) were tested. Figure 5.4 is a diagram indicating the standardised regression weights for each path between the student nonacademic SC variables and TE. The fit indices indicated that the model met acceptable thresholds ($\chi 2 /(df) = 2.331$, p > .311; TLI = .905; CFI = .920; SRMR = .0560; RMSEA = .062).

The model showed that TE were positively predicted by student initial SC of Honesty ($\beta = .22, p < .01$). That is, when students had higher SC of their Honesty, their HT had correspondingly higher levels of TE. More importantly, a statistically significant path was found from TE to student SC of Emotional Stability measured at Time 2 ($\beta = .17, p < .001$), which means that, controlling for baseline SC, higher TE might help students develop higher SC of their emotional stability. The size of the standardised beta values reported indicated that TE were a weak-to-moderate predictor of student SC of Emotional Stability.

Structural Equation Model Predicting Paths Between Nonacademic SC Variables and TE



Note. ****p* < .001, ***p* < .01, **p* < .05.

Relations Between TE and Academic Student SC. Similarly, a structural equation model in which all paths from 1) academic SC (Time 1) to TE, and 2) TE to the academic SC (Time 2) were tested. Figure 5.5 is a diagram indicating the standardised regression weights for each path between the student academic SC variables and TE. The fit indices indicated that the model met acceptable thresholds ($\chi 2 /(df) = 2.151$, p > .341; TLI = .907; CFI = .923; SRMR = .0547; RMSEA = .058).

The paths between TE and student SC of Verbal Ability at Times 1 and 2 were both not statistically significant. However, statistically significant paths were found from TE to student Problem-Solving SC and Mathematics SC measured at Time 2. The results showed that, controlling for baseline SC, higher levels of TE predicted higher levels of ProblemSolving SC ($\beta = .13, p < .05$) and Mathematics SC ($\beta = .11, p < .05$). The size of the standardised beta values reported indicated that TE were a weak-positive predictor of student academic SC of their problem-solving ability and mathematics ability.

Figure 5.5

Structural Equation Model Predicting Paths Between Academic SC Variables and TE



Note. ****p* < .001, ***p* < .01, **p* < .05.

Relations Between TE and Student Academic and Nonacademic SC. Figure 5.6 shows a diagram of the empirical model built in this study, with fit indices indicating that the model met acceptable thresholds ($\chi 2 /(df) = 1.809$, p > .404; TLI = .889; CFI = .900; SRMR = .0567; RMSEA = .048).
As shown in Figure 5.6, regarding student SC at Time 1, only the path between TE and student initial SC of Honesty was statistically significant ($\beta = .21, p < .01$). More importantly, the paths from TE to student SC (Time 2) of Emotional Stability ($\beta = .16, p < .001$), Problem-Solving Ability ($\beta = .13, p < .05$), and Mathematics ($\beta = .11, p < .05$) were found to be statistically significantly positive. The positive paths between TE and student SC at Time 2 meant that, controlling for baseline SC, higher TE predicted higher student SC of Emotional Stability, Problem-Solving Ability, and Mathematics; or lower TE predicted lower student SC of these variables.

Figure 5.6

Structural Equation Model Predicting Paths Between Both Academic and Nonacademic SC Variables and TE



Note. ****p* < .001, ***p* < .01, **p* < .05.

In sum, the results from the CFAs and SEMs above showed that 1) TE positively predicted student SC (Time 2) of Emotional Stability, Problem-Solving Ability, and Mathematics; 2) TE were not statistically significantly related to student SC (Time 2) of Honesty and Verbal Ability; and 3) all variables of Time 1 student SC (except for Honesty) were not statistically significantly related to TE.

5.3 Discussion

This study explored the relations between TE and student SC which involved two dimensions: nonacademic SC (Honesty/Truthfulness and Emotional Stability) and academic

SC (Problem Solving/Creativity, Mathematics, and Verbal), controlling for beginning-year student SC. Three statistically significant relations were found between TE and SC (Emotional Stability, Problem-Solving Ability/Creativity, and Mathematics) at the end of one academic year (addressing Research Question 1). In other words, the result indicated that students in the classes of HTs with higher TE were more likely to develop higher levels of SC of these three factors in a timeframe of 10 months. However, no statistically significant relations were found between TE and student SC of Honesty and Verbal Ability at the end of the year. This led to the second research question concerning the mixed patterns of the relations between TE and different facets of SC. In the following sections, each of these variances will be discussed with respect to the TE effects, domain-specific differences, and cultural background.

5.3.1 The Statistically Significant and Positive Relations of TE with Student SC

The first research question concerned whether there were any relations between TE and student SC. It was expected that higher TE would manifest in the development of higher levels of SC (S. Wang et al., 2018). The current longitudinal investigation partly supported this theoretical assumption and previous TE findings, by showing that all paths from TE to Time 2 SC were positive, though two of them were insignificant (TE with Honesty SC and Verbal SC respectively). The statistically significant and positive relations between TE and three kinds of SC (Emotional Stability, Problem-Solving Ability, and Mathematics Ability) could be understood in relation to the TE effect also known as the Pygmalion effect. Numerous earlier studies (e.g., Rubie-Davies, Peterson, et al., 2015) have shown that teachers communicated their TE to students in many ways that influenced students' motivational characteristics, including their SC (Upadyaya & Eccles, 2015). These are, among others: more frequently praising students, embracing students' ideas, having eye contact with their students, and smiling more at them (Brophy, 1983). Through these behaviours, HiExTs' beliefs that all students could achieve at higher levels could be transmitted and projected into students' own perceptions of their ability and further promote their confidence and interest in a specific field. According to our results, this seemed to be the case because the models showed that student SC at Time 2 increased as TE increased.

Furthermore, Rubie-Davies (2007, 2014) found that HiExTs tended to scaffold their students with high-level challenging learning tasks and set clear learning goals so that students knew what skills they had already learned and what to learn next. These kinds of active learning experiences have been found to be positively related to students' academic SC

(K. M. Cooper et al., 2018). At the same time, studies have also suggested that students might participate more in the class activities to satisfy their teachers' expectations (Schnitzler et al., 2021). In other words, higher TE might predict improved student engagement (e.g., hand-raising) which has been reported to be positively related to student academic SC and their achievement (Schnitzler et al., 2021).

Going further, in addition to subject teaching, HiExTs have been reported as being more likely to create a warm and supportive class atmosphere, which contributes to a positive and caring relationship between students and teachers (Rubie-Davies & Peterson, 2011). In the current study, the more secure students felt about their abilities and their interpersonal relationships, the more possible it was that they displayed a higher level of SC as emotionally stable later on. At the same time, studies (e.g., Urhahne et al., 2011) have also suggested that even underestimated students could perform as well as overestimated students, but the former showed lower expectations for success and lower academic SC; instead, they experienced higher levels of test anxiety. This finding could partly explain the relations between TE with Emotional Stability SC and also two academic SC observed in the current study. This effect is illustrative when it comes to the context of HTs, where HTs are the ones who spend the longest time with their students in schooling and are responsible for all fields of students' development (including their mental health). Inconsistent with Y.-H. Chen et al. (2011), a study conducted with Taiwanese elementary school children and their HTs, the current study showed statistically significant and positive relations between TE and one nonacademic SC. The different results might be partly caused from the research designs where Y.-H. Chen et al. explored individual-level TE and general nonacademic SC whereas the current study measured class-level TE and domain-specific SC variables. The findings that emerged from the current study could also be due to the fact that this study was conducted in high schools (the educational stage emphasising academic achievement and peer competition; H. Li, 2017). As suggested by Good and Lavigne (2017), highlighting both achievement and competition can be key factors for stronger TE effects.

In addition to the TE effect discussed above, another key area worth mentioning is the positive associations observed between TE and student achievement (S. Wang et al., 2018), which could in turn predict individual SC. As indicated by Z. Li and Rubie-Davies (2017), TE positively predicted student academic outcomes after 1 school year. Therefore, the higher SC at the end of the academic year might be partly attributed to the increased student achievement predicted by the higher TE (higher TE \rightarrow increased academic achievement \rightarrow higher SC). Thus, a question remains here that although TE holds potential to explain

students' enhanced SC, it does not explain whether it is the TE itself that contributes to this pattern or the corresponding improved achievement playing a key role instead.

Finally, this study contributes to current literature by showing that the Pygmalion effect exists not only among primary school students, but also in high school contexts. At the same time, the effect sizes of this phenomenon ($\beta = .16$ in the Emotional Stability SC, $\beta = .13$ in the Problem-Solving Ability SC, and $\beta = .11$ in the Mathematics SC) are comparable with what have been found among younger children in some of the previous research (effect sizes of 0.10–0.20; see, e.g., Gentrup et al., 2020). In this regard, this study went against the hypothesis that high school students are more resistant to biased TE than their counterparts in lower year levels (e.g., Y.-H. Chen et al., 2011; Jussim, 2012). The associations found in this study could be largely due to the context of Chinese HT in this research-the teacher who has the most intensive interactions with students in their daily school life and thus might have greater influence. Despite all this, a stronger predictor of student end-of-year SC than TE was beginning-year SC. This is probably unsurprising; nevertheless, it has been evidenced that TE can nudge the trajectory of student SC (Ding & Rubie-Davies, 2019). Additionally, the modest (yet still robust) effect size could be largely due to the short time period of one school year (10 months) covered in this empirical research. However, it seems necessary to consider that TE effects could accumulate and become stronger over longer periods (Rubie-Davies et al., 2014). Combining previous findings with that of this study, it could be argued that over and above student baseline SC, higher levels of TE can positively predict high student SC over time.

5.3.2 Variance in the Relations Between TE and Student SC

The second research question concerned whether there were any differences in the relations between TE and specific facets of SC. Based on previous findings in the field of TE effects, a mixture of patterns was anticipated in which most SC dimensions were predicted by TE despite the possibility that they might develop in inconsistent ways with higher or lower values on one dimension than on others. According to the model in this study, mixed patterns were observed with class-level TE of HTs predicting some, but not all, aspects of student SC. As discussed below, the underlying reasons related to two insignificant relations (TE with Honesty SC and Verbal Ability SC) could be understood against the formation of the SC, the domain-specific differences, and the contextual factors.

As one nonacademic SC, Honesty SC at Time 2 showed no statistically significant relation with TE. This result could be interpreted in two ways: 1) considering the baseline

Honesty SC was fairly high among all three TE groups (Mean = 5.28), there may be a ceiling effect where it would be difficult for Honesty SC to get even higher; or 2) students of HTs with different TE levels did differ in their cheating behaviours but these behaviours did not threaten students' SC as honest (e.g., students in the lower TE classes might have cheated more but thought it was acceptable). As shown in the literature, one of the key processes of maintaining SC is "categorisation of actions" (Mazar et al., 2008, p. 22), in which individuals regularly reframe the actions that violate social norms as acceptable. For example, some students might categorise looking at another student's answers as a form of cooperation or group work. As such, the cheating behaviour could be perceived as something endorsed by teachers instead of something dishonest. Similarly, when it comes to copying homework, an especially common form of dishonesty (McCabe et al., 2012), students who think that homework is unimportant and low-stakes might not take this cheating behaviour as seriously wrong (Rettinger, 2017). In this regard, despite the differences in cheating behaviours, in reality, there would be no salient variance reported in the level of student's Honesty SC. Therefore, it will be important for future research to explore 1) whether there are any relations between TE and students' actual cheating behaviours, as opposed to self-reported SC; and 2) whether there are any differences in the process of forming Honesty SC (especially the categorisation of actions) of students with teachers having different levels of TE.

The mixed pattern was also shown in the field of academic SC, where TE was directly related to student perceptions of their Mathematics and Problem-Solving Ability/Creativity but not student SC in Verbal Ability. This finding is interesting given some previous studies have shown direct associations between student Reading SC and class-level TE (Rubie-Davies, 2006). Again, there could also be a ceiling effect for Verbal SC in this project (*Mean* = 4.66). At the same time, another possible explanation for the insignificant relation between TE and Verbal Ability SC could be understood in relation to the mathematics preferences in the context of Chinese high schools. As suggested by earlier studies (S.-K. Chen et al., 2013; Dai, 2002), Chinese students, teachers, and parents usually value mathematics above other subjects, therefore, the feedback that students receive regarding their mathematics performance could be more salient compared to the performance in language. In this vein, there might be fewer possibilities for Chinese students' Verbal SC to be malleable at high school compared with their SC of mathematics and problem-solving ability. Thus, the role of contextual factors might explain the inconsistent findings of this study and that of previous research. This is a message worth listening to when exploring the TE effect in the Asian

context that the impact of other factors, such as subject preferences and parents' involvement, should be considered.

Generally, this study supports the idea that HTs' class-level TE can become a selffulfilling prophecy for student SC. Thus, it is important for teachers to be aware of the significance of having high TE for the development of all their students, which is not only in relation to students' academic SC but also in terms of their psychological well-being. Following the detailed consideration of the reasons underlying the diverse patterns found in this study, more complicated culture-related and domain-specific drivers should be taken into account to explain student SC. Further studies might elaborate on this topic and explore several potential causes together to explain why students SC is related to TE in specific ways.

5.4 Limitations

Even though this study addressed many gaps in the literature on SC and TE, caution should be exercised for some methodological issues when interpreting the findings. First, the research was correlational rather than experimental, which means that causal claims cannot be made. The findings highlighted that class-level TE of high school HTs was indeed a positive predictor of later student SC. However, other mechanisms cannot be ruled out through which TE predicted SC across multiple processes of forming or maintaining SC over time. For example, it might be that students benefitted from the advanced academic activities designed by their HiExTs, which improved their academic performance and then, in turn, boosted their respective SC. Thus, the observed relations in this study need to be further investigated to fully explain how these relations work. Nevertheless, in controlling for prior levels of SC, this research was able to show that the relations between TE and SC went beyond individual differences in students' previous SC.

The second, and perhaps more worrisome, limitation of the study was that there was a relatively high level of dropout from the research (134 students, 27.8%). This could be largely due to the longitudinal design and the practical constraints (the whole research was conducted remotely due to COVID-19). Totally relying on an online survey could have compromised the response rate, because as indicated by Baruch and Holtom (2008), the average response rate for in-person surveys (62.4%, SD = 16.9) is higher than that for online ones (38.9%, SD = 15.1). Thus, for future investigations, it is suggested to use both in-person and online surveys to overcome this limitation. However, despite this reduction in participants in this study, the sample size was still comparable to the sufficient sample size recommended for SEMs (Comrey & Lee, 2013). Furthermore, additional analyses suggested

that the students who dropped out halfway did not significantly differ from the remaining students regarding either academic SC or nonacademic SC.

Another limitation is related to the hierarchical data structure in this study where students were clustered in teachers. As explained previously, multilevel modelling was not used in the current study due to the small sample size in each class, and so any possible conclusions must be tentative. Future research with larger numbers of students in each class would enable multilevel modelling techniques and could lend further weight to the current study. Similarly, potentially confounding variables (e.g., students' socioeconomic status) could be identified and measured with a larger sample of teachers and students.

Finally, only self-reported measures were used in this study. There might be inconsistencies between the actual level of SC and what the student reported. For example, reporting on how honest one feels about oneself may cause students to answer the question in a socially desirable way. The lack of a follow-up qualitative design to explore the observed relations means that possible reasons to explain the findings can only be speculated upon. Therefore, in future explorations, it is recommended that researchers have a follow-up qualitative design to investigate and explain the patterns and results that emerged from the models.

5.5 Implications for Future Research and Practice

As well documented throughout the education literature, individual SC is significant in students' development, including their academic performance (Szumski & Karwowski, 2019) and mental health (Xu et al., 2019). The results of this study extended existing findings on the relations of TE to SC by showing the positive relations of class-level TE with student SC in the context of Chinese high school HTs. Despite general agreement on the predictive role of TE in student SC, there are inconsistencies in how TE has been shown to be related to specific domains (for a review, see S. Wang et al., 2018). This seems to be the case in the current study because different patterns from some of the previous findings were also revealed. The differences between this research and previous explorations might be caused by contextual characteristics, such as the year levels of students, the particular kinds of teachers investigated, and cultural differences. As such, it is important for future research to compare how TE effects work in different contexts. Further examination of these results would be highly useful and might provide insight into how an intervention tool could be designed to leverage TE and improve student SC in different contexts. In addition to the attention to academic SC, an increasing number of studies have started to point toward the importance of nonacademic SC by showing its positive relations with academic SC and academic achievement (e.g., Ogle et al., 2016). The current study contributed to this field by illuminating the significance of TE in relation to student Emotional Stability SC. Nevertheless, the findings that emerged from this study should be interpreted with the particularity of high school HTs who take care of students' overall development, including their emotions and mental health. However, the results signify the possibility that students' nonacademic SC could be improved within a supportive, caring, and high-TE framework. As suggested by Flaherty (2020), student emotions are underresearched regarding their dimensions, antecedents, and functions in various academic settings. Current interventions commonly attend to alleviating student negative emotions, such as their test anxiety, rather than the ways that positive emotions for interventions on student emotions, as opposed to the current priority on the enhancement of skills and performance on examinations (Usher, 2015).

Furthermore, potential avenues not evaluated in this study are the roles of subject teachers, parents, peers, or students themselves in both the formation of student SC and in TE effects over time. For example, parents' involvement might mediate the construction of student SC (Silinskas & Kikas, 2019), especially given that home-based types of involvement are relatively intensive in East Asian contexts (S. W. Kim, 2020). Thus, trying to understand what parents may be doing in the home environment to foster students' ability as well as their perceptions about children's competence are important areas to explore. Additionally, only academic TE were measured in this study, it is worth exploring how teachers' (especially HTs) nonacademic expectations of their students mediate student SC. Future research on the mediating role of these factors is required to fully untangle the mechanisms of TE effects on student SC.

Finally, although strong causal statements are unwarranted considering the current research design, there are indeed plausible reasons to believe that TE and specific behaviours play a role in students' SC, at least to some extent. However, the researcher did not wish to oversimplify the mechanisms by which TE influenced SC, rather, she encourages future studies to explore relevant factors further to investigate the complexity of why an individual student might feel confident, interested, and would seek to learn more. Bearing in mind that beliefs do not exist in isolation, it will be important for future research to focus on beliefs more broadly. Tracking various variables (e.g., motivations) together using additional time

points will help to reinforce and to better understand the associations identified here. These investigations will also help researchers to understand when and how to best leverage these relations within a high-expectation framework.

5.6 Conclusion

The present research investigated the relations of SC (both academic and nonacademic) with class-level TE. This study contributed to the nascent stream of explorations of Chinese high school HTs' TE relate to their students' SC. The structural equation models illuminated that some of both academic SC (namely, Problem-Solving Ability SC and Mathematics SC) and nonacademic SC (Emotional Stability SC) were malleable and could be positively predicted by TE. The findings emphasised the important role of TE in the development of student SC. This study contributed to the bridge between psychologically oriented studies in terms of student beliefs and class-level TE. The association of these two research strands also shed light on the design of further interventions which aim to improve student SC within a TE framework.

Chapter 6: General Discussion

The research in this thesis has explored the role of class-level TE in teacher emotions and student SC in the context of Chinese high school HTs. A major claim has been that the TE were significantly associated with teacher emotions and positively predicted some facets of their students' SC outcomes. At the same time, the patterns between TE and teacher emotions were explored and explained in more depth through individual interviews. Teachers with different levels of TE seemed to vary in their role identification and their capacity for resilience in their practices.

This final chapter aims to draw together the main findings from all three studies of this thesis and describe and discuss why and how this doctoral project provides significant contributions and insights into the fields of teacher and student beliefs. In doing so, this thesis builds and expands on the theoretical and empirical work cited in previous chapters. As shown below, the first section of this chapter will provide a brief revisit of the findings from all three studies, followed by a general discussion of the overall findings. The third and fourth sections will demonstrate the theoretical and educational implications of these findings. The limitations of this doctoral project will then be presented in the fifth section, followed by concluding thoughts.

6.1 Overall Summary of the Key Findings

As the initial step of the project, the first quantitative study (Chapter 3) measured the TE and emotions of HT participants and then examined the associations between these two factors, both of which are documented as significant in influencing teaching practices and students' outcomes (Frenzel, 2014; Rubie-Davies, 2017). The SEM showed that 1) TE positively predicted pleasant emotions of joy, pride, and love; and 2) TE negatively predicted negative emotions of anger, fatigue, hopelessness, and anxiety. In other words, the findings indicated that teachers' positive emotional experiences were related to higher levels of TE. As detailed in the discussion section of Chapter 2, this study meaningfully contributed to the scant base of knowledge on the interplay of these two factors in the teaching profession.

Further, based on the results regarding the TE group and associations between TE and their emotions found in Study 1, the second study (Chapter 4) drew upon individual interviews to establish understanding of the interplay of these two factors in depth. The interview results indicated that, compared to LoExTs, HiExTs appeared to highlight the student role instead of their own in education and suggested a more advanced capacity in

their emotional strategies, teaching approaches, and their use of contextual resources. These differences could partly explain the emerging patterns of TE and teacher emotions from Study 1, where a higher level of TE predicts a higher level of positive emotions and a lower level of negative emotions. Additionally, the findings from Studies 1 and 2 have practical implications which suggest teacher emotions might be improved by training teachers to behave in alignment with high-expectation principles (Rubie-Davies, Stephens, & Watson, 2015).

Finally, the last study (Chapter 5) in this thesis mapped out how student nonacademic SC (Honesty/Truthfulness and Emotional Stability) and academic SC (Problem Solving, Mathematics, and Verbal) appeared to alter according to the TE of their HTs. Controlling for beginning-year student SC, TE were found to be statistically significantly related to three facets of student SC (Emotional Stability, Problem-Solving Ability, and Mathematics) at the end of the academic year. In this regard, students in the classes of HTs with higher TE were likely to form higher levels of SC of these three factors over the 10 months from the first to the second measurement. However, no statistically significant relations were found between TE and student SC of Honesty and Verbal Ability. As discussed in Chapter 5, each of these variances should be understood in light of contextual considerations such as domain-specific differences and cultural background.

6.2 Overall Discussions of the Key Findings

Two themes emerged from interpretations and reflections on the key findings of the whole project: (1) the promising yet intricate nexus of TE, teacher emotions, and student SC; and (2) contextual interpretations under the umbrella of the TE framework. As shown in the following sections, the first theme concerns the overall picture from the current project where the significance of TE in teaching and learning was highlighted but variations and complexities were also captured. Furthermore, as shown in the discussion sections of the previous chapters, despite numerous consistencies between the present findings with TE theories, inconsistency and uniqueness also emerged from each study. Therefore, the second theme relates to the contextual considerations within the overall TE framework in which the particularity of Chinese HTs and the sociocultural background will be discussed. Together, this section discusses the findings across all three studies in a general way to portray the TE effects in the current context. It shows what is specific and unique to the Chinese high school, and specifically to HTs. The discussion will also show how findings from this project

contribute to the existing literature and how they could be used as a leverage in future interventions and in modifying daily HT practices.

6.2.1 Promising-Yet-Intricate Nexus

The findings from this project integrated and complemented that of previous literature and the TE framework by incorporating TE, teacher emotions, and student SC into one investigation. As detailed in Chapters 3 to 5, there is considerable alignment between the current findings and the existing literature. To illustrate the combined findings of all three studies in a clearer way, a schematic model (Figure 6.1) was developed based on both the results of this project and the existing framework presented in the second chapter. As shown in the model, this research complemented and extended existing literature by exploring TE effects at both the teacher level (left part of the model) and the student level (right part of the model). In terms of teachers, associations between TE and their emotions were found, which appeared to be mediated by teachers' role identifications and their capacity for resilience in their practices. The integration of TE and teacher emotions may have been reflected in teachers' behaviours, resulting in different teaching practices and teacher-student interactions. As shown in the literature, students are able to perceive TE through the nuance of teachers' behaviours (Babad & Taylor, 1992). Therefore, at the student level, their learning outcomes (SC explored in this research) could be associated with their teachers' beliefs and emotions. Overall, the model showed that the TE effects not only existed in student development but could also relate to teachers' own emotions and work as an integral part of teaching and learning. The model is discussed in detail in the following sections. As elaborated below, the overall results of this research either supported or extended existing findings on TE effects by showing differential associations of TE with teacher emotions and student SC.

Figure 6.1

The Role of Teacher Expectations in Teaching and Student Outcomes



Note. Three factors in grey shadow are the ones explored in this thesis.

One overarching goal achieved in Studies 1 and 2, as shown in the left part of Figure 6.1, was to do with bringing together two psychological variables (TE and teacher emotions) that are not often explored within the same context in the literature. Specifically, although a large body of literature recognises the role of TE in student development (S. Wang et al., 2018), less is known about how TE might be related to the emotional experiences of the teachers themselves. Studies 1 and 2 contribute to the existing literature by showing why and how teachers' positive emotions might be enhanced and, conversely, how their negative emotions might be alleviated to a certain degree, as TE of a given class increased. Findings from these two studies are important as they suggested that TE might be a function of how teachers experience their profession. In other words, TE could be perceived as one of the keys to understanding teacher emotions or even work as a leverage to improve their emotional experiences. As will be discussed later in Section 6.4 (Practical Implications), the findings on the relations of TE and teacher emotions highlighted the importance of incorporating a TE framework into professional development because it could work as a way to prevent teacher burnout and reduce teacher shortages. This is especially important given the higher occupational strain and stress HTs encounter, as reported in the literature (Baeriswyl et al., 2021), and the lack of research exploring these two factors together in psychological research. The theoretical and practical contributions will be elaborated further in the following sections.

When moving TE effects beyond teachers to students, as shown in the right part in the model (Figure 6.1), this project resonated with extant findings that TE could positively predict the development of student SC (e.g., Rubie-Davies et al., 2020). Regardless of the complexity in the relations presented and discussed within Study 3, the overall pattern confirmed the significance of TE in the development of high school students' SC, as has been found for their younger counterparts. As presented in the model, how TE are translated in reality and then predict student outcomes should be understood in relation to TE theories that students are able to perceive TE from interpreting differential teacher behaviours (Babad & Taylor, 1992), which may in turn mediate students' sociopsychological factors such as SC (Rubie-Davies et al., 2020) and self-efficacy (Karwowski et al., 2015). As a whole, the findings from the current project emphasised the significant role of TE in the development of both teachers and students. The emerging findings from all three studies echoed and complemented the theoretical framework proposed by other researchers in the literature review (Section 2.2). For example, Studies 1 and 2 contributed to the findings on the relation between teacher beliefs and emotions where the reciprocal relations have been found earlier

between self-efficacy (one construct of teacher belief) and their emotions (Burić et al., 2020).

However, despite the promising patterns discussed above, the findings from the current research also portrayed a complex web of interactions between TE, teacher emotions, and student SC, indicating an intricate mechanism standing behind the overall TE-effect framework. The emerging interrelations of TE and teacher emotions from the current studies suggested that teacher emotions were not simply influenced by met and unmet expectations. Instead, emotions and TE mutually influenced each other in a complicated way that took place in between daily teacher-student, teacher-parent, and teacher-teacher interactions. For example, how HTs interpreted their students' misbehaviours (e.g., a kind of provocation or a lack of security) appeared to orient some of the pedagogies that took place during these events. In other words, TE and the way they were interpreted or manifested in class played a critical role in their transition to teacher emotions. Nevertheless, although Studies 1 and 2 shed some light on this topic, the real mechanisms could be much more intricate than what has been found in this research. As mentioned earlier, the relations between TE and teacher emotions have been underexplored in the literature, which means that possible reasons to explain the current findings could only be speculated upon with reference to emotional theories (e.g., Scherer, 2009) and TE theories (e.g., Rubie-Davies, 2014), respectively, and associations found between teacher emotions and other teacher beliefs (e.g., self-efficacy; Burić et al., 2020).

Furthermore, a mixed pattern was also observed in the relations between TE and student SC. Study 3 showed the differences in the associations between TE with different components of SC, which draws attention to the many facets of student SC. In this regard, student SC should be considered as heterogeneous and multidimensional and its relations with other factors should be interpreted with caution. In addition, the results of Study 3 also showed inconsistencies with those of other research in different contexts, which highlighted that TE effects are likely to vary across contexts and/or individuals. Hence, although the model proposed in Figure 6.1 provides a concise way to understand the findings of the whole project, the mechanism behind the model should never be oversimplified. This study has only begun to unravel some of the complexities of these associations during some specific processes of the whole TE effects framework. Only by understanding the interrelations of all fields of TE effects and the perspectives of all involved is it possible to comprehend this unique process.

6.2.2 Contextual Interpretations

The current project explored three psychological variables (namely, TE, teacher emotions, and student SC) within the context of Chinese HTs, the group that has been neglected in previous TE literature. Given that all three factors are embedded in the context, the results should be interpreted with regard to the contextual factors, such as Chinese sociocultural (e.g., Confucianism, Communism, collectivism) and institutional (e.g., examination-oriented education) contexts. This is especially important considering the misalignments found in the results of this research from that of other contexts. Therefore, this section discusses the findings with the contextual considerations to show what is specific and unique to Chinese HTs and the educational climate.

TE and Chinese Approaches to Teaching. Influenced by the Confucianism tradition, the common teaching beliefs among Chinese teachers involve ideas such as "failure is the result of students' laziness" and "children are spoiled if praised" (Yin & Lee, 2012), which was also mentioned in some interviews in this project (Study 2). Following these views, Chinese teachers tend to suppress their spontaneity (Z. Wu & Chen, 2018) and behave in a neutral or stern way to maintain their authority in class (Yin, 2016). Most of these approaches seem to imply a harsh and oppressive emotional environment, which runs counter to high-expectation principles in the literature. However, despite repeated criticisms of the oppressive and unidirectional teaching approaches in China (H. Li & Du, 2013), the teacher–student interaction is not as detrimental as has been alleged by some Western theories (Yin & Lee, 2012). Instead, the teacher–student relationship is usually harmonious and close, which is typically marked by teachers' strong sense of responsibility and a caring climate (Yin, 2016).

Nevertheless, the particularity of teaching approaches (e.g., neutral) embedded in Confucianism does not mean that Chinese teachers are more resistant to biased TE or that it is less likely for them to communicate their expectations to their students. Instead, this doctoral project has not only uncovered Chinese HTs who do have differing TE (Study 1), but also explored how these TE then appear to translate into their practices (Study 2), which in turn predicted student SC outcomes (Study 3). The findings on the communication of TE in this research are critical because they suggested that, even in a more emotionally reserved culture where people tend to hide their feelings (Yin & Lee, 2012), HTs do develop biased TE and manage to convey their TE to students during their daily interactions. However, as reflected in the interview data in Study 2, it could be argued that the behaviours conveying TE in the Chinese context might be slightly different from what have been found in Western contexts. For example, regardless of TE levels, HT participants appeared to be more inclined to praise their students when they had made significant progress in tests, instead of making frequent and direct compliments to them in their daily interactions (Study 2). Additionally, HT participants rarely mentioned grouping students in class (which is one important instructional strategy in the TE framework), which could be partly due to the large class size in Chinese high schools. Although how Chinese teachers communicate their TE to their students was not the focus of this research, the examples here suggested possible variance existing in TE communications in different contexts. Therefore, it would be interesting to explore further to see, compared to existing literature in Western cultures, how teachers in different cultures convey their TE and influence their students.

Furthermore, another significant component of teaching in China is "heartconsuming" labour (Yin & Lee, 2012), where teachers devote themselves to the job fully (J. Chen, 2016) and assume high levels of responsibility for student development (J. Chen, 2015). This is partly driven by the collectivist culture where teacher–student relationships resemble a familial relationship (H. Li & Du, 2013) or paternalistic leadership that combines strong authority with fatherly benevolence (X. Zheng et al., 2020). This is especially typical in the HT context where HTs tend to be perceived and portrayed as parents of their students. This paternalistic leadership, on the one hand, explains the close teacher–student relationships and teachers' profound impacts on student development (for example, the role of TE in student SC found in this project); on the other hand, it also constitutes a source of teachers' fatigue and conflicting demands on their time.

However, within the same context of "heart-consuming" labour, a different trend was observed in the interview data from HT participants in the current research. The association between TE and role identification found in Study 2 showed that HiExTs were more inclined to perceive themselves as a "sister/brother" or "listener" of their students instead of their traditional roles as parents or guardians. The disparities in role identification highlighted the interrelations between TE and other teachers' beliefs, for example, their self-perceptions. Furthermore, this discrepancy in role identification also suggested that HTs with higher expectations tended to incorporate a more equal teacher–student relationship into the traditional familial culture, which could in turn contribute to teachers' emotional experiences and their teaching practices. As suggested in the literature, in the past few decades, China has launched initiatives to encourage more student-centred teaching approaches (H. Li & Du, 2013). Thus, it is possible that the changes captured in Study 2 could be partly attributed to how today's younger generation of teachers think about their jobs differently from the

perceptions of previous generations. Despite the various potential attributions, however, it is acknowledged that being a HiExT implies the potential to balance the advantage of democratic approaches to teaching with the traditional caring components of the HTs' job. Therefore, in addition to the mediating role of TE in teacher emotions discussed in this thesis, it is exciting to see that increasing HTs' awareness of their TE might also contribute to the discourse of Chinese education changing from a "parental directing" style (L. Gao, 1998) to a combination of the traditional caring model with more student-centred approaches.

TE and the Tensions Between Academic and Nonacademic Development. Given that subject teachers in China are mainly responsible for student academic performance rather than their nonacademic development, this project focusing on HTs provided a possibility to explore how TE related to both components in the Chinese context. HTs' commitments and their differences from subject teachers have been detailed in the literature review (Chapter 2). In summary, the most salient particularities of HTs are their unparalleled close relationships with other stakeholders in schooling and their responsibility for student overall development, which suggests an excessive workload, on one hand, but also a chance to make real differences in learning outcomes, on the other hand. These particularities have also been reflected in the results of this project. Findings across all three studies identified the reciprocal relations of HTs' experiences with student overall development, especially their nonacademic outcomes. For example, the emotion questionnaire and the interview results revealed that students' misbehaviours constituted one major source of HTs' negative emotions. Similarly, the role of TE in student nonacademic SC demonstrated that HTs indeed played a critical role in students' emotional experiences.

However, as revealed in the interviews, HTs reported inadequate support or preparation for their nonacademic accountability (e.g., moral education and classroom management). Whereas teachers nowadays are starting to increasingly attend to students' well-being and overall development (Yan, 2015), it is reported in the literature that the public, including parents, have placed more attention on student academic performance (Lee & Yin, 2011), especially at the high school stage where students are required to sit a national examination to be admitted to universities (M. Li et al., 2018). However, when treating the results of the current project as a whole, it is interesting to find that these two parts (academic and nonacademic development) are likely to be complementary rather than oppositional. Specifically, data in this research showed that teachers' teaching practices, including their subject teaching in class, were influenced by student nonacademic behaviours, and vice versa. Bearing in mind that the TE measured in this project was only about students' academic ability, its statistically significant relations with teacher emotions and student nonacademic SC could be even more thought-provoking. In other words, teachers' academic expectations of students predicted student nonacademic SC development, while student nonacademic behaviours constituted a major influence on teachers' emotions and their practices in class—a key predictor of student academic outcomes. It is hoped that this interesting finding might encourage people to stop treating these two fields entirely separately, or at least help different stakeholders to increase their awareness of the importance of both components.

TE and HTs' Dilemma. HTs' negative emotions and their vulnerability have been explored in previous literature and various sources have been found (e.g., Huang & Gove, 2015). As presented in the literature review chapter, these sources ranged from structural conditions (e.g., ambiguity in commitments) to sociocultural factors (e.g., strict public expectations). Some of these factors were confirmed in the questionnaire items in Study 1 and especially the interview data in Study 2. However, finding ways to address these concerns could be a long-term topic which is beyond the scope of this research. Instead, this project shed light on the possibility that HTs could, at least to some degree, prevent themselves from the emotional dilemma (e.g., professional vulnerability) by teaching in line with high-expectation principles. The differences observed between LoExTs and HiExTs and the possible contributors to positive emotional experiences have been carefully discussed in Study 2. Based on these findings, TE could be regarded as one kind of leverage to prevent teachers from emotional trauma. Section 6.4 will discuss how further interventions could be developed by integrating current findings. To avoid repetition, overlapping details are not described here.

Combining the results across all three studies, one message worth attending to is that a balance should be achieved between what Chinese HTs' are *expected to do* and what they *can do*. As revealed in the interview data (Study 2), one main tension resulting in LoExTs' negative emotions was the gap between high external demands or expectations from parents and teachers' relatively low beliefs in students' ability. External expectations in education have also been reported for many years in the Chinese literature (J. Chen, 2016), which suggests that Chinese parents usually have high expectations for their children's achievement and thus have a strict or sometimes unrealistic demand for teacher responsibility (Y. Wang et al., 2015). This phenomenon has been intensified in the one-child generation (Fong, 2004) which was the student sample explored in the current project. The need to interact and cooperate with parents is particularly relevant for HTs because they are the ones who keep parents continuously informed and work as a bridge between different stakeholders in schooling, which applies to both subject-specific and interdisciplinary issues (Baeriswyl et al., 2021). However, it is important to distinguish the external expectations, from the public and parents, from the TE explored in current research. The high expectations from parents reported in the literature are usually associated with strict demands and unjustified blame directed at teachers (Z. Wu & Chen, 2018), which more resemble the requirement that "I want my child to achieve higher" than the TE explored here that "I believe these children can achieve higher." Bearing this in mind, it is important for HTs to understand the differences between the expectations held by various stakeholders and then figure out an effective manner to respond to these external demands. To achieve this, realising the nature of their own TE and the associated effects could be the first step. Given the close relationships between HTs and parents, communicating the discrepancy in expectations of children to achieve a more holistic educational view could be another step forward.

6.3 Theoretical Contributions

Compared to previous research on TE, the spectrum of this project extended the literature in three ways. Firstly, this research was conducted to complement the current TE literature by providing possible associations of TE with different variables. The interaction of teacher emotions and their expectations of students was a focus in Studies 1 and 2. Although both variables have been considered as influential factors in teaching practices, earlier studies have underexplored how TE and emotions interact with each other. Investigating TE effects in the literature has traditionally been done with student outcomes (e.g., Szumski & Karwowski, 2019), and less emphasis has been given to teachers' own emotional experiences. This project filled this gap by comprehensively exploring the interplay of these two factors. Additionally, TE effects were studied with respect to students' SC outcomes in Study 3. Although changes in intelligence or achievement in relation to TE have clearly drawn the most research interest (e.g., Agirdag et al., 2013; Thomas & Strunk, 2017), expectancy effects on student SC (especially nonacademic facets of SC) have been less frequently investigated empirically (S. Wang et al., 2018). Student SC, however, deserves the same attention as cognitive variables given its key role in student overall development, including academic achievement (M. Richardson et al., 2012) and mental health (Xu et al., 2019).

Secondly, this research is seen as important because it has revealed novel insights into researching TE effects in the Chinese high school context and HT context. As noted in the literature review (Chapter 2), only a small number of relevant studies have been conducted in

the Chinese context, with their foci being on teachers in university (Z. Li & Rubie-Davies, 2017, 2018)) and junior high school (Ding & Rubie-Davies, 2019; S. Wang et al., 2021;S. Wang et al., 2019, 2020). Explorations within high school education are of significance due to its crucial role in students' transition from fundamental education to the tertiary level. Therefore, one major contribution of this project, which will inform future researchers, lies within the discussion of its findings in relation to the Chinese high school context. Going further, as discussed earlier, many scholars have conducted similar research on TE effects but obtained varying conclusions in different settings (for a review, see S. Wang et al., 2018). The findings from this thesis echo this mixed picture by showing what was specific to the Chinese high school context and HT context compared with that explored by the available theoretical and empirical literature in the field. Therefore, it would be worth further exploring how TE might be differently manifested, conveyed, and function in various contexts.

Thirdly, what also distinguishes this research is the findings of TE effects on both students and teachers. Reviewing the literature on the associations between various psychosocial variables suggests that a lot is already known about TE effects on an individual student or a specific learning outcome (S. Wang et al., 2018). What is suggested by this thesis, however, is the need to start questioning what TE means to teachers themselves. In doing so, it is hoped that the literature would become refined to focus on teachers' inner selves where TE effects initially happen. The predictive role of TE found in a range of teacher emotions suggested that researchers need to move away from investigating direct impacts of teachers on students and instead consider the interactions between a combination of psychosocial constructs of teachers' own experiences and how these interactions function in teaching before exerting impacts on their students.

6.4 Practical Implications

Numerous practical implications have been discussed as part of the chapters corresponding to Studies 1 to 3. As a whole, exploring the relations between TE, teacher emotions, and student SC development is seen as important, and informative, to various audiences. Whereas this research focused on the group of HTs, as shown below, the results have significant implications for multiple stakeholders who are directly and indirectly involved in teaching and learning in and outside schools.

6.4.1 Implications for Teacher Education and Professional Development

The present project aimed to explore multiple links between individual beliefs and emotions taking place in teaching and learning. This research is seen as important because it shows that TE could be a possible lever to regulate teacher emotions and enhance student SC. It is hoped that further teacher education and professional development could integrate the findings of this research to equip teachers with high-expectation principles in their practices.

Despite the emotional dilemma teachers are experiencing and the significant role of teacher emotions in teaching and learning, the existing teacher education programmes unfortunately rarely weave such content into their programmes (Yin, 2016). Before introducing specific emotional regulation strategies, it is essential for teachers to first understand the nature of their emotions and individual beliefs, for example, how their emotions are triggered and what kind of mediators work in their experiences. It is hoped that the findings from the current project could help relevant content to be developed and inserted into initial teacher education and professional learning development. It is only when teachers are able to resolve their inner-psychological tensions, that they can perform in a way which indeed makes a difference to teaching effectiveness and to teachers' own experiences and lives. Additionally, sociocultural factors and psychological factors should also be considered given their mediating role in individual's emotions (Yin, 2016).

Understanding emotions and beliefs is one thing, but the development of advanced teaching strategies in relevant settings is a separate issue. Teachers need to be facilitated to master emotional regulation competence and high-expectation strategies to meet the challenges in reality rather than just maintaining the status quo. This is especially important in the context of the instructional changes needed during the COVID-19 pandemic and the consequent emotionally charged atmosphere in schools. It is hoped that teacher education programmes will equip teachers with specific techniques (combining the themes that emerged from Study 2) so that teachers are prepared to respond to their emotions and handle emotion-eliciting situations in a more effective manner. In doing so, a sustainable educational change and consequent improved learning outcomes can be achieved. In sum, it is hoped that these findings will help researchers develop interventions that will make teachers aware of what is happening emotionally and unconsciously, what they can do to improve their emotions, and the various practices they could potentially improve to strengthen their students' positive SC.

6.4.2 Implications for Stakeholders in Education

Whereas a number of studies have argued that teachers' negative emotions are embedded in the teaching profession (e.g., Frenzel et al., 2021; Kelchtermans, 2005), it is hoped that sharing the findings of this research with teachers will help them to be more aware of their TE and emotional experiences, and of how these factors influence their professional lives, and facilitate an increase in their capacity to self-regulate those vulnerabilities. Furthermore, emotional competence and high-expectation principles could help teachers to improve their teaching practices and help teachers to achieve better student outcomes.

In addition to the implications for teachers who are currently experiencing emotional pressures, it is hoped that the findings of this research will call for a collaboration between teachers with their school leaders and other colleagues (e.g., experienced teachers or other subject teachers). As revealed in the interview data in Study 2, and in previous literature, how teachers feel about themselves is directly related to the ways in which they are positioned by those around them (J. Chen, 2016). Therefore, improvements in teachers' emotional experiences and changes to their beliefs and practices can never be achieved by individual teachers, but, rather, should be carefully considered by different stakeholders within the level of support provided by the school climate. Given the multiple demands that have emerged in classrooms and standards by which teachers are judged, attention to teachers' experiences within schools is necessary. Specifically, it is important for school leaders to realise their responsibility for the psychological well-being of the teaching profession rather than merely for teaching effectiveness, especially in the context of the hierarchical Chinese culture where the voices at the educational frontline are more likely to be marginalised or even ignored during the decision-making process (Yin, 2013). It is hoped that the findings of this project could help raise school leaders' awareness of HTs' devotion to their work and recognise the role that leadership can play in supporting teachers' positive emotions and reducing their experience of negative emotions.

One step forward would be to develop ongoing partnerships between teacher colleagues in a way that moves beyond the pedagogical issue. As mentioned in the literature review chapter, in response to the experienced teachers' reluctance to take up an HT position, the age of current HTs was younger rather than older. However, it is reported that new teachers are more likely to experience emotional quandaries, and one crucial way for them to empower themselves is to seek support from their colleagues (Väisänen et al., 2017). Combined with the findings that contextual resources and support are key factors for teachers' positive emotions (Study 2), it is suggested that efforts should be made to initiate collaborations between teachers. New teachers can be trained in emotional regulation strategies and high-expectation principles, by experienced teachers, internally. In particular, an apprenticeship system could be set up and experience-sharing sessions could be arranged. When teachers share their beliefs, confusions, various aspects of teaching, and ways of responding to dilemmas, in a collaborative and dynamic way, they are more likely to develop coping solutions and relieve their difficulties (Ali et al., 2006).

6.5 Limitations and Future Research Directions

Some limitations of this research, such as the reliance on self-report data and the exploratory nature of the studies, have already been discussed in previous chapters and considered when interpreting the study findings. Thus, only the limitations of the whole research project will be addressed in this section, to avoid repetition. This section will also provide suggestions for future directions which are testimony to the implications of this research project.

One limitation to the overall design of this project is the lack of student voice and views of TE. Although students' SC outcomes were measured through a questionnaire, it would have been even better to have also approached the other side of the story, for example, how students perceived and interpreted their HT's TE. These views are worthy of investigation given that the way in which students conceive and perceive TE is a key part of TE effects. Therefore, a qualitative follow-up study (e.g., interview with students, class observations) could be conducted to enable additional understanding about why and how any significant relations emerged as they did.

Secondly, one thing that can be perceived as a limitation and a point of discussion is the lack of, and thus the need for, experimental studies that explore other types of TE. The studies in this project only focused on the academic TE, nevertheless, teachers may hold TE of other fields such as expectations for students' moral development. Therefore, future investigations could explore teachers' nonacademic expectations for their students and their possible relations with students' personal development. Whereas this project was conducted within the group of HTs, it is hoped that the findings can be transferred to subject teachers. Whether such TE effects exist among subject teachers and, if so, whether the patterns work in similar ways to those of their HT counterparts are worthy of further exploration.

Lastly, due to the time limit of a PhD project and the constraints of COVID-19, the current research was only able to track teacher and student data for 1 school year. A more dynamic trajectory of TE could be expected to emerge from the research if it covered more school years with more data points. Additionally, fewer time points measured in this research indicate that caution should be taken when interpreting the trends. Given the differences found between the students of high- and low-expectation teachers by the end of the school year, it would seem worthwhile to examine the accumulation of TE effects longitudinally. Additionally, longitudinal studies would also enable researchers to build trusted relationships with teachers and students, which would help participants feel more comfortable to share the crucial and underexplored experiences concerning their emotions and beliefs.

6.6 Concluding Comments

This doctoral project took 3 years, three studies, two methods of inquiry, and a total of 135 HTs and 348 students to generate an overall picture of TE effects in the Chinese high school context. The associations found between TE, teacher emotions, and student SC outcomes can be seen as a stepping-stone towards a more comprehensive understanding of TE effects.

Although this research attempted to portray how TE effects were exerted in Chinese high school contexts, and the possible explanations for the underlying mechanism, this work is still in its infancy. Therefore, this thesis has not only answered but has also raised profound questions regarding what could be done in future explorations to provide insights into the social-psychological field of education following the findings of this doctoral project.

References

- Agirdag, O., Van Avermaet, P., & Van Houtte, M. (2013). School segregation and math achievement: A mixed-method study on the role of self-fulfilling prophecies. *Teachers College Record*, 115(3), 1–50. https://doi.org/10.1177/016146811311500305
- Ali, M. M., Mustapha, R., & Jelas, Z. M. (2006). An empirical study on teachers' perceptions towards inclusive education in Malaysia. *International Journal of Special Education*, 21(3), 36–44.
- Alreck, P. L., & Settle, R. B. (1995). *The survey research handbook: Guidelines and strategies for conducting a survey*. IRWIN Professional Publishing.
- Ambady, N., & Rosenthal, R. (1992). Thin slices of expressive behavior as predictors of interpersonal consequences: A meta-analysis. *Psychological Bulletin*, 111(2), 256– 274. https://doi.org/10.1037/0033-2909.111.2.256
- Ansari, A., Pianta, R. C., Whittaker, J. V., Vitiello, V. E., & Ruzek, E. A. (2020). Preschool teachers' emotional exhaustion in relation to classroom instruction and teacher-child interactions. *Early Education and Development*, 33(1), 107–120. https://doi.org/10.1080/10409289.2020.1848301
- Archambault, I., Janosz, M., & Chouinard, R. (2012). Teacher beliefs as predictors of adolescents' cognitive engagement and achievement in mathematics. *The Journal of Educational Research*, 105(5), 319–328. https://doi.org/10.1080/00220671.2011.629694
- Arens, A. K., & Morin, A. J. S. (2016). Relations between teachers' emotional exhaustion and students' educational outcomes. *Journal of Educational Psychology*, *108*(6), 800– 813. https://doi.org/10.1037/edu0000105
- Aultman, L. P., Williams-Johnson, M. R., & Schutz, P. A. (2009). Boundary dilemmas in teacher–student relationships: Struggling with "the line." *Teaching and Teacher Education*, 25(5), 636–646. https://doi.org/10.1016/j.tate.2008.10.002
- Babad, E. (1998). Preferential affect: The crux of the teacher expectancy issue. Advances in Research on Teaching, 7, 183–214.
- Babad, E., Bernieri, F., & Rosenthal, R. (1989a). Nonverbal communication and leakage in the behavior of biased and unbiased teachers. *Journal of Personality and Social Psychology*, 56(1), 89–94. https://doi.org/10.1037/0022-3514.56.1.89

- Babad, E., Bernieri, F., & Rosenthal, R. (1989b). When less information is more informative:
 Diagnosing teacher expectations from brief samples of behaviour. *British Journal of Educational Psychology*, 59(3), 281–295. https://doi.org/10.1111/j.2044-8279.1989.tb03103.x
- Babad, E., Bernieri, F., & Rosenthal, R. (1991). Students as judges of teachers' verbal and nonverbal behavior. *American Educational Research Journal*, 28(1), 211–234. https://doi.org/10.3102/00028312028001211
- Babad, E., & Taylor, P. J. (1992). Transparency of teacher expectancies across language, cultural boundaries. *The Journal of Educational Research*, 86(2), 120–125. https://doi.org/10.1080/00220671.1992.9941148
- Babad, E. Y., Inbar, J., & Rosenthal, R. (1982). Pygmalion, Galatea, and the Golem:
 Investigations of biased and unbiased teachers. *Journal of Educational Psychology*, 74(4), 459–474. https://doi.org/10.1037/0022-0663.74.4.459
- Baeriswyl, S., Bratoljic, C., & Krause, A. (2021). How homeroom teachers cope with high demands: Effect of prolonging working hours on emotional exhaustion. *Journal of School Psychology*, 85, 125–139. https://doi.org/10.1016/j.jsp.2021.02.002
- Bandura, A. (1977). Self-efficacy: Toward a unifying theory of behavioral change. *Psychological Review*, 84(2), 191–215. https://doi.org/10.1037/0033-295X.84.2.191
- Bandura, A. (1986). *Social foundations of thought and action: A social cognitive theory*. Prentice-Hall.
- Barcelos, A. M. F., & Ruohotie-Lyhty, M. (2018). Teachers' emotions and beliefs in second language teaching: Implications for teacher education. In J. de D. Martínez Agudo (Ed.), *Emotions in second language teaching* (pp. 109–124). Springer International Publishing. https://doi.org/10.1007/978-3-319-75438-3_7
- Baruch, Y., & Holtom, B. C. (2008). Survey response rate levels and trends in organizational research. *Human Relations*, 61(8), 1139–1160. https://doi.org/10.1177/0018726708094863
- Birbili, M. (2000). Translating from one language to another. *Social Research Update*, *31*(1), 1–7.
- Braun, S. S., Roeser, R. W., Mashburn, A. J., & Skinner, E. (2019). Middle school teachers' mindfulness, occupational health and well-being, and the quality of teacher–student interactions. *Mindfulness*, 10(2), 245–255. https://doi.org/10.1007/s12671-018-0968-2
- Braun, V., & Clarke, V. (2006). Using thematic analysis in psychology. *Qualitative Research in Psychology*, *3*(2), 77–101. https://doi.org/10.1191/1478088706qp063oa

- Brislin, R. W. (1970). Back-translation for cross-cultural research. *Journal of Cross-Cultural Psychology*, *1*(3), 185–216. https://doi.org/10.1177/135910457000100301
- Brophy, J. (1985). Teacher–student interaction. In J. B. Dusek, V. Hall, & W. Meyer (Eds.), *Teacher expectancies* (pp. 303–328). Erlbaum.
- Brophy, J. E. (1983). Research on the self-fulfilling prophecy and teacher expectations. *Journal of Educational Psychology*, 75, 631–661. https://doi.org/10.1037/0022-0663.75.5.631
- Brophy, J. E., & Good, T. L. (1970). Teachers' communication of differential expectations for children's classroom performance: Some behavioral data. *Journal of Educational Psychology*, 61(5), 365–374. https://doi.org/10.1037/h0029908
- Brown, T. A. (2015). *Confirmatory factor analysis for applied research* (2nd ed.). Guilford Publications.
- Bullough, R. V. (2005). Teacher vulnerability and teachability: A case study of a mentor and two interns. *Teacher Education Quarterly*, 32(2), 23–39.
- Burić, I., & Macuka, I. (2018). Self-efficacy, emotions and work engagement among teachers: A two wave cross-lagged analysis. *Journal of Happiness Studies*, 19(7), 1917–1933. https://doi.org/10.1007/s10902-017-9903-9
- Burić, I., Penezić, Z., & Sorić, I. (2017). Regulating emotions in the teacher's workplace:
 Development and initial validation of the Teacher Emotion-Regulation Scale.
 International Journal of Stress Management, 24(3), 217–246.
 https://doi.org/10.1037/str0000035
- Burić, I., Slišković, A., & Macuka, I. (2018). A mixed-method approach to the assessment of teachers' emotions: Development and validation of the Teacher Emotion
 Questionnaire. *Educational Psychology*, *38*(3), 325–349.
 https://doi.org/10.1080/01443410.2017.1382682
- Burić, I., Slišković, A., & Sorić, I. (2020). Teachers' emotions and self-efficacy: A test of reciprocal relations. *Frontiers in Psychology*, 11, 1650. https://doi.org/10.3389/fpsyg.2020.01650
- Byrne, B. M. (1988). Measuring adolescent self-concept: Factorial validity and equivalency of the SDQ III across gender. *Multivariate Behavioral Research*, 23(3), 361–375. https://doi.org/10.1207/s15327906mbr2303_5
- Carter, D. L. (1971). The effect of teacher expectations on the self-esteem and academic performance of seventh grade students. *Dissertation Abstracts International Section A: Humanities and Social Sciences*, 31(9-A), 4539.

- Castro, A. J., Kelly, J., & Shih, M. (2010). Resilience strategies for new teachers in highneeds areas. *Teaching and Teacher Education*, 26(3), 622–629. https://doi.org/10.1016/j.tate.2009.09.010
- Chang, M.-L. (2013). Toward a theoretical model to understand teacher emotions and teacher burnout in the context of student misbehavior: Appraisal, regulation and coping. *Motivation and Emotion*, 37(4), 799–817. https://doi.org/10.1007/s11031-012-9335-0
- Chang, M.-L. (2020). Emotion display rules, emotion regulation, and teacher burnout. *Frontiers in Education*, *5*, 90. https://doi.org/10.3389/feduc.2020.00090
- Chen, F. F., Sousa, K. H., & West, S. G. (2005). Teacher's corner: Testing measurement invariance of second-order factor models. *Structural Equation Modeling: A Multidisciplinary Journal*, 12(3), 471–492. https://doi.org/10.1207/s15328007sem1203_7
- Chen, H.-Y. (2004). Developing a model of spinal cord injury rehabilitation nursing using grounded theory [Doctoral dissertation, University of Ulster]. https://ethos.bl.uk/OrderDetails.do?uin=uk.bl.ethos.413285
- Chen, H.-Y., & Boore, J. R. (2010). Translation and back-translation in qualitative nursing research: Methodological review. *Journal of Clinical Nursing*, 19(1–2), 234–239. https://doi.org/10.1111/j.1365-2702.2009.02896.x
- Chen, J. (2015). Teachers' conceptions of approaches to teaching: A Chinese perspective. *The Asia-Pacific Education Researcher*, 24(2), 341–351. https://doi.org/10.1007/s40299-014-0184-3
- Chen, J. (2016). Understanding teacher emotions: The development of a teacher emotion inventory. *Teaching and Teacher Education*, 55, 68–77. https://doi.org/10.1016/j.tate.2016.01.001
- Chen, J. (2019). Exploring the impact of teacher emotions on their approaches to teaching: A structural equation modelling approach. *British Journal of Educational Psychology*, 89(1), 57–74. https://doi.org/10.1111/bjep.12220
- Chen, J. (2020). Teacher emotions in their professional lives: Implications for teacher development. Asia-Pacific Journal of Teacher Education, 48(5), 491–507. https://doi.org/10.1080/1359866X.2019.1669139
- Chen, J. (2021). Refining the teacher emotion model: Evidence from a review of literature published between 1985 and 2019. *Cambridge Journal of Education*, 51(3), 327–357. https://doi.org/10.1080/0305764X.2020.1831440

- Chen, S.-K., Yeh, Y.-C., Hwang, F.-M., & Lin, S. S. J. (2013). The relationship between academic self-concept and achievement: A multicohort–multioccasion study. *Learning and Individual Differences*, 23, 172–178. https://doi.org/10.1016/j.lindif.2012.07.021
- Chen, Y.-H., Thompson, M. S., Kromrey, J. D., & Chang, G. H. (2011). Relations of student perceptions of teacher oral feedback with teacher expectancies and student selfconcept. *The Journal of Experimental Education*, 79(4), 452–477. https://doi.org/10.1080/00220973.2010.547888
- Cohen, J. (1960). A coefficient of agreement for nominal scales. *Educational and Psychological Measurement*, 20(1), 37–46. https://doi.org/10.1177/001316446002000104
- Cohen, J. (1968). Weighted kappa: Nominal scale agreement provision for scaled disagreement or partial credit. *Psychological Bulletin*, 70(4), 213–220. https://doi.org/10.1037/h0026256
- Cohen, J. (1988). *Statistical power analysis for the behavioral sciences* (2nd ed.). L. Erlbaum Associates. http://dx.doi.org/10.4324/9780203771587
- Comrey, A. L., & Lee, H. B. (2013). *A first course in factor analysis*. Psychology Press. http://dx.doi.org/10.4324/9781315827506
- Cooper, H. M., & Tom, D. Y. (1984). Teacher expectation research: A review with implications for classroom instruction. *The Elementary School Journal*, 85(1), 77–89.
- Cooper, K. M., Krieg, A., & Brownell, S. E. (2018). Who perceives they are smarter? Exploring the influence of student characteristics on student academic self-concept in physiology. *Advances in Physiology Education*, 42(2), 200–208. https://doi.org/10.1152/advan.00085.2017
- Corcoran, R. P., & Tormey, R. (2012). Developing emotionally competent teachers: Emotional intelligence and pre-service teacher education. *Peter Lang*.
- Cowan, P. A., Cowan, C. P., Heming, G., & Miller, N. (1995). *The child adaptive behavior inventory*. Institute of Human Development, University of California at Berkeley.
- Creswell, J. W., & Miller, D. L. (2000). Determining validity in qualitative inquiry. *Theory Into Practice*, *39*(3), 124–130. https://doi.org/10.1207/s15430421tip3903_2
- Cross, D. I., & Hong, J. Y. (2009). Beliefs and professional identity: Critical constructs in examining the impact of reform on the emotional experiences of teachers. In P. A. Schutz & M. Zembylas (Eds.), *Advances in teacher emotion research* (pp. 273–296). Springer. https://doi.org/10.1007/978-1-4419-0564-2_14

- Dai, D. Y. (2002). Incorporatingparent perceptions: A replication and extension study of the internal-external frame of reference model of self-concept development. *Journal of Adolescent Research*, 17(6), 617–645. https://doi.org/10.1177/074355802237467
- Day, C., & Gu, Q. (2013). Resilient teachers, resilient schools: Building and sustaining quality in testing times. Routledge.
- de Boer, H., Bosker, R. J., & van der Werf, M. P. (2010). Sustainability of teacher expectation bias effects on long-term student performance. *Journal of Educational Psychology*, 102(1), 168–179. https://doi.org/10.1037/a0017289
- de Ruiter, J. A., Poorthuis, A. M., Aldrup, K., & Koomen, H. M. (2020). Teachers' emotional experiences in response to daily events with individual students varying in perceived past disruptive behavior. *Journal of School Psychology*, 82, 85–102. https://doi.org/10.1016/j.jsp.2020.08.005
- de Ruiter, J. A., Poorthuis, A. M., & Koomen, H. M. (2019). Relevant classroom events for teachers: A study of student characteristics, student behaviors, and associated teacher emotions. *Teaching and Teacher Education*, 86, 102899. https://doi.org/10.1016/j.tate.2019.102899
- de Ruiter, J. A., Poorthuis, A. M., & Koomen, H. M. (2021). Teachers' emotional labor in response to daily events with individual students: The role of teacher–student relationship quality. *Teaching and Teacher Education*, 107, 103467. https://doi.org/10.1016/j.tate.2021.103467
- Demanet, J., & Van Houtte, M. (2012). Teachers' attitudes and students' opposition. School misconduct as a reaction to teachers' diminished effort and affect. *Teaching and Teacher Education*, 28(6), 860–869. https://doi.org/10.1016/j.tate.2012.03.008
- Denner, J., Valdes, O., Dickson, D. J., & Laursen, B. (2019). Math interest and self-concept among latino/a students: Reciprocal influences across the transition to middle school. *Journal of Adolescence*, 75, 22–36. https://doi.org/10.1016/j.adolescence.2019.06.015
- Diamond, J. B., Randolph, A., & Spillane, J. P. (2004). Teachers' Expectations and Sense of Responsibility for Student Learning: The Importance of Race, Class, and Organizational Habitus. *Anthropology & Education Quarterly*, 35(1), 75–98. https://doi.org/10.1525/aeq.2004.35.1.75
- Ding, H., & Rubie-Davies, C. M. (2019). Teacher expectation intervention: Is it effective for all students? *Learning and Individual Differences*, 74, 101751. https://doi.org/10.1016/j.lindif.2019.06.005

- Donker, M. H., Erisman, M. C., van Gog, T., & Mainhard, T. (2020). Teachers' emotional exhaustion: Associations with their typical use of and implicit attitudes toward emotion regulation strategies. *Frontiers in Psychology*, 11, 867. https://doi.org/10.3389/fpsyg.2020.00867
- Dryman, M. T., & Heimberg, R. G. (2018). Emotion regulation in social anxiety and depression: A systematic review of expressive suppression and cognitive reappraisal. *Clinical Psychology Review*, 65, 17–42. https://doi.org/10.1016/j.cpr.2018.07.004

Dusek, J. B., Hall, V., & Meyer, W. (1985). Teacher expectancies. Lawrence Erlbaum.

- Efklides, A. (2006). Metacognition and affect: What can metacognitive experiences tell us about the learning process? *Educational Research Review*, 1(1), 3–14. https://doi.org/10.1016/j.edurev.2005.11.001
- Esnaola, I., Sesé, A., Antonio-Agirre, I., & Azpiazu, L. (2020). The development of multiple self-concept dimensions during adolescence. *Journal of Research on Adolescence*, 30(S1), 100–114. https://doi.org/10.1111/jora.12451
- Evans, J. T., & Rosenthal, R. (1969). Interpersonal self-fulfilling prophecies: Further extrapolations from the laboratory to the classroom. *Proceedings of the Annual Convention of the American Psychological Association*, *4*, 371–372. https://eric.ed.gov/?id=ED034276
- Faria, L. (1996). Marsh's self-description questionnaire III (SDQ III): Adaptation study with portuguese college students. *Social Behavior and Personality: An International Journal*, 24(4), 343–349. https://doi.org/10.2224/sbp.1996.24.4.343
- Filep, B. (2009). Interview and translation strategies: Coping with multilingual settings and data. *Social Geography*, 4(1), 59–70. https://doi.org/10.5194/sg-4-59-2009
- Fives, H., & Buehl, M. M. (2012). Spring cleaning for the "messy" construct of teachers' beliefs: What are they? Which have been examined? What can they tell us? In K. R. Harris, S. Graham, T. Urdan, S. Graham, J. M. Royer, & M. Zeidner (Eds.), *APA educational psychology handbook, Vol 2: Individual differences and cultural and contextual factors*. (pp. 471–499). American Psychological Association. https://doi.org/10.1037/13274-019
- Flaherty, A. A. (2020). A review of affective chemistry education research and its implications for future research. *Chemistry Education Research and Practice*, 21(3), 698–713. https://doi.org/10.1039/C9RP00200F
- Fong, V. L. (2004). Only hope: Coming of age under China's one-child policy. Stanford University Press.

- Frenzel, A. C. (2014). Teacher emotions. In E. A. Linnenbrink-Garcia & R. Pekrun (Eds.), *International handbook of emotions in education* (pp. 494–519). Routledge/Taylor & Francis Group. http://dx.doi.org/10.4324/9780203148211.ch25
- Frenzel, A. C., Daniels, L., & Burić, I. (2021). Teacher emotions in the classroom and their implications for students. *Educational Psychologist*, 56(4), 250–264. https://doi.org/10.1080/00461520.2021.1985501
- Frenzel, A. C., Fiedler, D., Marx, A. K. G., Reck, C., & Pekrun, R. (2020). Who enjoys teaching, and when? Between- and within-person evidence on teachers' appraisalemotion links. *Frontiers in Psychology*, 11. https://doi.org/10.3389/fpsyg.2020.01092
- Frenzel, A. C., Pekrun, R., Goetz, T., Daniels, L. M., Durksen, T. L., Becker-Kurz, B., & Klassen, R. M. (2016). Measuring teachers' enjoyment, anger, and anxiety: The Teacher Emotions Scales (TES). *Contemporary Educational Psychology*, 46, 148– 163. https://doi.org/10.1016/j.cedpsych.2016.05.003
- Friedrich, A., Flunger, B., Nagengast, B., Jonkmann, K., & Trautwein, U. (2015). Pygmalion effects in the classroom: Teacher expectancy effects on students' math achievement. *Contemporary Educational Psychology*, 41, 1–12. https://doi.org/10.1016/j.cedpsych.2014.10.006
- Frijda, N. H., Manstead, A. S., & Bem, S. (Eds.). (2000). Emotions and belief: How feelings influence thoughts. Cambridge University Press. http://dx.doi.org/10.1017/CBO9780511659904
- Frijda, N. H., & Mesquita, B. (2000). Beliefs through emotions. In N. H. Frijda, A. S.
 Manstead, & S. Bem (Eds.), *Emotions and belief: How feelings influence thoughts* (pp. 45–77). Cambridge University Press. https://doi.org/10.1017/CBO9780511659904.003
- Furinghetti, F., & Pehkonen, E. (2002). Rethinking characterizations of beliefs. In G. C. Leder, E. Pehkonen, & G. Törner (Eds.), *Beliefs: A hidden variable in mathematics education?* (pp. 39–57). Springer Netherlands. https://doi.org/10.1007/0-306-47958-3_3
- Gallagher, M. W., & Brown, T. A. (2013). Introduction to Confirmatory Factor Analysis and Structural Equation Modeling. In T. Teo (Ed.), *Handbook of quantitative methods for educational research* (pp. 287–314). Brill. http://dx.doi.org/10.1007/978-94-6209-404-8_14

- Gao, L. (1998). Cultural context of school science teaching and learning in the People's Republic of China. *Science Education*, 82(1), 1–13. https://doi.org/10.1002/(SICI)1098-237X(199801)82:1<1::AID-SCE1>3.0.CO;2-L
- Gao, X. (2008). Teachers' professional vulnerability and cultural tradition: A Chinese paradox. *Teaching and Teacher Education*, 24(1), 154–165. https://doi.org/10.1016/j.tate.2006.11.011
- Gao, X. (2011). The 'falling' language standards and teachers' professional vulnerability in Hong Kong. *Research Papers in Education*, 26(4), 485–498. https://doi.org/10.1080/02671521003637146
- Gentrup, S., Lorenz, G., Kristen, C., & Kogan, I. (2020). Self-fulfilling prophecies in the classroom: Teacher expectations, teacher feedback and student achievement. *Learning* and Instruction, 66, 101296. https://doi.org/10.1016/j.learninstruc.2019.101296
- Good, T. L., & Lavigne, A. L. (2017). *Looking in classrooms* (11th ed.). Routledge. https://doi.org/10.4324/9781315627519
- Gray, D. (2014). Doing research in the real world (3rd ed.). Sage.
- Gross, J. J. (1998). The emerging field of emotion regulation: An integrative review. *Review* of General Psychology, 2(3), 271–299. https://doi.org/10.1037/1089-2680.2.3.271
- Gross, J. J., & John, O. P. (2003). Individual differences in two emotion regulation processes: Implications for affect, relationships, and well-being. *Journal of Personality and Social Psychology*, 85(2), 348–362. https://doi.org/10.1037/0022-3514.85.2.348
- Gujare, S. K., & Tiwari, G. K. (2016). Academic self-concept and academic outcome of the graduate students: The mediating role of socioeconomic status and gender.
 International Journal of Education and Psychological Research, 5(4), 1–7.
- Hair, J. F., Black, W. C., Babin, B. J., & Anderson, R. E. (2013). Multivariate data analysis: Pearson new international edition. Pearson Education. http://ebookcentral.proquest.com/lib/auckland/detail.action?docID=5175365
- Han, J., Yin, H., & Wang, J. (2020). Examining the relationships Between Job Characteristics, Emotional Regulation and University Teachers' Well-Being: The Mediation of Emotional Regulation. *Frontiers in Psychology*, 11. https://doi.org/10.3389/fpsyg.2020.01727
- Han, X. (2012). Big moves to improve the quality of teacher education in China. On the Horizon, 20(4), 324–335. https://doi.org/10.1108/10748121211272461
- Hargreaves, A. (1998). The emotional practice of teaching. *Teaching and Teacher Education*, 14(8), 835–854. https://doi.org/10.1016/S0742-051X(98)00025-0

- Hargreaves, A. (2000). Mixed emotions: Teachers' perceptions of their interactions with students. *Teaching and Teacher Education*, *16*(8), 811–826.
- Hargreaves, A. (2001). Emotional geographies of teaching. *Teachers College Record*, 103(6), 1056–1080.
- Harris, M. J., & Rosenthal, R. (1985). Mediation of interpersonal expectancy effects: 31 meta-analyses. *Psychological Bulletin*, 97, 363–386. https://doi.org/10.1037/0033-2909.97.3.363
- Hayes, A. F., & Coutts, J. J. (2020). Use omega rather than Cronbach's alpha for estimating reliability. But.... Communication Methods and Measures, 14(1), 1–24. https://doi.org/10.1080/19312458.2020.1718629
- Hermsen, L. A., Leone, S. S., Smalbrugge, M., Knol, D. L., van der Horst, H. E., & Dekker, J. (2013). Exploring the aggregation of four functional measures in a population of older adults with joint pain and comorbidity. *BMC Geriatrics*, 13(1), 119. https://doi.org/10.1186/1471-2318-13-119
- Hinnant, J. B., O'Brien, M., & Ghazarian, S. R. (2009). The longitudinal relations of teacher expectations to achievement in the early school years. *Journal of Educational Psychology*, 101(3), 662–670. https://doi.org/10.1037/a0014306
- Holstein, J. A., & Gubrium, J. F. (2000). *The self we live by: Narrative identity in a postmodern world*. Oxford University Press.
- Hong, J. Y. (2012). Why do some beginning teachers leave the school, and others stay? Understanding teacher resilience through psychological lenses. *Teachers and Teaching*, 18(4), 417–440. https://doi.org/10.1080/13540602.2012.696044
- Honig, H. G. (1997). Positions, power and practice: Functionalist approaches and translation quality assessment. *Current Issues in Language and Society*, 4(1), 6–34. https://doi.org/10.1080/13520529709615477
- Hooper, D., Coughlan, J., & Mullen, M. (2008). Structural equation modelling: Guidelines for determining model fit. *Electronic Journal of Business Research Methods*, 6(1), 53–60.
- Howard, S. K., & Gigliotti, A. (2016). Having a go: Looking at teachers' experience of risktaking in technology integration. *Education and Information Technologies*, 21(5), 1351–1366. https://doi.org/10.1007/s10639-015-9386-4
- Hu, L., & Bentler, P. M. (1999). Cutoff criteria for fit indexes in covariance structure analysis: Conventional criteria versus new alternatives. *Structural Equation*

Modeling: A Multidisciplinary Journal, *6*(1), 1–55. https://doi.org/10.1080/10705519909540118

- Hu, T., Zhang, D., Wang, J., Mistry, R., Ran, G., & Wang, X. (2014). Relation between emotion regulation and mental health: A meta-analysis review. *Psychological Reports*, 114(2), 341–362. https://doi.org/10.2466/03.20.PR0.114k22w4
- Huang, G. H.-C., & Gove, M. (2015). Confucianism, Chinese families, and academic achievement: Exploring how Confucianism and Asian descendant parenting practices influence children's academic achievement. In M. S. Khine (Ed.), *Science education in East Asia: Pedagogical innovations and research-informed practices* (pp. 41–66). Springer International Publishing. https://doi.org/10.1007/978-3-319-16390-1_3
- Janghorban, R., Roudsari, R. L., & Taghipour, A. (2014). Skype interviewing: The new generation of online synchronous interview in qualitative research. *International Journal of Qualitative Studies on Health and Well-Being*, 9(1), 24152. https://doi.org/10.3402/qhw.v9.24152
- Jennings, P. A., & Greenberg, M. T. (2009). The prosocial classroom: Teacher social and emotional competence in relation to student and classroom outcomes. *Review of Educational Research*, 79(1), 491–525. https://doi.org/10.3102/0034654308325693
- Jia, Z. (2012). Jiaoshi qiwang dui zhongxue yingyu ketang shisheng hudong de yingxiang [The influence of teacher's expectation on teacher-student interaction in junior English classroom] [Master's thesis, Shanxi Normal University.]. https://cdmd.cnki.com.cn/Article/CDMD-10118-1012340677.htm
- Jiang, Z., Mok, I. A., & Yin, H. (2021). The relationships between teacher emotions and classroom instruction: Evidence from senior secondary mathematics teachers in China. *International Journal of Educational Research*, 108, 101792. https://doi.org/10.1016/j.ijer.2021.101792
- Jussim, L. (1989). Teacher expectations: Self-fulfilling prophecies, perceptual biases, and accuracy. *Journal of Personality and Social Psychology*, 57(3), 469–480. https://doi.org/10.1037/0022-3514.57.3.469
- Jussim, L. (2012). Social perception and social reality: Why accuracy dominates bias and self-fulfilling prophecy. Oxford University Press. http://dx.doi.org/10.1093/acprof:oso/9780195366600.001.0001
- Jussim, L., Eccles, J., & Madon, S. (1996). Social perception, social stereotypes, and teacher expectations: Accuracy and the quest for the powerful self-fulfilling prophecy. In M.
P. Zanna (Ed.), *Advances in experimental social psychology* (Vol. 28, pp. 281–388). Academic Press. https://doi.org/10.1016/S0065-2601(08)60240-3

- Jussim, L., & Eccles, J. S. (1992). Teacher expectations: II. Construction and reflection of student achievement. *Eacher Expectations: II. Construction and Reflection of Student Achievement*, 63(5), 947–961. https://doi.org/10.1037/0022-3514.63.6.947
- Jussim, L., & Harber, K. D. (2005). Teacher expectations and self-fulfilling prophecies: Knowns and unknowns, resolved and unresolved controversies. *Personality and Social Psychology Review*, 9(2), 131–155. https://doi.org/10.1207/s15327957pspr0902_3
- Karwowski, M., Gralewski, J., & Szumski, G. (2015). Teachers' effect on students' creative self-beliefs is moderated by students' gender. *Learning and Individual Differences*, 44, 1–8. https://doi.org/10.1016/j.lindif.2015.10.001
- Kelchtermans, G. (1996). Teacher vulnerability: Understanding its moral and political roots. *Cambridge Journal of Education*, 26(3), 307. https://doi.org/10.1080/0305764960260302
- Kelchtermans, G. (2005). Teachers' emotions in educational reforms: Self-understanding, vulnerable commitment and micropolitical literacy. *Teaching and Teacher Education*, 21(8), 995–1006. https://doi.org/10.1016/j.tate.2005.06.009
- Kim, H. S. (2015). Foregone opportunities: Unveiling teacher expectancy effects in kindergarten using counterfactual predictions. *Social Psychology of Education*, 18(2), 273–296. https://doi.org/10.1007/s11218-014-9284-4
- Kim, H.-Y. (2013). Statistical notes for clinical researchers: Assessing normal distribution (2) using skewness and kurtosis. *Restorative Dentistry & Endodontics*, 38(1), 52–54. https://doi.org/10.5395/rde.2013.38.1.52
- Kim, S. W. (2020). Meta-analysis of parental involvement and achievement in East Asian countries. *Education and Urban Society*, 52(2), 312–337. https://doi.org/10.1177/0013124519842654
- Kitayama, S., Park, J., Boylan, J. M., Miyamoto, Y., Levine, C. S., Markus, H. R., Karasawa, M., Coe, C. L., Kawakami, N., Love, G. D., & Ryff, C. D. (2015). Expression of anger and ill health in two cultures: An examination of inflammation and cardiovascular risk. *Psychological Science*, *26*(2), 211–220. https://doi.org/10.1177/0956797614561268
- Klassen, R., Wilson, E., Siu, A. F. Y., Hannok, W., Wong, M. W., Wongsri, N., Sonthisap,P., Pibulchol, C., Buranachaitavee, Y., & Jansem, A. (2013). Preservice teachers'

work stress, self-efficacy, and occupational commitment in four countries. *European Journal of Psychology of Education*, 28(4), 1289–1309. https://doi.org/10.1007/s10212-012-0166-x

- Kline, R. B. (2015). *Principles and practice of structural equation modeling* (4th ed.). Guilford Publications.
- Kline, T. (2005). *Psychological testing: A practical approach to design and evaluation*. SAGE Publications. https://doi.org/10.4135/9781483385693
- Koufteros, X., Babbar, S., & Kaighobadi, M. (2009). A paradigm for examining second-order factor models employing structural equation modeling. *International Journal of Production Economics*, 120(2), 633–652. https://doi.org/10.1016/j.ijpe.2009.04.010
- Kyriazos, T. A. (2018). Applied psychometrics: Sample size and sample power considerations in factor analysis (EFA, CFA) and SEM in general. *Psychology*, 9(08), 2207. https://doi.org/10.4236/psych.2018.98126
- Landis, J. R., & Koch, G. G. (1977). The measurement of observer agreement for categorical data. *Biometrics*, *33*(1), 159–174. https://doi.org/10.2307/2529310
- Lazarus, R. S. (1991). Progress on a cognitive-motivational-relational theory of emotion. *American Psychologist*, *46*(8), 819–834. https://doi.org/10.1037/0003-066X.46.8.819
- Lazarus, R. S. (2001). Relational meaning and discrete emotions. In K. R. Scherer, A. Schorr, & T. Johnstone (Eds.), *Appraisal processes in emotion: Theory, methods, research* (pp. 37–67). Oxford University Press.
- Leavitt, J. (1996). Meaning and feeling in the anthropology of emotions. *American Ethnologist*, 23(3), 514–539.
- Lee, J. C.-K., Huang, Y. X.-H., Law, E. H.-F., & Wang, M.-H. (2013). Professional identities and emotions of teachers in the context of curriculum reform: A Chinese perspective. *Asia-Pacific Journal of Teacher Education*, 41(3), 271–287. https://doi.org/10.1080/1359866X.2013.809052
- Lee, J. C.-K., & Yin, H.-B. (2011). Teachers' emotions and professional identity in curriculum reform: A Chinese perspective. *Journal of Educational Change*, 12(1), 25–46. https://doi.org/10.1007/s10833-010-9149-3
- Levin, B. B. (2014). The development of teachers' beliefs. In H. Fives & M. G. Gill (Eds.), International handbook of research on teachers' beliefs (pp. 60–77). Routledge Handbooks Online. http://dx.doi.org/10.4324/9780203108437

- Li, H. (2017). The 'secrets' of Chinese students' academic success: Academic resilience among students from highly competitive academic environments. *Educational Psychology*, *37*(8), 1001–1014. https://doi.org/10.1080/01443410.2017.1322179
- Li, H., & Du, X.-Y. (2013). Confronting cultural challenges when reconstructing the teacher– student relationship in a Chinese context. In M. J. Kirkebæk, X.-Y. Du, & A. A. Jensen (Eds.), *Teaching and learning culture* (pp. 79–94). SensePublishers. https://doi.org/10.1007/978-94-6209-440-6_6
- Li, M., Zheng, C., Liang, J.-C., Zhang, Y., & Tsai, C.-C. (2018). Conceptions, selfregulation, and strategies of learning science among Chinese high school students. *International Journal of Science and Mathematics Education*, 16(1), 69–87. https://doi.org/10.1007/s10763-016-9766-2
- Li, Z. (2014). Teachers matter: Expectation effects in foreign language classrooms at university [Doctoral dissertation, University of Auckland]. https://researchspace.auckland.ac.nz/handle/2292/23050
- Li, Z., & Rubie-Davies, C. M. (2017). Teachers matter: Expectancy effects in Chinese university English-as-a-foreign-language classrooms. *Studies in Higher Education*, 42(11), 2042–2060. https://doi.org/10.1080/03075079.2015.1130692
- Li, Z., & Rubie-Davies, C. M. (2018). Teacher expectations in a university setting: The perspectives of teachers. *Educational Research and Evaluation*, 24(3–5), 201–220. https://doi.org/10.1080/13803611.2018.1550835
- Lim, S. C., & Thien, L. M. (2020). Chinese academic leadership from the perspective of Confucian virtues and its effects on teacher commitment. *International Online Journal of Educational Leadership*, 4(1), 37–51.
- Liu, J.-Q. (1997). The emotional bond between teachers and students: Multi-year relationships. *Phi Delta Kappan*, 79(2), 156–157.
- Liu, P., Liu, L., & Xie, Y. (2018). Leadership of class teachers (banzhuren) in improving student learning: Case studies of Chinese primary schools. *Frontiers of Education in China*, 13(1), 28–55. https://doi.org/10.1007/s11516-018-0002-4
- Liu, S., & Hallinger, P. (2018). Teacher development in rural China: How ineffective school leadership fails to make a difference. *International Journal of Leadership in Education*, 21(6), 633–650. https://doi.org/10.1080/13603124.2017.1294266
- Liu, S., & Onwuegbuzie, A. J. (2012). Chinese teachers' work stress and their turnover intention. *International Journal of Educational Research*, 53, 160–170. https://doi.org/10.1016/j.ijer.2012.03.006

- Lo, R. (2001). The role of class teachers in a key secondary school in Shanghai. *Pastoral Care in Education*, 19(1), 20–27. https://doi.org/10.1111/1468-0122.00185
- Madon, S., Jussim, L., & Eccles, J. (1997). In search of the powerful self-fulfilling prophecy. Journal of Personality and Social Psychology, 72(4), 791–809. https://doi.org/10.1037/0022-3514.72.4.791
- Maneesriwongul, W., & Dixon, J. K. (2004). Instrument translation process: A methods review. *Journal of Advanced Nursing*, 48(2), 175–186. https://doi.org/10.1111/j.1365-2648.2004.03185.x
- Mann, M., Hosman, C. M. H., Schaalma, H. P., & de Vries, N. K. (2004). Self-esteem in a broad-spectrum approach for mental health promotion. *Health Education Research*, 19(4), 357–372. https://doi.org/10.1093/her/cyg041
- Marsh, H. W., & Craven, R. (1997). Academic self-concept: Beyond the dustbowl. In G. D. Phye (Ed.), *Handbook of classroom assessment* (pp. 131–198). Elsevier. https://doi.org/10.1016/B978-012554155-8/50008-9
- Marsh, H. W., & Craven, R. G. (2006). Reciprocal effects of self-concept and performance from a multidimensional perspective: Beyond seductive pleasure and unidimensional perspectives. *Perspectives on Psychological Science*, 1(2), 133–163.
- Marsh, H. W., & O'Neill, R. (1984). Self description questionnaire III: The construct validity of multidimensional self-concept ratings by late adolescents. *Journal of Educational Measurement*, 21(2), 153–174. https://doi.org/10.1111/j.1745-3984.1984.tb00227.x
- Marsh, H. W., Richards, G. E., & Barnes, J. (1986). Multidimensional self-concepts: The effect of participation in an Outward Bound Program. *Journal of Personality and Social Psychology*, 50(1), 195–204. https://doi.org/10.1037/0022-3514.50.1.195
- Marsh, H. W., & Shavelson, R. (1985). Self-concept: Its multifaceted, hierarchical structure. *Educational Psychologist*, 20(3), 107–123. https://doi.org/10.1207/s15326985ep2003 1
- Marsh, H. W., Trautwein, U., Lüdtke, O., Köller, O., & Baumert, J. (2006). Integration of multidimensional self-concept and core personality constructs: Construct validation and relations to well-being and achievement. *Journal of Personality*, 74(2), 403–456. https://doi.org/10.1111/j.1467-6494.2005.00380.x
- Mazar, N., Amir, O., & Ariely, D. (2008). The dishonesty of honest people: A theory of selfconcept maintenance. *Journal of Marketing Research*, 45(6), 633–644. https://doi.org/10.1509/jmkr.45.6.633

- McCabe, D. L., Butterfield, K. D., & Treviño, L. K. (2012). *Cheating in college: Why students do it and what educators can do about it.* JHU Press.
- McDermott, M. A. N., & Palchanes, K. (1994). A literature review of the critical elements in translation theory. *Image: The Journal of Nursing Scholarship*, 26(2), 113–118. https://doi.org/10.1111/j.1547-5069.1994.tb00928.x
- McDonald, R. P. (1978). Generalizability in factorable domains: "Domain validity and generalizability." *Educational and Psychological Measurement*, 38(1), 75–79. https://doi.org/10.1177/001316447803800111
- McKown, C., & Weinstein, R. S. (2002). Modeling the role of child ethnicity and gender in children's differential response to teacher expectations. *Journal of Applied Social Psychology*, 32(1), 159–184. https://doi.org/10.1111/j.1559-1816.2002.tb01425.x
- McNaughton, S., & Li, J. (2021). Well-being in schools: Chinese and New Zealand approaches. *Educational Research for Policy and Practice*, 1–17. https://doi.org/10.1007/s10671-021-09296-8
- Meade, A. W., Johnson, E. C., & Braddy, P. W. (2008). Power and sensitivity of alternative fit indices in tests of measurement invariance. *Journal of Applied Psychology*, 93(3), 568–592. https://doi.org/10.1037/0021-9010.93.3.568
- Merriam, S. B. (1998). Qualitative research and case study applications in education. Revised and expanded from "Case study research in education." Jossey-Bass Publishers.
- Merton, R. K. (1948). The self-fulfilling prophecy. *The Antioch Review*, 8(2), 193–210. https://doi.org/10.2307/4609267
- Merton, R. K. (1957). Social theory and social structure. Free Press.
- Mevarech, Z. R., & Maskit, D. (2015). The teaching experience and the emotions it evokes. Social Psychology of Education, 18(2), 241–253. https://doi.org/10.1007/s11218-014-9286-2
- Mutch, C. (2013). *Doing educational research: A practitioner's guide to getting started* (2nd ed.). NZCER Press.
- Mynott, G. J. (2018). The academic self-concept of business and management students: A review of the literature. *The International Journal of Management Education*, 16(3), 515–523. https://doi.org/10.1016/j.ijme.2018.10.003
- Nakagawa, S., & Cuthill, I. C. (2007). Effect size, confidence interval and statistical significance: A practical guide for biologists. *Biological Reviews*, 82(4), 591–605. https://doi.org/10.1111/j.1469-185X.2007.00027.x

- Nichols, S. L., Schutz, P. A., Rodgers, K., & Bilica, K. (2017). Early career teachers' emotion and emerging teacher identities. *Teachers and Teaching*, 23(4), 406–421. https://doi.org/10.1080/13540602.2016.1211099
- Ogle, R. R., Frazier, S. L., Nichols-Lopez, K., Cappella, E., & The LINKS (Linking Neighborhoods Kids Schools) Center. (2016). Non-academic self-concept among urban youth: Associations with academic success. *School Mental Health*, 8(2), 278– 291. https://doi.org/10.1007/s12310-015-9162-z
- Organisation for Economic Co-operation and Development. (2003). *Learners for life. Student approaches to learning. Results from PISA 2000.* Author. https://www.oecdilibrary.org/education/learners-for-life-student-approaches-tolearning_9789264103917-en
- Pajares, M. F. (1992). Teachers' beliefs and educational research: Cleaning up a messy construct. *Review of Educational Research*, 62(3), 307–332. https://doi.org/10.3102/00346543062003307
- Parkinson, B. (2006). *Ideas and realities of emotion*. Routledge. http://dx.doi.org/10.4324/9780203135648
- Patrician, P. A. (2002). Multiple imputation for missing data. *Research in Nursing & Health*, 25(1), 76–84. https://doi.org/10.1002/nur.10015
- Pekrun, R., Goetz, T., Perry, R. P., Kramer, K., Hochstadt, M., & Molfenter, S. (2004).
 Beyond test anxiety: Development and validation of the Test Emotions Questionnaire (TEQ). *Anxiety, Stress & Coping*, 17(3), 287–316.
- Pekrun, R., & Linnenbrink-Garcia, L. (2014). International handbook of emotions in education. Routledge. http://dx.doi.org/10.4324/9780203148211
- Pekrun, R., Muis, K. R., Frenzel, A. C., & Goetz, T. (2017). *Emotions at school*. Routledge. http://dx.doi.org/10.4324/9781315187822
- Pellegrini, R. J., & Hicks, R. A. (1972). Prophecy effects and tutorial instruction for the disadvantaged child. *American Educational Research Journal*, 9(3), 413–419. https://doi.org/10.3102/00028312009003413
- Pesu, L., Viljaranta, J., & Aunola, K. (2016). The role of parents' and teachers' beliefs in children's self-concept development. *Journal of Applied Developmental Psychology*, 44, 63–71. https://doi.org/10.1016/j.appdev.2016.03.001
- Price, D. D., & Barrell, J. J. (1984). Some general laws of human emotion: Interrelationships between intensities of desire, expectation, and emotional feeling. *Journal of Personality*, 52(4), 389–409. https://doi.org/10.1111/j.1467-6494.1984.tb00359.x

- Pugh, L. G. (1974, April). Teacher attitudes and expectations associated with race and social class (Chicago). [Paper presentation]. The Annual Meeting of the American Educational Research Association, Chicago. https://eric.ed.gov/?id=ED094018
- Raudenbush, S. W. (1984). Magnitude of teacher expectancy effects on pupil IQ as a function of the credibility of expectancy induction: A synthesis of findings from 18 experiments. *Journal of Educational Psychology*, 76(1), 85–97. https://doi.org/10.1037/0022-0663.76.1.85
- Rettinger, D. A. (2017). The role of emotions and attitudes in causing and preventing cheating. *Theory Into Practice*, 56(2), 103–110. https://doi.org/10.1080/00405841.2017.1308174
- Richardson, M., Abraham, C., & Bond, R. (2012). Psychological correlates of university students' academic performance: A systematic review and meta-analysis. *Psychological Bulletin*, 138(2), 353–387. https://doi.org/10.1037/a0026838
- Richardson, V. (1996). The role of attitudes and beliefs in learning to teach. In J. Sikula (Ed.), *Handbook of research on teacher education* (2nd ed., pp. 102–119). Macmillan.
- Rimm-Kaufman, S. E., & Sawyer, B. E. (2004). Primary-grade teachers' self-efficacy beliefs, attitudes toward teaching, and discipline and teaching practice priorities in relation to the" responsive classroom" approach. *The Elementary School Journal*, 104(4), 321– 341. https://doi.org/10.1086/499756
- Rosenthal, R. (1963). On the social psychology of the psychological experiment: The experimenter's hypothesis as unintended determinant of experimental results. *American Scientist*, *51*(2), 268–283.
- Rosenthal, R. (1974). On the social psychology of the self-fulfilling prophecy: Further evidence for Pygmalion effects and their mediating mechanisms. MSS Modular Publications.
- Rosenthal, R. (1976). Experimenter effects in behavioral research. Irvington.
- Rosenthal, R. (1991). Teacher expectancy effects: A brief update 25 years after the Pygmalion experiment. *Journal of Research in Education*, *1*(1), 3–12.
- Rosenthal, R., & Jacobson, L. (1968). *Pygmalion in the classroom: Teacher expectation and pupils' intellectual development*. Holt, Rinehart and Winston.
- Rosenthal, R., & Rubin, D. B. (1978). Interpersonal expectancy effects: The first 345 studies. *Behavioral and Brain Sciences*, 1(3), 377–386. https://doi.org/10.1017/S0140525X00075506

- Rubie-Davies, C. (2014). *Becoming a high expectation teacher: Raising the bar*. Routledge. http://dx.doi.org/10.4324/9781315761251
- Rubie-Davies, C., Hattie, J., & Hamilton, R. (2006). Expecting the best for students: Teacher expectations and academic outcomes. *British Journal of Educational Psychology*, 76(3), 429–444. https://doi.org/10.1348/000709905X53589
- Rubie-Davies, C. M. (2006). Teacher expectations and student self-perceptions: Exploring relationships. *Psychology in the Schools*, 43(5), 537–552. https://doi.org/10.1002/pits.20169
- Rubie-Davies, C. M. (2007). Classroom interactions: Exploring the practices of high- and low-expectation teachers. *British Journal of Educational Psychology*, 77(2), 289–306. https://doi.org/10.1348/000709906X101601
- Rubie-Davies, C. M. (2008). Teacher beliefs and expectations: Relationships with student learning. In C. M. Rubie-Davies & C. Rawlinson (Eds.), *Challenging thinking about teaching and learning* (pp. 25–39). Nova.
- Rubie-Davies, C. M. (2010). Teacher expectations and perceptions of student attributes: Is there a relationship? *British Journal of Educational Psychology*, 80(1), 121–135. https://doi.org/10.1348/000709909X466334
- Rubie-Davies, C. M. (2017). *Teacher expectations in education*. Routledge. https://doi.org/10.4324/9781315520483
- Rubie-Davies, C. M., Flint, A., & McDonald, L. G. (2012). Teacher beliefs, teacher characteristics, and school contextual factors: What are the relationships? *British Journal of Educational Psychology*, 82(2), 270–288. https://doi.org/10.1111/j.2044-8279.2011.02025.x
- Rubie-Davies, C. M., & Peterson, E. R. (2011). Teacher expectations and beliefs: Influences on the socioemotional environment of the classroom. In C. M. Rubie-Davies (Ed.), *Educational psychology: Concepts, research and challenges* (pp. 134–149). Routledge.
- Rubie-Davies, C. M., Peterson, E. R., Sibley, C. G., & Rosenthal, R. (2015). A teacher expectation intervention: Modelling the practices of high expectation teachers. *Contemporary Educational Psychology*, 40, 72–85. https://doi.org/10.1016/j.cedpsych.2014.03.003
- Rubie-Davies, C. M., Stephens, J. M., & Watson, P. (Eds.). (2015). Routledge international handbook of social psychology of the classroom. Routledge. https://doi.org/10.4324/9781315716923

- Rubie-Davies, C. M., Watson, P. W., Flint, A., Garrett, L., & McDonald, L. (2018). Viewing students consistently: How stable are teachers' expectations? *Educational Research and Evaluation*, 24(3–5), 221–240. https://doi.org/10.1080/13803611.2018.1550836
- Rubie-Davies, C. M., Weinstein, R. S., Huang, F. L., Gregory, A., Cowan, P. A., & Cowan,
 C. P. (2014). Successive teacher expectation effects across the early school years. *Journal of Applied Developmental Psychology*, *35*(3), 181–191.
 https://doi.org/10.1016/j.appdev.2014.03.006
- Rubie-Davies, C., Meissel, K., Alansari, M., Watson, P., Flint, A., & McDonald, L. (2020). Achievement and beliefs outcomes of students with high and low expectation teachers. *Social Psychology of Education*, 23(5), 1173–1201. https://doi.org/10.1007/s11218-020-09574-y
- Ruohotie-Lyhty, M. (2013). Struggling for a professional identity: Two newly qualified language teachers' identity narratives during the first years at work. *Teaching and Teacher Education*, 30, 120–129. https://doi.org/10.1016/j.tate.2012.11.002
- Ruohotie-Lyhty, M. (2016). Stories of change and continuity: Understanding the development of the identities of foreign language teachers. In P. Kalaja, A. M. Barcelos, M. Aro, & M. Ruohotie-Lyhty (Eds.), *Beliefs, Agency and Identity in Foreign Language Learning and Teaching* (pp. 172–201). Palgrave Macmillan. https://doi.org/10.1057/9781137425959_9
- Saunders, R. (2013). The role of teacher emotions in change: Experiences, patterns and implications for professional development. *Journal of Educational Change*, 14(3), 303–333. https://doi.org/10.1007/s10833-012-9195-0
- Savage, J. (2004). Researching emotion: The need for coherence between focus, theory and methodology. *Nursing Inquiry*, *11*(1), 25–34.
- Scherer, K. R. (1999). Appraisal theory. In T. Dalgleish & M. J. Power (Eds.), Handbook of cognition and emotion (pp. 637–663). John Wiley & Sons. https://doi.org/10.1002/0470013494.ch30
- Scherer, K. R. (2001). Appraisal considered as a process of multilevel sequential checking. In
 K. R. Scherer, A. Schorr, & T. Johnstone (Eds.), *Appraisal processes in emotion: Theory, methods, research* (pp. 92–120). Oxford University Press.
- Scherer, K. R. (2004). Feelings integrate the central representation of appraisal-driven response organization in emotion. In A. S. Manstead, N. Frijda, & A. Fischer (Eds.), *Feelings and emotions: The Amsterdam symposium* (pp. 136–157). Cambridge University Press. https://doi.org/10.1017/CBO9780511806582.009

- Scherer, K. R. (2009). The dynamic architecture of emotion: Evidence for the component process model. *Cognition & Emotion*, 23(7), 1307–1351. https://doi.org/10.1080/02699930902928969
- Schnitzler, K., Holzberger, D., & Seidel, T. (2021). All better than being disengaged: Student engagement patterns and their relations to academic self-concept and achievement. *European Journal of Psychology of Education*, 36(3), 627–652. https://doi.org/10.1007/s10212-020-00500-6
- Schreiber, J. B., Nora, A., Stage, F. K., Barlow, E. A., & King, J. (2006). Reporting structural equation modeling and confirmatory factor analysis results: A review. *The Journal of Educational Research*, 99(6), 323–338. https://doi.org/10.3200/JOER.99.6.323-338
- Shavelson, R. J., Hubner, J. J., & Stanton, G. C. (1976). Self-concept: Validation of construct interpretations. *Review of Educational Research*, 46(3), 407–441. https://doi.org/10.2307/1170010
- Shi, Q., Liu, X., & Leuwerke, W. (2014). Students' perceptions of school counselors: An investigation of two high schools in Beijing, China. *The Professional Counselor*, 4(5), 519–530. https://doi.org/10.15241/qs.4.5.519
- Silinskas, G., & Kikas, E. (2019). Parental involvement in math homework: Links to children's performance and motivation. *Scandinavian Journal of Educational Research*, 63(1), 17–37. https://doi.org/10.1080/00313831.2017.1324901
- Smith, C. A., & Kirby, L. D. (2009). Putting appraisal in context: Toward a relational model of appraisal and emotion. *Cognition and Emotion*, 23(7), 1352–1372. https://doi.org/10.1080/02699930902860386
- Smith, H. J., Chen, J., & Liu, X. (2008). Language and rigour in qualitative research: Problems and principles in analyzing data collected in Mandarin. *BMC Medical Research Methodology*, 8(1), 44. https://doi.org/10.1186/1471-2288-8-44
- Sorhagen, N. S. (2013). Early teacher expectations disproportionately affect poor children's high school performance. *Journal of Educational Psychology*, 105(2), 465–477. https://doi.org/10.1037/a0031754
- Speybroeck, S., Kuppens, S., Van Damme, J., Van Petegem, P., Lamote, C., Boonen, T., & de Bilde, J. (2012). The role of teachers' expectations in the association between children's SES and performance in kindergarten: A moderated mediation analysis. *PLoS ONE*, 7(4), e34502. https://doi.org/10.1371/journal.pone.0034502

- Sui, C. (2013). Discussion on problems in the specialization development of high school homeroom teachers [Master's thesis, Inner Mongol Normal University]. http://cdmd.cnki.com.cn/Article/CDMD-10135-1013276729.htm
- Susperreguy, M. I., Davis-Kean, P. E., Duckworth, K., & Chen, M. (2018). Self-concept predicts academic achievement across levels of the achievement distribution: Domain specificity for math and reading. *Child Development*, 89(6), 2196–2214. https://doi.org/10.1111/cdev.12924
- Sutton, R. E., & Wheatley, K. F. (2003). Teachers' emotions and teaching: A review of the literature and directions for future research. *Educational Psychology Review*, 15(4), 327–358. https://doi.org/10.1023/A:1026131715856
- Szumski, G., & Karwowski, M. (2019). Exploring the Pygmalion effect: The role of teacher expectations, academic self-concept, and class context in students' math achievement. *Contemporary Educational Psychology*, 59, 101787. https://doi.org/10.1016/j.cedpsych.2019.101787
- Tabachnick, B. G., & Fidell, L. S. (2007). *Experimental designs using ANOVA*. Thomson/Brooks/Cole.
- Tarrant, M., MacKenzie, L., & Hewitt, L. A. (2006). Friendship group identification, multidimensional self-concept, and experience of developmental tasks in adolescence. *Journal of Adolescence*, 29(4), 627–640. https://doi.org/10.1016/j.adolescence.2005.08.012
- Taxer, J. L., & Frenzel, A. C. (2015). Facets of teachers' emotional lives: A quantitative investigation of teachers' genuine, faked, and hidden emotions. *Teaching and Teacher Education*, 49, 78–88. https://doi.org/10.1016/j.tate.2015.03.003
- Thomas, J. A., & Strunk, K. K. (2017). Expectancy-value and children's science achievement: Parents matter. *Journal of Research in Science Teaching*, 54(6), 693– 712. https://doi.org/10.1002/tea.21382
- Thompson, A. G. (1992). Teachers' beliefs and conceptions: A synthesis of the research. In
 D. A. Grouws (Ed.), *Handbook of research on mathematics teaching and learning: A project of the National Council of Teachers of Mathematics* (pp. 127–146). Macmillan Publishing.
- Thompson, B. (2004). Exploratory and confirmatory factor analysis: Understanding concepts and applications. American Psychological Association. http://dx.doi.org/10.1037/10694-000

- Timoštšuk, I., & Ugaste, A. (2012). The role of emotions in student teachers' professional identity. *European Journal of Teacher Education*, 35(4), 421–433. https://doi.org/10.1080/02619768.2012.662637
- Trigwell, K. (2012). Relations between teachers' emotions in teaching and their approaches to teaching in higher education. *Instructional Science*, 40(3), 607–621. https://doi.org/10.1007/s11251-011-9192-3
- Tsang, K. K., & Kwong, T. L. (2017). Teachers' emotions in the context of education reform: Labor process theory and social constructionism. *British Journal of Sociology of Education*, 38(6), 841–855. https://doi.org/10.1080/01425692.2016.1182007
- Upadyaya, K., & Eccles, J. (2015). Do teachers' perceptions of children's math and reading related ability and effort predict children's self-concept of ability in math and reading? *Educational Psychology*, *35*(1), 110–127. https://doi.org/10.1080/01443410.2014.915927
- Urhahne, D., Chao, S.-H., Florineth, M. L., Luttenberger, S., & Paechter, M. (2011). Academic self-concept, learning motivation, and test anxiety of the underestimated student. *British Journal of Educational Psychology*, 81(1), 161–177. https://doi.org/10.1348/000709910X504500
- Usher, E. L. (2015). Personal capability beliefs. In *Handbook of Educational Psychology* (3rd ed., pp. 160–173). Routledge.
- Väisänen, S., Pietarinen, J., Pyhältö, K., Toom, A., & Soini, T. (2017). Social support as a contributor to student teachers' experienced well-being. *Research Papers in Education*, 32(1), 41–55. https://doi.org/10.1080/02671522.2015.1129643
- Van Zoest, L. R., Breyfogle, M. L., & Ziebarth, S. W. (2002). Self-perceived and observed practices of secondary school mathematics teachers. *Teacher Development*, 6(2), 245– 268. https://doi.org/10.1080/13664530200200167
- Veen, K. van, & Lasky, S. (2005). *Emotions as a lens to explore teacher identity and change: Different theoretical approaches.*
- Vygotsky, L. S., & Cole, M. (1978). *Mind in society: Development of higher psychological processes*. Harvard university press.
- Wang, G. (1997). *The homeroom teacher's role in psychological counseling at school*. https://eric.ed.gov/?id=ED439325
- Wang, H., Hall, N. C., & Taxer, J. L. (2019). Antecedents and consequences of teachers' emotional labor: A systematic review and meta-analytic investigation. *Educational Psychology Review*, 31(3), 663–698. https://doi.org/10.1007/s10648-019-09475-3

- Wang, S., Meissel, K., & Rubie-Davies, C. M. (2021). Teacher expectation effects in Chinese junior high schools: Exploring links between teacher expectations and student achievement using a hierarchical linear modelling approach. *Social Psychology of Education*, 24(5), 1305–1333. https://doi.org/10.1007/s11218-021-09654-7
- Wang, S., Rubie-Davies, C. M., & Meissel, K. (2018). A systematic review of the teacher expectation literature over the past 30 years. *Educational Research and Evaluation*, 24(3–5), 124–179. https://doi.org/10.1080/13803611.2018.1548798
- Wang, S., Rubie-Davies, C. M., & Meissel, K. (2019). Instructional practices and classroom interactions of high and low expectation teachers in China. *Social Psychology of Education*, 22(4), 841–866. https://doi.org/10.1007/s11218-019-09507-4
- Wang, S., Rubie-Davies, C. M., & Meissel, K. (2020). The stability and trajectories of teacher expectations: Student achievement level as a moderator. *Learning and Individual Differences*, 78, 101819. https://doi.org/10.1016/j.lindif.2019.101819
- Wang, T., & Yang, D. (2021). Do homeroom teachers affect students' academic achievement in China's middle schools? *Applied Economics Letters*, 28(4), 329–333. https://doi.org/10.1080/13504851.2020.1752897
- Wang, Y., Ramos, A., Wu, H., Liu, L., Yang, X., Wang, J., & Wang, L. (2015). Relationship between occupational stress and burnout among Chinese teachers: A cross-sectional survey in Liaoning, China. *International Archives of Occupational and Environmental Health*, 88(5), 589–597. https://doi.org/10.1007/s00420-014-0987-9
- Wu, T.-J., Wang, L.-Y., Gao, J.-Y., & Wei, A.-P. (2020). Social support and well-being of Chinese special education teachers—An emotional labor perspective. *International Journal of Environmental Research and Public Health*, 17(18), 6884. https://doi.org/10.3390/ijerph17186884
- Wu, Z., & Chen, J. (2018). Teachers' emotional experience: Insights from Hong Kong primary schools. Asia Pacific Education Review, 19(4), 531–541. https://doi.org/10.1007/s12564-018-9553-6
- Xu, Q., Li, S., & Yang, L. (2019). Perceived social support and mental health for college students in mainland China: The mediating effects of self-concept. *Psychology, Health & Medicine*, 24(5), 595–604. https://doi.org/10.1080/13548506.2018.1549744
- Yamasaki, K., Sakai, A., & Uchida, K. (2006). A longitudinal study of the relationship between positive affect and both problem-and emotionfocused coping strategies. *Social Behavior & Personality: An International Journal*, 34(5), 499–510. https://doi.org/10.2224/sbp.2006.34.5.499

- Yan, C. (2015). 'We can't change much unless the exams change': Teachers' dilemmas in the curriculum reform in China. *Improving Schools*, 18(1), 5–19. https://doi.org/10.1177/1365480214553744
- Yao, M., Kadetz, P. I., Sidibe, A. M., Wu, Y., Li, J., Lyu, J., Ma, C., & Hesketh, T. (2021). Teachers' perceptions of student mental health in eastern China: A qualitative study. *International Journal of Environmental Research and Public Health*, 18(14), 7271. https://doi.org/10.3390/ijerph18147271
- Ye, W., Wang, Z., Zhang, X., Ding, Y., & Ye, W. (2021). Comparing motivations of preservice and beginning teachers in China: Impact of culture and experience. *Journal of Education for Teaching*, 0(0), 1–14. https://doi.org/10.1080/02607476.2021.1898911
- Yin, H. (2013). Societal culture and teachers' responses to curriculum reform: Experiences from China. Asia Pacific Education Review, 14(3), 391–401. https://doi.org/10.1007/s12564-013-9266-9
- Yin, H. (2016). Knife-like mouth and tofu-like heart: Emotion regulation by Chinese teachers in classroom teaching. *Social Psychology of Education*, 19(1), 1–22. https://doi.org/10.1007/s11218-015-9319-5
- Yin, H., & Lee, J. C.-K. (2012). Be passionate, but be rational as well: Emotional rules for Chinese teachers' work. *Teaching and Teacher Education*, 28(1), 56–65. https://doi.org/10.1016/j.tate.2011.08.005
- Zembylas, M. (2005). Beyond teacher cognition and teacher beliefs: The value of the ethnography of emotions in teaching. *International Journal of Qualitative Studies in Education*, 18(4), 465–487. https://doi.org/10.1080/09518390500137642
- Zembylas, M. (2007). Theory and methodology in researching emotions in education. *International Journal of Research & Method in Education*, 30(1), 57–72.
- Zhao, X., Selman, R. L., & Haste, H. (2015). Academic stress in Chinese schools and a proposed preventive intervention program. *Cogent Education*, 2(1), 1000477. https://doi.org/10.1080/2331186X.2014.1000477
- Zhao, Z. (2014). The teacher–state relationship in China: An exploration of homeroom teachers' experiences. *International Studies in Sociology of Education*, 24(2), 148– 164. https://doi.org/10.1080/09620214.2014.896569
- Zheng, M. (2020). When and why negative emotional appeals work in advertising: A review of research. Open Journal of Social Sciences, 8(03), 7–16. https://doi.org/10.4236/jss.2020.83002

- Zheng, X., Shi, X., & Liu, Y. (2020). Leading teachers' emotions like parents: Relationships between paternalistic leadership, emotional labor and teacher commitment in China. *Frontiers in Psychology*, 11, 519. https://doi.org/10.3389/fpsyg.2020.00519
- Zhu, M., & Urhahne, D. (2015). Teachers' judgements of students' foreign-language achievement. *European Journal of Psychology of Education*, 30(1), 21–39. https://doi.org/10.1007/s10212-014-0225-6
- Zhu, X., & Liu, C. (2004). Teacher training for moral education in China. *Journal of Moral Education*, *33*(4), 481–494. https://doi.org/10.1080/0305724042000315608

Appendices

Appendix A. Teacher Questionnaires

There are three sections of the questionnaires. Please answer each question to the best of your ability.

Section 1 Demographic Information

	Demographic Information
	Your name: ()
	The homeroom class you are in charge of this school year: Class () Grade ()
	I'm willing to attend the individual interview.
	Yes () Please contact me at:
	No ()
1.	Are you currently a homeroom teacher?
	Yes / No
2.	What is your gender?
	Female / Male
3.	What is your age?
	21-25 / 26-30 / 31-35 / 36-40 / 41-45 / 46-50 / 51-55 / 56-60 / Over 60
4.	What is the highest level of formal education you have completed?
	Bachelor's Degree / Master's Degree / Doctorate
5.	Are you graduated from Normal University?
	Yes / No
6.	Did you receive training to become a homeroom teacher? If so, what was the length of the
	training?
	Yes / No
	Less than 20 hours / 20-40 hours / 40-60 hours / More than 60 hours
7.	How many total years of teaching experience do you have?
	Less than 1 month / Less than 6 months / Less than 1 year / 1-3 years / 4-6 years / 7-10 years
	/ 11-15 years / 16-20 years / More than 20 years
8.	How many total years have you been a homeroom teacher?
	Less than 1 month / Less than 6 months / Less than 1 year / 1-3 years / 4-6 years / 7-10 years
	/ 11-15 years / 16-20 years / More than 20 years

9.	Which of the following best describes the subject/s that you teach this school year? Check all
	that apply.
	Chinese / Mathematics / English / Chemistry / Physics / Arts /
	Physical education / Psychology / Other (please specify) ()
10.	In terms of your subject teaching, which of the following best describes the grade level/s that
	you are teaching this school year? Check all that apply.
	Grades 10 / Grades 11 / Grades 12
11.	How many classes do you teach your subject each week this school year? (assuming that one
	class lasts around 40 minutes)
	Less than 1 class / 1-5 classes / 6-10 classes / 11-15 classes / 16-20 classes /
	21-25 classes / 26-30 classes / More than 30 classes
12.	How many total students are in your homeroom class this year?
	Below 25 / 25-30 / 31-35 / 36-40 / 41-45 / 45-50 / More than 50
13.	About how many hours do you spend per day on work related to your homeroom class?
	Less than 1 hour / 1-2 hours / 2-4 hours / 4-6 hours / 6-8 hours / Over 8 hours
14.	Was in-service training pertaining to being a homeroom teacher provided by your school? If
	yes, how often was it?
	Yes / No
	Once a school year / Once a semester / Twice a semester / More than twice a semester

Section 2 Homeroom Teacher Emotion Scale

Instructions:

In this section, please indicate how often you think or feel with the following statements.

Please choose the number which is the most appropriate for you.

		strongly disagree	disagree	slightly disagree	Neutral	slightly agree	agree	strongly agree
1.	I am glad when I achieve my teaching goals.	1	2	3	4	5	6	7
2.	I am joyful when the class atmosphere is positive.	1	2	3	4	5	6	7
3.	I am happy when I manage to motivate students to learn.	1	2	3	4	5	6	7
4.	I am happy when students understand the material.	1	2	3	4	5	6	7
5.	Exerting a positive influence on my students makes me happy.	1	2	3	4	5	6	7
6.	I feel like a winner when my students succeed.	1	2	3	4	5	6	7
7.	Due to my students' achievements, I feel as if I am 'growing.'	1	2	3	4	5	6	7
8.	I am filled with pride when I make a student interested in my subject.	1	2	3	4	5	6	7
9.	Meetings with successful former students of mine make me proud.	1	2	3	4	5	6	7
10.	When I am proud of my students, I feel that my confidence is growing.	1	2	3	4	5	6	7
11.	Pride due to my students' achievements confirms to me that I am doing a good job.	1	2	3	4	5	6	7
12.	I feel warmth when I just think about my students.	1	2	3	4	5	6	7
13.	I love my students.	1	2	3	4	5	6	7
14.	My students evoke feelings of love inside me.	1	2	3	4	5	6	7
15.	I feel affection towards my students.	1	2	3	4	5	6	7
16.	I wish to praise my students since I like them so much.	1	2	3	4	5	6	7
17.	I honestly care about each of my students.	1	2	3	4	5	6	7

18.	I sweat from frustration when the class does not operate in the way that I want it to.	1	2	3	4	5	6	7
19.	The reactions of some students frustrate me so much that I would rather just quit the job.	1	2	3	4	5	6	7
20.	The frustration I feel while working with students undermines my job motivation.	1	2	3	4	5	6	7
21.	Some students make me so angry that my face goes red.	1	2	3	4	5	6	7
22.	I get an anger-caused headache from the behaviour of some students.	1	2	3	4	5	6	7
23.	At the end of my working day, I just want to rest.	1	2	3	4	5	6	7
24.	When I finish classes, I feel numb.	1	2	3	4	5	6	7
25.	My job sometimes makes me so tired that all I want to do is 'switch off.'	1	2	3	4	5	6	7
26.	Due to the speedy pace of work, at the end of the day I feel as if I am going to fall down.	1	2	3	4	5	6	7
27.	Sometimes I am so exhausted at work that I only think about how to endure.	1	2	3	4	5	6	7
28.	When I finish my work, I feel drained.	1	2	3	4	5	6	7
29.	Sometimes working with children makes me so tired that I can barely move.	1	2	3	4	5	6	7
30.	I feel I cannot do anything more to correct the behaviour of some students.	1	2	3	4	5	6	7
31.	While working with completely unmotivated students, I feel there is no way out.	1	2	3	4	5	6	7
32.	Because of the behaviour of some students, I feel completely helpless.	1	2	3	4	5	6	7
33.	I feel hopeless when I think about the achievement of some students.	1	2	3	4	5	6	7
34.	It seems to me that I cannot do anything to get through to some students.	1	2	3	4	5	6	7
35.	I feel defenceless because I cannot help some of my students.	1	2	3	4	5	6	7
36.	I feel tense and nervous while teaching my students.	1	2	3	4	5	6	7

27	T () ' 1/1 / / 1' ' 2/ ' 11 '/1	1	0	2	4	~	(7
37.	I am often worried that my teaching isn't going so well with	1	2	3	4	Э	6	/
	my students.							
38.	Preparing to teach my students often causes me to worry.	1	2	3	4	5	6	7
39.	I feel uneasy when I think about teaching my students.	1	2	3	4	5	6	7

Section 3 Teacher Expectancy Survey

Student	Instructions:							
number								
	Please indicate the scores you believe your students will achieve at the end							
	of this school year. You may need a list of student names and student number							
	on hand to finish this survey.							
	Note: please assume a	total score per subject is 150.						
	Level 1 (below 60), Le	evel 2 (60–69), Level 3 (70–79	9), Level 4 (80–89),					
	Level 5 (90–99), Level	6 (100–109), Level 7 (110–1	19), Level 8 (120–					
	129), Level 9 (130–139	9), and Level 10 (140–150)						
	Chinese	Mathematics	English					
1.								
2.								
3.								
4.								
5.								
6.								
7.								
8.								
9.								
10.								
11.								
12.								
13.								
14.								
15.								
16.								
17.								
18.								
19.								
20.								
21.								
22.								

23.		
24.		
25.		
26.		
27.		
28.		
29.		
30.		
31.		
32.		
33.		
34.		
35.		
36.		
37.		
38.		
39.		
40.		
41.		
42.		
43.		
44.		
45.		
46.		
47.		
48.		
49.		
50.		
51.		
52.		
53.		
54.		
55.		

56.		
57.		
58.		
59.		
60.		

Thank you for your time and cooperation!

Appendix B. Participant Information Sheets, Consent/Assert Forms



School of Learning, Development, and Professional Practice Faculty of Education and Social Work Private Bag 92019 Auckland 1142, New Zealand The University of Auckland Private Bag 92019 Auckland, New Zealand

PARTICIPANT INFORMATION SHEET

(PRINCIPAL)

Project title: Chinese Homeroom Teachers' Expectations and Relations with their Emotions and Students' Self-Concept
Principal Investigator: Professor Christine Rubie-Davies
Co-investigator: Associate Professor Mei Kuin Lai
Researcher: Mengnan Li

I am Mengnan Li, a doctoral student in the Faculty of Education and Social Work. I am conducting research for my PhD thesis in Education, supervised by Professor Christine Rubie-Davies and Associate Professor Mei Kuin Lai, at the Faculty of Education and Social Work, the University of Auckland, New Zealand.

Project Description and Invitation

The aim of this project is to explore the relations between teacher expectations and their relations with teachers' emotions, and how such interactions might be related to their students' self-concept. There will be three studies in this project, including one questionnaire for teachers, two questionnaires for students, and some possible individual interviews with participating teachers. This project could reveal interesting results and future research directions, which will contribute to the understanding of teacher expectations and their emotions, and relations with students' self-concepts. Therefore, I would like to seek your permission to invite homeroom teachers and students in your school to participate in my research.

Project Procedures

Your participation. All participation in the research will be completely voluntary. If you agree that your school can be involved in this research, you will be asked to sign the Consent Form (CF). If your school agrees to participate, we will ask you to provide a contact list of homeroom teachers for the researcher and allow your students who consent to participate to provide their scores in the latest examination.

Participants' recruiting procedure. By signing the CF, you will give the researcher permission to contact the homeroom teachers and students in consenting homeroom teachers' classrooms. The researcher will email the Participation Information Sheets (PISs) and consent forms (CFs) to all homeroom teachers in your school. Meanwhile, there will be instructions in the PISs for homeroom teachers who agree to participate to distribute students' PISs and CFs to all the participants and parents/caregivers of students under 16 years old. The Qualtrics link of student questionnaire will be printed on student PISs.

Given that the project is entirely voluntary, I seek your assurance that your staff and students are free to participate in the research, and non-participation will not influence teachers' relations with the school, their employment, performance evaluation, and wellbeing, and your students' relation with the school, curriculum grades, dignity, and well-being.

Data Collection Procedure. Participating teachers will be asked to complete an online questionnaire, which will take approximately 20 minutes to complete. Teachers will also be asked to help send the PISs and CFs to their students. Additionally, some homeroom teachers might be invited to attend an individual interview, which will take around 30-60 minutes. All these parts will happen in teachers' work time. To thank them for their participation, each interviewee will be given a \$20 gift card. This will be indicated in their PISs and CFs.

Participating students will be asked to complete an online questionnaire twice. They will need 10 minutes to complete each questionnaire. This point will also be indicated in the PIS for homeroom teachers. As a part of the measure of teacher expectations, student participants will be asked in the questionnaire about their scores of the final examination of the last semester in Maths, English, and Chinese. This could be their senior high school entrance examination scores (for Grade 10 students) or school final examination scores from the last semester (for Grade 11 and 12 students).

Anonymity and Confidentiality

Participants' responses to the questionnaire will not be anonymous because we need to compare teachers' data with those for students. Hence, students will be asked to provide their student ID numbers along with their answers. Additionally, participants in interviews will expose their identity to the researcher only. This will be explicitly communicated to them via their PISs and the CFs they will have to sign prior to participating in this research. However, participants will be given an absolute assurance that their data will be reported anonymously.

Confidentiality is completely guaranteed, as the data collected from participants will be reported anonymously, and the researcher will not be exposing the identity of any participant when reporting the findings from the data. All responses provided in Qualtrics are protected through Transport Layer Security encryption and hosted by data centres using the industry standard SSAE-16 method. Additionally, all data will be coded by a coding method with numbers and letters when transcribed into computer documents. No schools, teachers, or students will be identified by name in any further publications related to this project. Further, no names will be entered into the research database.

Right to Withdraw from Participation

Participation in this research is completely voluntary. You have the right to withdraw your school from the research up to two weeks after the start of data collection without giving a reason. Any data from your school will be destroyed if you choose to withdraw from the research. The researcher will only use previously captured data with your permission.

All teacher and student participants also have the right to choose whether or not to participate in the research. Teacher interviewees will have the right to go through their transcribed interview and confirm whether any part of it needs to be changed or altered. This will be explicitly communicated to participants via their PISs and CFs. Also, participants may withdraw their participation up to two weeks after their completion of their questionnaires/ interviews without giving a reason.

Data Storage/Retention/Future Use

Audio recordings transcribed by the researcher will be deleted from the digital voice recorder within seven days after the date of the interview. Once data have been collected and all information has been transcribed, all study data will be locked in a file cabinet whereas the digital file will be stored in a password protected University of Auckland computer on a University of Auckland server (separate from any consent forms). All study data will be destroyed six years after the doctoral project has been completed.

The data will be used for the student researcher's PhD thesis, and in further academic publications and/or presentations.

A summary of the study findings can be provided to you upon request following the completion of the research in April 2022.

Thank you again for taking the time to consider and participate in this research.

Contact Details

Student Researcher	Supervisors					
Mengnan Li	Christine Margaret Rubie-	Associate Professor Mei				
	Davies	Kuin Lai				
Doctoral Student	Professor	Associate Professor				
Learning, Development and	Learning, Development and	Curriculum and Pedagogy				
Professional Practice	Professional Practice	Faculty of Education and				
Faculty of Education and	Faculty of Education and	Social Work				
Social Work	Social Work	The University of Auckland				
The University of Auckland	The University of Auckland	Private Bag 92601				
Ph: 021 2632779 (NZ) /	Private Bag 92601	Auckland, New Zealand				
+86-13764919859 (CH)	Auckland, New Zealand	Ph: +64 9 373 7999 ext.				
Email:	Ph: (+649) 923 2974	48658				
mengnan.li@auckland.ac.nz	Email:	Email:				
	c.rubie@auckland.ac.nz	mei.lai@auckland.ac.nz				
Head of School: Associate Pro	ofessor Richard Joseph Hamilton	l				
School of Learning, Developm	nent and Professional Practice					
Faculty of Education and Soci	al Work					
The University of Auckland						
Private Bag 92019						
Auckland, New Zealand						
Ph: (+64 9) 923 5619						
Email: rj.hamilton@auckland.	ac.nz					

For any queries regarding ethical concerns you may contact the Chair, The University of Auckland Human Participants Ethics Committee, The University of Auckland, Research Office, Private Bag 92019, Auckland 1142. Telephone 09 373-7599 ext. 87830/83761. E-mail: <u>humanethics@auckland.ac.nz</u>



CONSENT FORM

(PRINCIPAL)

This form will be kept for a period of six years

Project title: Chinese Homeroom Teachers' Expectations and Relations with their Emotions and Students' Self-Concept
Principal Investigator: Professor Christine Rubie-Davies
Co-investigator: Associate Professor Mei Kuin Lai
Researcher: Mengnan Li

I have read the Participant Information Sheet carefully. I understand the nature of the research and why my school has been selected. I have had the opportunity to ask questions and have them answered to my satisfaction.

- I agree to take part in this research conducted by Mengnan Li as part of her doctoral study.
- I agree to provide the contact list of all homeroom teachers in my school (including their e-mail address).
- I agree the consented students in my school to share their latest examination scores with the researcher.
- I allow the researchers to contact homeroom teachers and students in my school.
- I allow the homeroom teachers and students to participate in the project in school time.
- I provide an assurance that my staff and students are free to participate in the research, and non-participation will not influence my staff's relationships with the school, employment, performance evaluation, and

well-being, and my students' relation with the school, curriculum grades, dignity, and well-being as well.

I understand that:

- My participation in this study is entirely voluntary.
- Confidentiality will be completely guaranteed to teachers and students.
- No third party will have access to any data collected in this research.
- If any provided information is reported or published, it will be in a way that does not identify any teacher, students, or my school as a source of the information.
- I am totally free to withdraw my school's participation and/or relevant data within two weeks after the start of data collection without giving a reason.
- I wish/do not wish to receive a summary of findings, which can be emailed to me at this email address: _____

Please write your name, sign and date below if you consent to participating in this research.

Name _____

Signature _____ Date _____



PARTICIPANT INFORMATION SHEET

(TEACHERS)

Project title: Chinese Homeroom Teachers' Expectations and Relations with their Emotions and Students' Self-Concept

Principal Investigator: Professor Christine Rubie-Davies

Co-investigator: Associate Professor Mei Kuin Lai

Researcher: Mengnan Li

I am Mengnan Li, a doctoral student in the Faculty of Education and Social Work. I am conducting research for my PhD thesis in Education, supervised by Professor Christine Rubie-Davies and Associate Professor Mei Kuin Lai, at the Faculty of Education and Social Work, the University of Auckland, New Zealand.

Project Description and Invitation

The aim of this project is to explore the relations between teacher expectations and their emotions, and how such interactions might be related to their students' self-concept. There will be three studies in this project, including one questionnaire for teachers, two questionnaires for students, and some possible individual interviews with participating teachers. This project could reveal interesting results and future research directions, which will contribute to the understanding of teacher expectations, their emotions, and students' self-concepts. Therefore, I would like to invite you to take part in this project as you are a homeroom teacher working in a high school in China.

Project Procedures

Participation in the research is entirely voluntary and I have an assurance from your principal that your participation or non-participation will not affect your relationship with the school, employment, performance evaluation, and well-being. If you agree to participate in this research, you will be asked to sign the Consent Form (CF). Your participation is entirely

voluntary. You are assured that no information you give will/can be shared with your principal and he/she won't even be able to tell whether you have participated in the study.

Your participation in this research might include the following three parts:

(1) Completing an on-line questionnaire, which will take approximately 20 minutes. You are allowed to complete the questionnaire during your work time as the researcher has already acquired the permission of your principal for that. Please enter the link:

https://auckland.au1.qualtrics.com/jfe/form/SV_aV7Fnm5oLUF0BSe

or scan the code:



- (2) You will need to help send the Participation information sheets (PISs), consent forms (CFs) to students and students' (who are under 16 years old) parents/caregivers in your homeroom class. The online link of student questionnaire is shown on student PISs. As a part of the measure of teacher expectations, student participants will be asked in the questionnaire about their scores of the final examination of the last semester in Maths, English, and Chinese. This could be their senior high school entrance examination scores (for Grade 10 students) or school final examination scores from the last semester (for Grade 11 and 12 students). Please provide an assurance that your students are free to participate in the research. Students who are willing to participate in the project will need approximately 10 minutes to finish their questions. Please also inform the students that you have no access to their answers and their participation is totally voluntary.
- (3) Indicating on your CF whether you are willing to attend the follow-up individual interview. You might be invited to attend an individual interview if you volunteered, which will take around 30-60 minutes. Both questionnaire and interviews will happen in your work time. To thank you for your participation in the interviews, you will be given a \$20 gift card. Your interest in participating in the interviews will not affect your participation in the rest of the project. You can participate in the questionnaire but not the interview.

Anonymity and Confidentiality

Your responses to the questionnaire will not be anonymous, because we need to compare your answers with students' data. Hence, you will be asked to provide your homeroom class number on the questionnaire. If you consent to be interviewed, the participants in interviews will expose their identity to the researcher only.

However, the researchers give you an assurance of confidentiality. All responses provided in Qualtrics are protected through Transport Layer Security encryption and hosted by data centres using the industry standard SSAE-16 method. Additionally, all data will be coded by a coding method with numbers and letters when transcribed into computer documents. No schools, teachers, or students will be identified by name in any further publications related to this project. Further, no names will be entered into the research database.

Right to Withdraw from Participation

Participation in this research is completely voluntary. You have the right to withdraw from the research up to two weeks after your completion of the questionnaire /interviews without giving a reason. Any data provided by you will be destroyed if you choose to withdraw from the research. The researcher will only use previously captured data with your permission.

Data Storage/Retention/Future Use

Audio recordings transcribed by the researcher will be deleted from the digital voice recorder within seven days after the date of the interview. Once data have been collected and all information has been transcribed, all study data will be locked in a filing cabinet whereas the digital file will be stored in a password protected University of Auckland computer on a University of Auckland server (separate from any consent forms). All study data will be destroyed six years after the doctoral project has been completed.

The data will be used for the student researcher's PhD thesis, and in further academic publications and/or presentations.

A summary of the study findings can be provided for you upon request following the completion of the research in April 2022.

Thank you again for taking the time to consider and participate in this research.

Contact Details

Student Researcher	Supervisors			
Mengnan Li	Christine Margaret Rubie-	Associate	Professor	Mei
	Davies	Kuin Lai		

Doctoral Student	Professor	Associate Professor					
Learning, Development and	Learning, Development and	Curriculum and Pedagogy					
Professional Practice	Professional Practice	Faculty of Education and					
Faculty of Education and	Faculty of Education and	Social Work					
Social Work	Social Work	The University of Auckland					
The University of Auckland	The University of Auckland	Private Bag 92601					
Ph: 021 2632779 (NZ) /	Private Bag 92601	Auckland, New Zealand					
+86-13764919859 (CH)	Auckland, New Zealand	Ph: +64 9 373 7999 ext.					
Email:	Ph: (+649) 923 2974	48658					
mengnan.li@auckland.ac.nz	Email:	Email:					
	c.rubie@auckland.ac.nz	mei.lai@auckland.ac.nz					
Head of School: Associate Pro	ofessor Richard Joseph Hamilton	1					
School of Learning, Developm	nent and Professional Practice						
Faculty of Education and Soci	ial Work						
The University of Auckland							
Private Bag 92019							
Auckland, New Zealand							
Ph: (+64 9) 923 5619							
Email: rj.hamilton@auckland.ac.nz							
For any queries regarding ethic	For any queries regarding ethical concerns you may contact the Chair, The University of						

Auckland Human Participants Ethics Committee, The University of Auckland, Research Office, Private Bag 92019, Auckland 1142. Telephone 09 373-7599 ext. 87830/83761. Email: <u>humanethics@auckland.ac.nz</u>



CONSENT FORM

(HOMEROOM TEACHERS' QUESTIONNAIRE)

This form will be kept for a period of six years

Project title: Chinese Homeroom Teachers' Expectations and Relations with their Emotions and Students' Self-Concept
Principal Investigator: Professor Christine Rubie-Davies
Co-investigator: Associate Professor Mei Kuin Lai
Researcher: Mengnan Li

I have read the Participant Information Sheet and have understood the nature of the research and why I have been selected. I have had the opportunity to ask questions and have them answered to my satisfaction.

- I agree to take part in the questionnaire.
- I agree to send the Participant Information Sheet and the Consent Form to the students in my homeroom class.
- I agree to allow students who are willing to participate in the project approximately 10 minutes to finish their questions.
- I provide an assurance that my students are free to participate in the research.
- I agree/do not agree to participate in the following individual interview. If Yes please provide your Contact details:

I understand that

• My participation is voluntary and will take around 20-30 minutes for the questionnaire.

- I'm free to withdraw participation and any data traceable to me up to two weeks from completing the questionnaire without giving a reason.
- I do not have to answer any questions that I do not wish to answer, without giving a reason.
- My responses will be treated in a confidential manner.
- The data I provide will be stored securely on the University of Auckland server and will be kept for a minimum of 6 years or until the project is completed, after which they will be destroyed.
- The data collected from me will be used for the student researcher's PhD thesis and other publications or conference presentations.
- An assurance has been given by the principal that I am free to participate in the research, and non-participation will not influence my relations with the school, employment, performance evaluation, and well-being.
- No third party will have access to any data collected in this research.
- I wish/do not wish to receive a summary of findings, which can be emailed to me at this email address: _____

Please write your name, sign and date below if you consent to participating in this research.

Name _____

Signature _____ Date _____



CONSENT FORM

(HOMEROOM TEACHERS' INTERVIEWS)

This form will be kept for a period of six years

Project title: Chinese Homeroom Teachers' Expectations and Relations with their Emotions and Students' Self-Concept

Principal Investigator: Professor Christine Rubie-Davies

Co-investigator: Associate Professor Mei Kuin Lai

Researcher: Mengnan Li

I have read the Participant Information Sheet and have understood the nature of the research and why I have been selected. I have had the opportunity to ask questions and have them answered to my satisfaction.

• I agree to take part in the interview.

I understand that

- My participation is voluntary and will take between 30 to 60 minutes.
- I'm free to withdraw participation and any data traceable to me up to two weeks from completing the interview.
- I will receive a \$20 gift voucher for my participation at the end of the interview.
- My interview will be audio-recorded and transcribed.
- I have the option to stop the recording at any time during the interview.
- I do not have to answer any questions that I do not wish to answer, without giving a reason.
- The audio recording in digital voice recorder will be deleted within seven days after the completion of the interview.
- My responses will be treated in a confidential manner.
- The data I provide will be stored securely on the University of Auckland server and will be kept for a minimum 6 years or until the project is completed, after which they will be destroyed.
- The data collected from my interview will be used for the student researcher's PhD thesis and other publications or conference presentations.
- No third party will have access to any data collected in this research.
- I wish / do not wish to receive a copy of the transcription. If yes, I will be allowed to change and withdraw any data I like without giving a reason within two weeks of receiving the transcripts.

If Yes, please provide your contact details:

• I wish/do not wish to receive a summary of findings, which can be emailed to me at this email address: ______

Please write your name, sign and date below if you consent to participating in this research.

Name _____

Signature _____ Date _____



PARTICIPANT INFORMATION SHEET

(Students 16 years old and over)

Project title: Chinese Homeroom Teachers' Expectations and Relations with their Emotions and Students' Self-Concept

Principal Investigator: Professor Christine Rubie-Davies

Co-investigator: Associate Professor Mei Kuin Lai

Researcher: Mengnan Li

I am Mengnan Li, a doctoral student in the Faculty of Education and Social Work. I am conducting research for my PhD thesis in Education, supervised by Professor Christine Rubie-Davies and Associate Professor Mei Kuin Lai, at the Faculty of Education and Social Work, the University of Auckland, New Zealand.

Project Description and Invitation

The aim of this project is to explore the relations between teacher expectations and their emotions, and how such interactions might be related to their students' self-concept. I would like to invite you to take part in this project as you are a high school student and your information will be valuable for this project. The questionnaire has been designed to understand and track the development of your self-concepts.

Project Procedures

Your participation. If you agree to participate in this research, you will be asked to sign the Consent Form (CF). You can then click a button at the end of the form and it will come to us. Your participation in this research is entirely voluntary. You are assured that no information you give will/can be shared with your principal or teachers and they won't even be able to tell whether you have participated in the study.

Your participation will be completing a questionnaire twice. Each of them will take approximately 10 minutes and you can complete them online. The questionnaire involves 1) a

question on your latest scores in Chinese, Mathematics and English; and 2) questions related to your self-concepts. It is important to know that your willingness to participate in these two parts will not affect your taking part in either one. No one except the researcher will get access to your answers.

If you are willing to participate, please enter the link: https://auckland.au1.qualtrics.com/jfe/form/SV_cBGa0fJuhTltKTQ

or scan the code:



Anonymity and Confidentiality

Your responses to the questionnaire will not be anonymous, because we need to compare your answers with your homeroom teachers' data, you will be asked to provide your student ID number on the questionnaire.

However, the researchers give you an assurance of confidentiality. Regarding questionnaires, all responses provided in Qualtrics are protected through Transport Layer Security encryption and hosted by data centres using the industry standard SSAE-16 method. Additionally, all data will be coded using a coding method with numbers and letters when transcribed into computer documents. No schools, teachers, or students will be identified by name in any publications related to this project. Further, no names will be entered into the research database.

Right to Withdraw from Participation

Participation in this research is completely voluntary. You have the right to withdraw from the research up to two weeks after completing the questionnaires without giving a reason. Any data provided by you will be destroyed if you choose to withdraw from the research. The researcher will only use previously captured data with your permission.

Data Storage/Retention/Future Use

Once data have been collected and all information has been transcribed, all study data will be locked in a filing cabinet whereas the digital file will be stored in a password protected University of Auckland computer on a University of Auckland server (separate from any consent forms). All study data will be destroyed six years after the doctoral project has been completed.

The data will be used for the student researcher's PhD thesis, and in further academic publications and/or presentations.

Thank you again for taking the time to consider participating in this research.

Contact Details

Student Researcher	Supervisors				
Mengnan Li	Christine Margaret Rubie-	Associate Professor Mei			
	Davies	Kuin Lai			
Doctoral Student	Professor	Associate Professor			
Learning, Development and	Learning, Development and	Curriculum and Pedagogy			
Professional Practice	Professional Practice	Faculty of Education and			
Faculty of Education and	Faculty of Education and	Social Work			
Social Work	Social Work	The University of Auckland			
The University of Auckland	The University of Auckland	Private Bag 92601			
Ph: 021 2632779 (NZ) /	Private Bag 92601	Auckland, New Zealand			
+86-13764919859 (CH)	Auckland, New Zealand	Ph: +64 9 373 7999 ext.			
Email:	Ph: (+649) 923 2974	48658			
mengnan.li@auckland.ac.nz	Email:	Email:			
	c.rubie@auckland.ac.nz	mei.lai@auckland.ac.nz			
Head of School: Associate Pro	ofessor Richard Joseph Hamilton	l			
School of Learning, Developm	nent and Professional Practice				
Faculty of Education and Soci	ial Work				
The University of Auckland					
Private Bag 92019					
Auckland, New Zealand					
Ph: (+64 9) 923 5619					
Email: rj.hamilton@auckland	.ac.nz				

For any queries regarding ethical concerns you may contact the Chair, The University of Auckland Human Participants Ethics Committee, The University of Auckland, Research Office, Private Bag 92019, Auckland 1142. Telephone 09 373-7599 ext. 87830/83761. E-mail: <u>humanethics@auckland.ac.nz</u>



CONSENT FORM

(Students 16 years old and over)

This form will be kept for a period of six years

Project title: Chinese Homeroom Teachers' Expectations and Relations with their Emotions and Students' Self-Concept
Principal Investigator: Professor Christine Rubie-Davies
Co-investigator: Associate Professor Mei Kuin Lai
Researcher: Mengnan Li

I have read the Participant Information Sheet and have understood the nature of the research and why I have been selected. I have had the opportunity to ask questions and have them answered to my satisfaction.

• I agree to complete the two questionnaires.

I understand that

- My participation is voluntary and will take around 20 minutes in total.
- My willingness to share my examination scores with the researchers will not affect my participation in the questionnaire, and vice versa.
- I'm free to withdraw participation and any data traceable to me up to two weeks from completing the questionnaire/sharing my scores without giving a reason. I do not have to answer any questions that I do not wish to answer, without giving a reason.
- My responses and my scores will be treated in a confidential manner if I choose to share them with the researchers.
- The data I provide will be stored securely on the University of Auckland server and will be kept for a minimum 6 years or until the project is completed, after which they will be destroyed.

- The data collected from me will be used for the student researcher's PhD thesis and other publications or conference presentations.
- No third party will have access to any data collected in this research.

Please write your name, sign and date below if you consent to participating in this research.

Name _____

Signature _____ Date _____

Please click this button if you have read the Consent Form and agree to be part of this study.



CONSENT FORM

(Parents/Caregivers)

This form will be kept for a period of six years

Project title: Chinese Homeroom Teachers' Expectations and Relations with their Emotions and Students' Self-Concept
Principal Investigator: Professor Christine Rubie-Davies
Co-investigator: Associate Professor Mei Kuin Lai
Researcher: Mengnan Li

I have read the Participant Information Sheet and have understood the nature of the research and why my child has been selected. I have had the opportunity to ask questions and have them answered to my satisfaction.

• I voluntarily agree that my child may participate in completing two questionnaires for this research.

I understand that

- The participation of my child is voluntary and will take around 20 minutes in total.
- The willingness to share the examination scores with the researchers will not affect the participation in the questionnaire, and vice versa.
- I'm free to withdraw my child's participation and any data traceable to my child up to two weeks from completing the questionnaire/sharing the scores without giving a reason.
- My child does not have to answer any questions that my child does not wish to answer, without giving a reason.
- The responses and scores my child provide will be treated in a confidential manner if my child choose to share them with the researchers.

- The data my child provide will be stored securely on the University of Auckland server and will be kept for a minimum 6 years or until the project is completed, after which they will be destroyed.
- The data collected from my child will be used for the student researcher's PhD thesis and other publications or conference presentations.
- No third party will have access to any data collected in this research.

Please write your name, sign and date below if you consent your child to participating in this research, and then click the button below which will send your consent to the student researcher. If you do not consent to your child taking part in this study, simply close your browser. Your form will not be submitted.

Name		
Signature	Date	

Please click this button if you have read the Consent Form and agree for your child to be part of this study.



ASSENT FORM

(Students under 16 years old)

To all students who are under 16 years old from _____School

I am Mengnan Li, a doctoral student in the Faculty of Education and Social Work. I am conducting research for my PhD thesis in Education, supervised by Professor Christine Rubie-Davies and Associate Professor Mei Kuin Lai, at the Faculty of Education and Social Work, the University of Auckland, New Zealand.

The aim of this project is to explore the relations between teacher expectations and their emotions, and how such interactions might be related to their students' self-concept. I would like to invite you to take part in this project as you are a high school student and your information will be valuable for this project. The questionnaire has been designed to understand and track the development of your self-concepts.

I will be working in your school and other schools conduct a student questionnaire twice, one homeroom teacher questionnaire and collecting your scores. The purpose of the student questionnaire is to investigate how student self-concepts are related to teacher expectations. The collection of your scores is to evaluate the level of teacher expectations which requires a comparison between your actual achievement and how your teacher believe you will achieve in future. You are warmly invited to participate in this research.

If you agree to participate in this research, you will be asked to sign the Consent Form (CF). Your participation in this research is entirely voluntary. You are assured that no information you give will/can be shared with your principal or teachers and they won't even be able to tell whether you have participated in the study.

If you consent to be involved in this study, you will be asked to complete a questionnaire twice. Each of them will take approximately 10 minutes and you can complete

them either online. The questionnaire involves 1) a question on your latest scores in Chinese, Mathematics, and English; and 2) questions related to your self-concepts. It is important to know that your willingness to participate in these two parts will not affect your taking part in either one. No one except the researcher will get access to your answers.

If you would like to participate in two questionnaires, and / or sharing your latest scores, please sign your name below:

Name _____

Signature _____ Date _____

Please click this button if you have read the Consent Form and agree to be part of this study.

Appendix C. Student Self-Description Questionnaires

In this section, please indicate how often you think or feel with the following statements.

Please choose the number which is the most appropriate for you.

	Student number: ()	ee		ee				د
	Grade () Class ()	lisagr	ree	isagr	ral	agree	ee	agre
	My recent score is ()	ongly d	disag	ghtly d	Neut	ightly	agre	rongly
		stro		slig		sl		st
1.	I often tell small lies to avoid embarrassing situations.	1	2	3	4	5	6	7
2.	People can always rely on me.	1	2	3	4	5	6	7
3.	Being honest is not particularly important to me.	1	2	3	4	5	6	7
4.	I nearly always tell the truth.	1	2	3	4	5	6	7
5.	I sometimes take things that do not belong to me.	1	2	3	4	5	6	7
6.	I never cheat.	1	2	3	4	5	6	7
7.	Being dishonest is often the lesser of two evils.	1	2	3	4	5	6	7
8.	I am a very honest person.	1	2	3	4	5	6	7
9.	I would feel OK about cheating on a test as long as I did	1	2	3	4	5	6	7
	not get caught.							
10.	I value integrity above all other virtues.	1	2	3	4	5	6	7
11.	I am not a very reliable person.	1	2	3	4	5	6	7
12.	I have never stolen anything of consequence.	1	2	3	4	5	6	7
13.	I am usually pretty calm and relaxed.	1	2	3	4	5	6	7
14.	I worry a lot.	1	2	3	4	5	6	7
15.	I am happy most of the time.	1	2	3	4	5	6	7
16.	I am anxious much of the time.	1	2	3	4	5	6	7
17.	I hardly ever feel depressed.	1	2	3	4	5	6	7
18.	I tend to be high-strung, tense, and restless.	1	2	3	4	5	6	7
19.	I do not spend a lot of time worrying about things	1	2	3	4	5	6	7
20.	I am often depressed.	1	2	3	4	5	6	7
21.	I am inclined towards being an optimist.	1	2	3	4	5	6	7
22.	I tend to be a very nervous person.	1	2	3	4	5	6	7
23.	I am never able to think up answers to problems that	1	2	3	4	5	6	7
	haven't already been figured out.							

24.	I am good at combining ideas in ways that others have	1	2	3	4	5	6	7
	not tried.							
25.	I wish I had more imagination and originality.	1	2	3	4	5	6	7
26.	I enjoy working out new ways of solving problems.	1	2	3	4	5	6	7
27.	I'm not much good at problem solving.	1	2	3	4	5	6	7
28.	I have a lot of intellectual curiosity.	1	2	3	4	5	6	7
29.	I am not very original in my ideas, thoughts, and	1	2	3	4	5	6	7
	actions.							
30.	I am an imaginative person.	1	2	3	4	5	6	7
31.	I would have no interest in being an inventor.	1	2	3	4	5	6	7
32.	I can often see better ways of doing routine tasks.	1	2	3	4	5	6	7
33.	I find many mathematical problems interesting and	1	2	3	4	5	6	7
	challenging.							
34.	I have hesitated to take courses that involve	1	2	3	4	5	6	7
	mathematics.							
35.	I have generally done better in mathematics courses than	1	2	3	4	5	6	7
	other courses.							
36.	Mathematics makes me feel inadequate.	1	2	3	4	5	6	7
37.	I am quite good at mathematics.	1	2	3	4	5	6	7
38.	I have trouble understanding anything that is based upon	1	2	3	4	5	6	7
	mathematics.							
39.	I have always done well in mathematics classes.	1	2	3	4	5	6	7
40.	I never do well on tests that require mathematical reasoning.	1	2	3	4	5	6	7
41.	At school, my friends always come to me for help in	1	2	3	4	5	6	7
	mathematics.							
42.	I have never been very excited about mathematics.	1	2	3	4	5	6	7
43.	I have trouble expressing myself when trying to write	1	2	3	4	5	6	7
	something.							
44.	I can write effectively.	1	2	3	4	5	6	7
45.	I have a poor vocabulary.	1	2	3	4	5	6	7
46.	I am an avid reader.	1	2	3	4	5	6	7
47.	I do not do well on tests that require a lot of verbal	1	2	3	4	5	6	7
	reasoning ability.							

48.	Relative to most people, my verbal skills are quite good.	1	2	3	4	5	6	7
49.	I often have to read things several times before I	1	2	3	4	5	6	7
	understand them.							
50.	I am good at expressing myself.	1	2	3	4	5	6	7
51.	In school I had more trouble learning to read than most	1	2	3	4	5	6	7
	other students.							
52.	I have good reading comprehension.	1	2	3	4	5	6	7

Thank you for your time and cooperation!

Appendix D. Interview Protocol

Interview Protocol Form

(Homeroom teachers)

Interviewee:	 	
Interviewer:	 	

Protocol Sections:

A: Introductory Protocol

B: Interview Background

C: Interview Questions

D: Interview Wrap Up

E: Post Interview Comments and/or Observations

A: Introductory Protocol

Hello ______. Thanks for joining me today. My name is ______ and I will be conducting the interview today. How are you going today? [fill with appropriate response]

You have been selected to speak with me today because you wished to share what you know about teacher expectations and emotions. Our research project as a whole focuses on the relations between these two factors in the context of homeroom teachers, and how such interactions might affect your teaching practices. This research project will also look at students' self-concepts and how these are related to teacher expectations. Today's interview does not aim to evaluate your views or experiences. Rather, we are trying to learn more about how you believe that your expectations are related to your emotions, and to investigate how such relations predict your practice.

To facilitate my note-taking, I would like to audio tape our conversation today. For your information, only researchers on the project will be privy to the recordings which will be eventually destroyed after they are transcribed and the project has been completed. Please sign the consent form [if they have not already done so]. Essentially, this form states that: (1) all information will be held confidential, (2) your participation is voluntary and you may stop at any time without giving a reason, and (3) your interview will be audio-recorded and transcribed. Thank you for your agreeing to participate.

I have planned this interview to last no longer than one hour. During this time, I have several questions that I would like to cover. If time begins to run short, it may be necessary to interrupt you in order to push ahead and complete the questioning.

B: Interviewee Background

Tell me more about yourself (year level, degree, major, ethnicity, etc). How long have you been a high school homeroom teacher? What kind of grade and subjects do you currently teach?

C: Interview Questions

Interview Questions

Qu	iestions
Se	mistructured Questions
1.	Can you please describe in five words the most frequent emotions you have experienced
	as an HT?
2	In what ways if any do you believe that your emotions are related to your expectations

- In what ways, if any, do you believe that your emotions are related to your expectations of students in your role as an HT?
- 3. How do you think your emotions influence your teaching practices in class?
- 4. How do you think your expectations of students influence your teaching practices in class?
- 5. In what ways, if any, do you believe that you can deal with your negative emotions effectively in class?

Scenario-Based Questions

1. Improper behaviours of students

Please imagine a scenario where you are informed that some of your students have cheated in their exams. But the subject they cheated on was not the one that you teach. This means that you need to address this issue together with your colleagues (e.g., the subject teachers, the school leader....). How do you think this will affect you? How will you respond to this issue?

2. Pressures from parents

Please imagine a scenario where some parents of the students in your homeroom class blame you for the regressions in learning of their children. This means that the parents doubt the quality of your instruction and attribute the failure of the students to you. How do you think this will affect you? How will you respond to this issue?

3. Progress of students

Please imagine a scenario where some students in your homeroom class have made impressive progress in the final exam, especially the subject taught by you. How do you think this will affect you? How will you respond to this issue?

4. Disengagement of students

Please imagine a scenario where students in your homeroom class are relatively less active during class time, especially in the subject taught by you. For example, they are not engaged in the group discussions or are unresponsive in your class. How do you think this will affect you? How will you respond to this issue?

D: Interview Wrap Up

Is there anything else you would like to add?

Do you have any questions?

Your Participant Information Sheet includes all the relevant information regarding the project as well as our contact details should you have any further queries.

On behalf of the research team, I would like to thank you for participating in this research project. We really appreciate it.

E: Post Interview Comments and/or Observations: